



LOAN COPY: RETURN TO AFWL (WLOL) KIRTLAND AFB, N MEX

ANALYTICAL AND EXPERIMENTAL INVESTIGATION OF HIGH ENTRAINMENT JET PUMPS

by Kenneth E. Hickman, Gerald B. Gilbert, and John H. Carey

Prepared by DYNATECH R/D COMPANY Cambridge, Mass. for Ames Research Center

NASA CR-1602

NATIONAL AERONAUTICS AND SPACE ADMINISTRATION • WASHINGTON, D. C. • JULY 1970



ANALYTICAL AND EXPERIMENTAL INVESTIGATION OF HIGH ENTRAINMENT JET PUMPS

R.

By Kenneth E. Hickman, Gerald B. Gilbert, and John H. Carey

Vielander verster der son der s Son der son der

Prepared under Contract No. NAS 2-4824 by DYNATECH R/D COMPANY Correct Proceedings, Mass.

for Ames Research Center

NATIONAL AERONAUTICS AND SPACE ADMINISTRATION

For sale by the Clearinghouse for Federal Scientific and Technical Information Springfield, Virginia 22151 - CFSTI price \$3.00

TABLE OF CONTENTS

Section

4

. ----

SUMMARY

-

1	INTR	ODUCTIO	ON	2
	1.1 1.2 1.3	Backgr Previo Purpos	round us Work se	2 3 5
2	SYMI	BOLS		6
	2.1 2.2	Symbol Symbol	s Used in the Analysis and Test Results Is Used in the Computer Program	6 9
3	ANAI	LYSIS		12
	3.1	Formu	lation of the Mathematical Model	12
		$\begin{array}{c} 3. \ 1. \ 1 \\ 3. \ 1. \ 2 \\ 3. \ 1. \ 3 \\ 3. \ 1. \ 4 \\ 3. \ 1. \ 5 \\ 3. \ 1. \ 6 \\ 3. \ 1. \ 7 \end{array}$	Assumptions Used in the Analysis Analysis of the Accommodation Region Analysis of the Mixing Region Jet Pump Performance Parameters Inclusion of Mixing Tube Wall Friction Dimensionless Formulation Maximum Entrainment Ratio	13 13 16 18 19 22 28
	3.2	The Co	mputer Program	28
	3.3 3.4	Solution Jet Pur	ns for a Range of Jet Pump Designs np-Duct Matching Considerations	29 31
		3.4.1 3.4.2 3.4.3 3.4.4	Representation of Duct Loss Characteristics Evaluation of the Loss Coefficient, K, Development of System Performance Equations Calculation Procedure for Determining the	31 32 33
		3.4.5	Operating Point of a Jet Pump in a Duct System Sample Calculation	$\frac{38}{40}$
		3.4.6	Losses Upon Entrainment	42
		3.4.7 3.4.8	Losses Upon Thrust Augmentation Conclusions	44 45
4	TEST	PROGRA	Μ	47
	$4.1 \\ 4.2$	Test Ar Instrum	rangement entation and Data Reduction Procedures	47 50

ii

TABLE OF CONTENTS (Continued)

Section	

		$\begin{array}{c} 4. \ 2. \ 1 \\ 4. \ 2. \ 2 \\ 4. \ 2. \ 3 \\ 4 \ 2 \ 4 \end{array}$	Instrumentation Data Reduction Procedures Suction Duct and Nozzle Cluster Losses Pressure Loss Due to Wall Friction in the	50 51 53
		1.2.1	Constant-Area Mixing Tube	56
	4.3	Tabula	tion of Test Conditions	57
		4.3.1	Presentation of Data	58
	4.4	Discus	sion of Test Results	58
		4.4.1	Comparison of Constant-Area and NAS 2-2518 Verying, Area, Mixing Tubes	58
		4 4 2	Reduced Blockage Nozzle Clusters	63
		4.4.3	Comparison of Short and Extended Mixing Tubes	66
		4.4.4	Velocity Profile Investigations	71
5	COMPA RESUL	ARISON .TS	OF ANALYTICAL AND EXPERIMENTAL	7 4
	5.1	Jet Pu	mp Stagnation Pressure Rise	74
	5.2	Jet Pu	mp Static Pressure Rise	76
6	CONCI	LUSION	S	78
	APPEN	IDIX A	- Listing of the Computer Program	81
	APPEN	DIX B	Discussion of the Computer Program by Blocks	90
	APPEN	DIX C	- Typical Computer Solutions	98
	REFER	RENCES	3	107
	TABLE	s		110
	FIGUR	ES		129

- ----

ANALYTICAL AND EXPERIMENTAL INVESTIGATION OF HIGH ENTRAINMENT JET PUMPS

9

by Kenneth E. Hickman, Gerald B. Gilbert, and John H. Carey

SUMMARY

The use of jet pumps is of increasing interest for boundary layer control or control force augmentation in V/STOL aircraft. In typical applications, a small mass flow of primary air at pressures up to 400 psia can be used to entrain a much larger mass flow of secondary air at ambient conditions. The primary nozzle flow is supersonic while the secondary flow is subsonic. The jet pump system design objectives may be maximum entrainment, maximum thrust augmentation, or some combination of the two. Little information is available in the literature to guide the designer of jet pumps for such applications.

In this investigation, a simple analytical model was developed to predict the performance of high-entrainment compressible flow jet pumps with constant area mixing tubes. While the model is suitable for hand calculation, a computer program was prepared to facilitate calculation of jet pump performance curves and allow comparison of different jet pump designs. Analytical techniques were developed for matching the jet pump design to its associated duct system in order to achieve maximum entrainment or thrust augmentation.

The validity of the analytical model was confirmed by an extensive test program using a multiple-nozzle jet pump with two different mixing tube lengths. The primary-to-secondary flow area ratios were varied from 0.0013 to 0.0067. The primary flow pressure ranged from 55 psia to 350 psia and the primary flow temperature ranged from 200° F to 1200° F. The observed entrainment ratios varied from 15 to 37. The performance of each jet pump geometry was measured over a very broad range of operating conditions in order to develop performance maps for comparison with the analytical predictions.

Section 1

INTRODUCTION

1.1 Background

Jet pumps have been used for many years in industrial applications where a high-pressure gas such as steam is used to pump a lower-pressure gas. The jet pump is a simple low-cost device with no moving parts and is particularly convenient for use with troublesome fluids such as two-phase flows, high-temperature gases, or corrosive gases. Jet pumps are usually employed as low-pressure-rise devices and their thermodynamic efficiency is low, i.e., under 20%. Because they are lowcost devices of limited performance potential, there has not been a strong incentive for research and development work on industrial jet pumps.

In recent years, applications of jet pumps to boundary layer control systems have become of increasing interest for STOL aircraft. Systems have been proposed which use jet pumps to entrain a large flow of secondary air which is then directed over a deflected flap for lift augmentation. In a configuration patented by F. G. Wagner (references 1 and 2), a jet pump is used to entrain air from one section of the trailing edge of a wing (boundary layer suction upstream of a deflected flap) and then to discharge it over a deflected flap. In this way, the inherent inefficiency of the jet pump is partially balanced by the double employment of the entrained air for boundary layer control. Jet pumps may also have application in VTOL aircraft for direct lift or control force augmentation. The primary, high-pressure flow for the jet pumps can be provided by a bleed from the main engine compressors or by an auxiliary power unit.

The use of jet pumps as primary components of V/STOL aircraft systems places new emphasis upon development of design techniques for these devices. It is essential to be able to minimize the size of jet pumps for particular primary and secondary flow conditions, and to be able to predict the performance of jet pumps over a broad range of operating conditions. However, systematic design and analysis procedures are not available for high-entrainment-ratio compressible-flow jet pumps.

1.2 Previous Work

A number of investigators have carried out analytical and experimental studies of air-to-air jet pumps, primarily for applications requiring high pressure rise or thrust augmentation. The entrainment ratios developed by these jet pumps are low, generally less than 10. Thus, this work is not directly applicable to the high entrainment requirements of V/STOL aircraft systems. Nevertheless, this work provides useful guidance for the development of performance prediction techniques and design rules for high-entrainment jet pumps. A brief review of some of the principal air-to-air jet pump papers follows.

The performance of constant-area jet pumps was analyzed by McClintock and Hood for a range of design and operating conditions in reference 3. The analysis was prepared by assuming incompressible flow but the influence of compressibility was discussed in qualitative terms. Empirical coefficients derived by testing were included in the theory. The influence of mixing length and the use of various multiple-nozzle primary flow geometries were studied experimentally. The jet pumps treated had entrainment ratios of 10 or less; the design goal for the study was achievement of maximum thrust augmentation.

A one-dimensional method of analysis of jet pumps was developed by Keenan, Neumann, and Lustwerk in reference 4. The analysis was applied to both constant-area and constant-pressure mixing processes. Test results were obtained for jet pumps with secondary-to-primary area ratios up to 100, primary-to-secondary pressure ratios up to 200, and a primary-to-secondary temperature ratio of 1.0. The various regimes of operation of jet pumps with supersonic primary flows and both supersonic and subsonic primary flows were described. The analytical results given were for jet pumps developing substantial stagnation pressure rises (e.g., pressure ratios from 2 to 10) at low entrainment ratios (under 10).

Fabri carried out a number of experiments on jet pumps which supplement the results reported by Keenan, et al. Fabri's results, with supporting analysis, are given in references 5 and 6. These tests also were confined to low entrainment ratio jet pumps. Excellent agreement was obtained between analytical predictions and measured jet pump performance. An extensive analytical and experimental program was conducted at the University of Minnesota Rosemount Aeronautical Laboratories on jet pumps with secondary-to-primary area ratios up to 36, primary-to-secondary pressure ratios up to 32, and primary-to-secondary temperature ratios up to 3.3. The results are presented in references 7, 8, and 9. Single-nozzle primary flows with constantarea mixing tubes were used in these studies. High entrainment ratios were not an objective of the jet pump design; typical entrainment ratios reported were less than 3. A number of analytical results showing the influence of duct matching upon jet pump and system performance were included in the reports.

An analytical procedure for constant-area jet pumps with subsonic primary flow was developed in reference 10. A computer program was prepared for use in optimizing jet pump design for particular application requirements. An analysis applicable to supersonic primary flows is given in reference 11. The analysis was compared with test results, but was not otherwise applied for jet pump design or optimization.

The performance of a high-entrainment jet pump was measured in the Wagner "Jet Induced Lift" boundary layer control system (figure 1). These tests were performed by the present investigators under NASA contract No. NAS2-2518. The results are reported in reference 12. The jet pump component of the system employed a variable-area mixing tube (designed in an effort to obtain constant pressure mixing) and a 9-nozzle cluster for the primary flow. The jet pump was tested in the system at secondary-to-primary area ratios ranging from 150 to 800, primaryto-secondary pressure ratios up to 26, and primary-to-secondary temperature ratios up to 5.5. The desired constant pressure distribution in the mixing tube was not achieved. The entrainment ratios predicted for the complete system were not attained. The results of the NAS2-2518 program showed that the methods used to design the high-entrainment jet pump and to match it to the duct flow characteristics were inadequate. These results provided the impetus for the present study.

1.3 Purpose

The objectives of this investigation were as follows:

- to develop analytical procedures for predicting the performance of high-entrainment-ratio jet pumps
- to demonstrate the application of these procedures to match a jet pump design to its connecting duct system
- to verify the analytical procedures by testing a jet pump over a broad range of operating conditions and jet pump geometries.

The analysis and experimental work were confined to constant-area mixing tube geometries because both analysis and construction are simplified by this choice. The only other mixing process which can be analyzed without complication is the constant pressure case. However, no reliable methods are available for designing a mixing tube which will actually achieve constant pressure mixing. Furthermore, this condition can be achieved at only one operating point for a jet pump of fixed geometry.

Some of the test results obtained under contract NAS 2-2518 indicated that the design of the nozzle cluster and its position in the mixing tube may have created either high pressure losses in the secondary flow or poor mixing conditions at the mixing tube inlet. The mixing process did not seem to be completed within the length of the mixing tube used. These effects were thought to be partially responsible for the difference between predicted and measured performance for the complete system. Therefore, an additional objective of the present investigation was to test two alternative "low-drag" nozzle cluster designs and an extended mixing tube to determine whether these design changes would lead to significant performance improvements.

Section 2

.

• ••

. . .

. .

.

SYMBOLS

2.1	Symbols Used in the Analysis and Test Results
A	area, ft ²
Ā	area ratio = $A_{th}C_w/A_m$, dimensionless
С	gas specific heat at constant pressure, Btu/lbm-°F
C _w	nozzle flow coefficient, dimensionless
d	duct diameters, inches
f	friction coefficient, dimensionless
g _o	dimensional constant, 32.2 lbm-ft/lbf-sec 2
Н	stagnation enthalpy, Btu/lbm
J	conversion factor, 778.2 ft-lbf/Btu
k	gas specific heat ratio, dimensionless
K	duct loss coefficient , dimensionless
\mathbf{L}	duct length, inches
m	entrainment ratio = W_s/W_p , dimensionless
^m max	maximum entrainment ratio, dimensionless
М	Mach number, dimensionless
Р	pressure, psfa or psia
P	pressure ratio = P_{po}/P_{so} , dimensionless
P*	pressure ratio = P_m/P_{so} , dimensionless
q	dynamic head = $\rho V^2 / 2 g_0$, psfa
$\overline{\mathbf{q}}$	dimensionless dynamic head = $\rho V^2 / 2 g_0 P_{so}$,
r	radius, inches
r _o	outer radius of mixing tube cross section, inches
R	gas constant for air = 53.35 ft-lbf/lbm-°R

Т	temperature, °R
T	temperature ratio = T_{po}/T_{so} , dimensionless
Π	system thrust augmentation = $W_m V_b / W_p V_{p2}$, dimensionless
v	velocity, ft/sec
\overline{v}	velocity ratio = V_{s2}/V_{p2} , dimensionless
V*	velocity ratio = V_m/V_{p2} , dimensionless
W	mass flow rate, lbm/min
α	parameter defined by equation (34)
β	parameter defined by equation (35)
9	parameter defined by equation (45)
δ	parameter defined by equation (46)
ΔΡ	pressure change, psf
ΔP_{ext}	ambient pressure rise imposed upon jet pump system = $P_b - P_{ao}, \ psf$
ΔP_s	static pressure rise = $P_m - P_{so}$, psf
ΔP_s^*	dimensionless static pressure rise = $(P_m - P_{so})/P_{so}$,
ΔP_t	stagnation pressure rise = $P_{m0} - P_{s0}$, psf
ΔP_t^*	dimensionless stagnation pressure rise = $(P_{mo} - P_{so})/P_{so}$
ρ	density, lbm/ft ³
au	thrust augmentation = $W_m V_m / W_p V_{p2}$, dimensionless
$\tau_{\mathbf{w}}$	wall shearing stress, psi
arphi	impulse function

- ----

V

I _

Subscripts

a	atmospheric condition
b	blowing duct exit section
d	original conical diffuser exit section
е	test rig diffuser exit section
i	suction duct inlet section
l	overall duct loss coefficient
m	section at end of mixing region
MT	mixing tube
0	stagnation value
р	primary stream variable at nozzle exit
S	secondary stream variable at nozzle exit
SD	section duct
th	primary nozzle throat
1	section at primary nozzle exit
2	section at end of accommodation region

Superscripts

()'	value of parameter at end of frictional mixing tube extension (Section 3.1.5)
Ō	"mass-momentum" averaged value from test results

Formulation Name	Computer Name	Definition and units
Constants		
g _o	G0	dimensional constant, 32.2 ft-1bm 1bf-sec ²
J	CONV	conversion factor, 778 ft-lbf/Btu
R	R	gas constant, ft-lbf/lbm-°R
Variables		
Ā	ABAR1	$A_{th} C_w / A_m$, dimensionless
$\mathtt{A}_{ ext{th}}$	ATH	nozzle throat area, ${\rm ft}^2$
A _p	AP	nozzle exit area, ft^2
A _m	AM	mixing tube area, ft^2
Not Used	ABAR2	$\mathbf{A}_{\mathbf{p}}/\mathbf{A}_{\mathbf{m}}$, dimensionless
c_w	CW	nozzle flow coefficient, dimensionless
C_{p2}	CP2	primary specific heat (C $_{p}$), location 2, Btu/lbm-°R
C _{s2}	CS2	secondary specific heat (C_p), location 2, Btu/lbm-°R
C _m	CM	specific heat (C_p) at location m, Btu/lbm-°R
H _{p2}	HP2	primary stagnation enthalpy, station 2, Btu/lbm
H _{s2}	HS2	secondary stagnation enthalpy.station 2, Btu/lbm
К _{sd}	FDUCT	suction duct friction coefficient
K _{mt}	FTUBE	mixing tube friction coefficient
M_{p2}	РМОК	primary Mach Number, location 2, dimensionless

2.2 Symbols Used in the Computer Program

ġ

Formulation <u>Name</u>	Computer <u>Name</u>	Definition and units
M _{s2}	SMOK	secondary Mach Number, location 2, dimensionless
M _m	EMOK	Mach Number at location m, dimensionless
m	ENTR	entrainment ratio, dimensionless
P	PBAR	P_{po}/P_{so} , dimensionless
P _{po}	PPO	primary stagnation process, psi
P soi	PSOI	secondary stagnation pressure at duct inlet, psi
Pso	PSO	secondary stagnation pressure at nozzle exit, psi
P _m	PM	static pressure at location m, psi
P mo	PMTOT	stagnation pressure at location m, psi
$\overline{\mathrm{T}}$	TBAR	T_{po}/T_{so} , dimensionless
т _{ро}	TPO	primary stagnation temperature, °R
T _{so}	TSO	secondary stagnation temperature, °R
T_{p2}	TP2	primary temperature, location 2, $^{\circ}\mathrm{R}$
T_{s2}	TS2	secondary temperature, location 2, °R
T _m	TM	temperature at location m, °R
V _{p2}	VP2	primary velocity, location 2, ft/sec
v_{s2}	VS2	secondary velocity, location 2, ft/sec
V _m	VM	velocity at location m, ft/sec
Wp	WP	primary mass flow rate, lbm/min
Ws	WS	secondary mass flow rate, lbm/min
W _m	WM	total mass flow rate, $W_p + W_s$, lbm/min
$P_{mo} - P_{so}$	DELP	stagnation pressure rise, psi

.

....

Formulation Name	Computer Name	Definition and Units
P _{mo} - P _{so} P _{so}	DDELP	dimensionless stagnation pressure rise
P _m - P _{so}	DSTAT	static pressure change, psi
$\frac{P_{m} - P_{so}}{P_{so}}$	DDSTAT	dimensionless pressure change
$\frac{\rho_{\rm m}}{2{\rm g}_{\rm o}} \frac{{\rm V_m}^2}{2{\rm g}_{\rm o}}$	ENER	kinetic energy at location m, $1bf/ft^2$
$\frac{\rho_{\rm m} {\rm V_m}^2}{2 {\rm g_o} {\rm P_{so}}}$	DENER	dimensionless kinetic energy
τ	AUG	$W_m V_m / W_p V_{p2}$, momentum ratio, dimensionless
	Arrays	
	CHAR1	store solutions for dimensional table
	CHAR2	store solutions for dimensionless table
	COUNT	store number of solutions for plotting
	ENT	store entrainment ratio for plotting
	PRISE	store DDELP values for plotting
	PTITLE	store title for plotting
	SUM	store number of solutions for table
	THROS	store AUG values for plotting
	TTITLE	store title for plotting

Section 3

ANA LYSIS

3.1 Formulation of the Mathematical Model

In this section an analytical model is developed to predict the flow behavior in a compressible flow jet pump with a constant-area mixing tube. The analysis is intended to provide a complete description of the important flow parameters at specific locations within the jet pump and to describe the overall operation of the jet pump as an entrainment and thrust augmentation device. The analysis was prepared for air-to-air jet pumps. The parameters used in the analysis are listed in section 2. The geometrical parameters are shown in figure 2.

The fundamental purpose of the analytical model is to develop the performance characteristics of the jet pump directly. These performance characteristics can be represented by plots of jet pump pressure rise and momentum ratio as functions of entrainment ratio for a number of values of primary jet pressures and temperatures and for various area ratios. The performance characteristics in this form are analogous to head vs. capacity curves or performance maps which are commonly used for pumps and compressors.

The equations describing jet pump performance and flow behavior include the entrainment ratio as an independent parameter. The assumption of a particular value for the entrainment ratio (together with the inlet flow pressures and temperatures and the primary-to-secondary flow area ratio) allows calculation of all of the performance and flow parameters for that operating point. Then another value is assumed for the entrainment ratio and the calculation procedure is repeated. Successive points on the jet pump performance curves are determined in this way until the complete curve is traced out. In the present calculations, the entrainment ratios were limited arbitrarily to the range from 10 to 40. In some cases, the Mach number of the flow in the accommodation region reached 1.0 for an entrainment ratio less than 40. Higher entrainment ratios cannot be achieved in such cases because the constant-area mixing tube chokes when the secondary flow Mach number reaches 1.0.

3.1.1 Assumptions Used in the Analysis

The following assumptions are made to simplify the analysis without seriously compromising its accuracy:

- The values of specific heat at constant pressure (C) and the specific heat ratio (k) are expressed as functions of temperature; otherwise the gas is considered to be a perfect gas.
- 2. Wall shear forces are assumed to be negligible when compared to the pressure forces and the momentum of the primary and secondary streams. (This assumption is reviewed in section 3.1.4.)
- 3. No heat is transferred across the wall of the jet pump.
- 4. The mixing tube is assumed to have a constant crosssectional area along its entire length.
- 5. When the primary nozzle is operated at an off-design pressure ratio, the primary jet is assumed to expand or contract isentropically until the primary and secondary streams have equal static pressures. This adjustment process is assumed to take place in the accommodation region between sections 1 and 2 (see figure 2) and is assumed to be completed before any mixing takes place between the two streams.
- 6. The stagnation temperature of the primary flow is assumed to be sufficiently high that moisture condensation shocks do not occur as the flow expands.

3.1.2 Analysis of the Accommodation Region

The geometrical parameters and flow conditions in the jet pump are defined as shown in figure 2. The primary stream enters the accommodation region as a very high velocity jet; its Mach number may be as high as 3.5. The large momentum of the primary jet induces a secondary flow. In the region defined as the accommodation region, it is assumed that the primary and secondary jets do not mix, but the primary jet expands or contracts until its static pressure matches that of the secondary stream. At the point where the static pressures are equal, denoted as section 2, the accommodation process is assumed to be complete and the flows are parallel. This accommodation process is generally accompanied by a series of oblique expansion and contraction shock waves as the primary flow area adjusts to match the local static pressure outside the jet. However, if the jet pump is operated close to its design conditions, the degree of accommodation is small and the losses caused by the shock waves will be small. A simplified oblique shock analysis indicates that a nozzle designed for 350 psia supply pressure can be operated down to 200 psia with a total pressure loss due to shock waves of only 3%. For the values of supply pressure to be considered here, the error introduced by treating both streams as isentropic flows is negligible. In fact, one of the aims of this research was to show that an assumption of isentropic flow during the accommodation process will produce good results even when the system is operated at conditions quite far from the design point.

As long as the ratio of (P_{po}/P_{so}) is sufficiently high to guarantee a supersonic primary flow, as in the cases being considered here, the primary mass flow rate may be calculated directly using equation (1).

$$W_{p} = \sqrt{\frac{k g_{0}}{R}} \left(\frac{2}{k+1}\right)^{\frac{k+1}{k-1}} \frac{\frac{144 \times P_{po} (60 \times A_{th} C_{w})}{\sqrt{T_{po}}}$$
(1)

By specifying an entrainment ratio, $m = W_s/W_p$, the secondary stream mass flow rate can also be found directly.

$$W_{s} = M W_{p}$$
 (2)

In a perfect gas, the local values of total and static pressure are related to the local Mach number by the following equation.

$$P_0/P = (1 + \frac{k-1}{2} M^2)^{-\frac{k}{k-1}}$$
 (3)

At the end of the accommodation region, the static pressures of the primary and secondary streams are equal. Therefore, at the end of the accommodation region the following relation must be satisfied.

$$P_{po}/(1+\frac{k-1}{2}M_{p2}^2) = P_{so}/(1+\frac{k-1}{2}M_{s2}^2)^{\frac{k}{k-1}}$$
 (4)

The mass flow rate per unit area for an isentropic flow is given by the relation below:

$$W/A = \sqrt{\frac{kg_{0}}{R}} \sqrt{\frac{P_{0}}{T_{0}}} \left\{ \frac{M}{\left(1 + \frac{k-1}{2} M^{2}\right)^{\frac{k+1}{2(k-1)}}} \right\}$$
(5)

The geometry of the constant area mixing tube requires that $A_{p2} + A_{s2} = A_m$. Using equation (5) to represent A_{p2} and A_{s2} , and inserting the appropriate unit conversion factors, the geometry condition becomes as follows:

$$\frac{W_{p}}{144 \times 60 \sqrt{\frac{kg_{0}}{R}}} \left\{ \sqrt{\frac{T_{po} (1 + \frac{k-1}{2} M_{p2}^{2})^{2}}{\frac{P_{po} M_{p2}}{P_{po} M_{p2}}}} + \frac{m\sqrt{T_{so}} (1 + \frac{k-1}{2} M_{s2}^{2})^{\frac{k+1}{2(k-1)}}}{\frac{P_{so} M_{p2}}{P_{so} M_{p2}}} \right\} = A_{m}$$
(6)

When W_p , m, P_p , P_{so} , T_{po} , T_{so} and A_m are specified equations (4) and (6) can be solved simultaneously to obtain M_{p2} and M_{s2} . Equation (5) can be used to find A_{p2} and A_{s2} .

Since the flow of both the primary and secondary streams is assumed to be isentropic in the accommodation region, values of static pressure, temperature, and velocity can be obtained for each stream at location (2) by employing the following equations.

$$P_{p2} = P_{p0} / (1 + \frac{k-1}{2} M_{p2}^2)^{\frac{k}{k-1}}$$
(7)

$$T_{p2} = T_{p0} / (1 + \frac{k-1}{2} M_{p2}^2)$$
 (8)

$$T_{s2} = T_{s0} / (1 + \frac{k-1}{2} M_{s2}^2)$$
 (9)

$$V_{p2} = \sqrt{\frac{2 \, k \, g_0}{k-1}} R (T_{p0} - T_{p2})$$
(10)

$$V_{s2} = \sqrt{\frac{2 k g_0}{k-1}} R (T_{s0} - T_{s2})$$
 (11)

3.1.3 Analysis of the Mixing Region

The primary and secondary streams enter the mixing region with equal static pressures and parallel velocities. In this region, complete mixing takes place and a uniform flow with constant properties across the channel is obtained at section m. Treating the mixing region as a control volume with completely specified entering flows, the following equations can be applied.

Continuity Equation:
$$W_m = W_p (1 + m)$$
 (12)

Mass Flow Rate: $W_m = 60 \times \rho_m V_m A_m$ (13)

Equation of State:
$$P_m = \rho_m R T_m / 144$$
 (14)

Momentum Equation:

$$144 \times 60 \times g_0 A_m (P_{s2} - P_m) = W_m V_m - W_p V_{p2} - W_s V_{s2}$$
(15)

Energy Equation:

1939

$$W_{p}H_{p2} + W_{s}H_{s2} = W_{m}C_{m}\left(T_{m} + \frac{V_{m}^{2}}{2g_{0}JC_{m}}\right)$$
 (16)

where H is stagnation enthalpy;

$$H_{p2} = C_{p2} T_{p2} + \frac{V_{p2}^{2}}{2g_{0}J}$$
$$H_{s2} = C_{s2} T_{s2} + \frac{V_{s2}^{2}}{2g_{0}J}$$

The equation for the specific heat at constant pressure for air as given in the <u>Gas Turbine Engineering Handbook</u>, Gas Turbine Publications, Inc., 1966, page 4, is presented below.

$$C_i = .24916 - .482 \times 10^{-4} T_i + .681 \times 10^{-7} T_i^2$$
 (17)

Equations (13) and (14) are combined to give

$$T_{m} = 60 \times 144 \times (P_{m} V_{m} A_{m} / R W_{m})$$
(18)

Equation (15) is written in the form

$$V_{m} = \frac{W_{p}}{W_{m}} \left[V_{p2} + m V_{s2} + 60 \times 144 \times \frac{g_{o} A_{m}}{W_{p}} (P_{s2} - P_{m}) \right]$$
(19)

and equations (16) and (18) are combined to yield

$$P_{m} = \frac{RW_{p}}{60 \times 144 \times V_{m} A_{m} C_{m}} \left[H_{p2} + m H_{s2}\right] - \frac{RW_{m} V_{m}}{2 \times 60 \times 144 \times g_{o} J C_{m} A_{m}}$$
(20)

Since the values of T_{p2} and T_{s2} are known at the inlet to the mixing region, C_{p2} and C_{s2} can be evaluated directly using equation (17). Equations (12),

 $\mathbf{17}$

(17), (18), (19), and (20) represent five equations for the five unknown exit parameters, W_m , T_m , C_m , P_m and V_m . Therefore, these five equations represent a complete set which describe the properties of the flow leaving the mixing section. In most instances, T_{p2} and T_{s2} are sufficiently close in value to permit the use of a constant specific heat, $C_{p2} = C_{s2} = C_m$. This simplifies the solution by removing equation (17) from the equation set.

3.1.4 Jet Pump Performance Parameters

The jet pump performance parameters which are of particular interest are the stagnation pressure rise and the momentum augmentation. These parameters can be evaluated by using the following expressions.

$$\Delta P_{t} = P_{mo} - P_{so}$$

$$\Delta P_{t} = P_{m} (1 + \frac{k-1}{2} M_{m}^{2}) - P_{so}$$
(21)

and

 τ = outlet momentum / primary momentum

$$r = \frac{W_{m}V_{m}}{W_{p}V_{p2}} = \frac{(m+1)V_{m}}{V_{p2}}$$
(22)

During the jet pump test program, the measurement of the stagnation pressure rise produced by the jet pump proved to be difficult to accomplish. It was much easier to measure the stagnation-to-static pressure rise in the jet pump. This pressure rise can be used to define another jet pump performance parameter, ΔP_{e} , which serves as an alternative to the stagnation pressure rise parameter, ΔP_{t} .

$$\Delta P_{\rm s} = P_{\rm m} - P_{\rm so} \tag{23}$$

All of the values needed to compute ΔP_t , ΔP_s , and τ are provided by the analyses of sections 3.1.2 and 3.1.3.

3.1.5 Inclusion of Mixing Tube Wall Friction

The analysis of the mixing region presented in section 3.1.3 neglected the effect of wall friction; this simplifies the equations describing the mixing process. In this section, a procedure is developed to include the effects of wall friction in the jet pump analysis.

The mixing region was assumed to extend from the point where the primary and secondary stream pressures are equal to the point where they have merged into a uniform flow with constant properties across the channel. In reality, the wall friction effects occur in conjunction with the mixing process. Unfortunately, it is difficult with the current state of knowledge to predict wall friction losses accurately in the mixing region. Therefore, rather than adjust the mixing region analysis to include the wall friction effects, we considered that it would be preferable at this time to treat the mixing process and wall friction as independent effects by imagining the mixing tube to extend as shown in figure 3 beyond the point where the mixing process is complete. The flow phenomena occurring in the mixing portion, segment I, is a mixing process without wall friction as analyzed in section 3.1.3. The frictional portion, segment II, represents the effect of wall shear forces upon a uniform adiabatic flow. The hypothetical extension of the mixing tube is meant to represent the friction occurring within the actual mixing tube.

The effect of wall shear forces in ducts is commonly represented by a coefficient of friction defined as follows:

$$\mathbf{f} = \frac{\tau_{\mathbf{w}}}{\rho \, \mathbf{v}^2 / 2 \, \mathbf{g}_0} \tag{24}$$

where $\tau_{\rm W}$ is the shearing stress exerted upon the stream by the wall. The corresponding stagnation pressure loss is given by equation (25).

$$\Delta P_{0} = 4 f \frac{L}{D} \frac{\rho V^{2}}{2g_{0}}$$
(25)

where

$$L = duct length$$

D = duct diameter

In many flow analyses, the value of the coefficient of friction is taken from test results for pipe flow. The friction coefficient is a function of wall roughness and Reynolds number. However, in the case of the mixing tube, the wall friction occurs in a very non-uniform flow which has a high level of turbulence. The value of the coefficient of friction for the mixing tube cannot be accurately determined from pipe flow data. Therefore, it is convenient to represent the mixing tube wall friction loss in terms of a head loss factor, K_{MT}, which must be determined experimentally.

$$\Delta P_{o} \bigg)_{\text{mixing tube}} = K_{\text{MT}} \frac{\rho_{\text{m}} V_{\text{m}}^{2}}{2 g_{o}}$$
(26)

The equations for an adiabatic flow in a constant-area tube with a stagnation pressure loss are given below. Primed variables denote parameters at the end of the hypothetical extension of the mixing tube.

Momentum Equation:

$$144 P_{m} \left(1 + \frac{k-1}{2} M_{m}^{2}\right)^{\frac{k}{k-1}} - 144 P_{m'} \left(1 + \frac{k-1}{2} M_{m'}^{2}\right)^{\frac{k}{k-1}} = K_{MT} \frac{\rho_{m} V_{m}^{2}}{2g_{0}}$$
(27)

Energy Equation:

$$C_{m}T_{m} + \frac{V_{m}^{2}}{2g_{o}} = C_{m}T_{m}' + \frac{V_{m}'^{2}}{2g_{o}}$$
 (28)

Continuity Equation:

$$\rho_{\mathbf{m}} \mathbf{V}_{\mathbf{m}} = \rho_{\mathbf{m}}' \mathbf{V}_{\mathbf{m}}'$$
(29)

State Equation:

$$P_{m} = \rho_{m} R T_{m}$$
(30)

$$\mathbf{P}_{\mathbf{m}}' = \rho_{\mathbf{m}}' \mathbf{R} \mathbf{T}_{\mathbf{m}}' \tag{31}$$

Definition of Mach Number:

$$M_{m}^{2} = \frac{V_{m}^{2}}{g_{o}^{k} R T_{m}}$$
(32)

$$M_{m}'^{2} = \frac{V_{m}'^{2}}{g_{o} k R T_{m}'}$$
(33)

These equations can be combined to give the following results: k

$$P_{m'} \left(1 + \frac{k-1}{2} M_{m'}^{\prime} \right)^{\frac{K}{k-1}} = \alpha$$
 (34)

$$P_{m'}^{2} M_{m'}^{2} \left(1 + \frac{k-1}{2} M_{m'}^{2}\right) = \beta$$
(35)

where

$$\alpha = P_{m} \left\{ \left(1 + \frac{k-1}{2} M_{m}^{2} \right)^{-\frac{k}{k-1}} - K_{MT} - \frac{k M_{m}^{2}}{2} \right\} \quad (36)$$

$$\beta = P_{m}^{2} M_{m}^{2} \left(1 + \frac{k-1}{2} M_{m}^{2} \right)$$
(37)

Eliminating P_m' between equation (34) and equation (35) yields equation (38) which can be used to determine M_m' .

$$\frac{1}{M_{m'}} \left(1 + \frac{k-1}{2} M_{m'}^{2}\right)^{\frac{k+1}{2(k-1)}} = \alpha/\sqrt{\beta}$$
(38)

After equation (38) is solved for M_{m}' , equation (34) can be used to determine P_{m}' . Combining equations (29), (30), (31), (32), and (33), an equation for T_{m}' is obtained.

$$T_{m'} = T_{m} (P_{m'}/P_{m})^{2} (M_{m'}/M_{m})^{2}$$
 (39)

Equation (33) is then used to determine V_m' .

The equations developed in this section can be used to compute the values of P_m' , T_m' , M_m' , and V_m' for a jet pump when the "ideal" analysis of sections 3.1.2 and 3.1.3 is completed and the value of K_{MT} is known or assumed. Alternatively, these equations may be used to deduce the value of K_{MT} when values of P_m' , T_m' , and V_m' are known from test results and values of P_m , T_m , and V_m are computed by using the "ideal" analysis.

3.1.6 Dimensionless Formulation

In this section the equations describing the jet pump operation are formulated in terms of dimensionless variables. The non-dimensional formulation is valuable for two reasons:

- According to the principles of dimensional analysis, a solution in terms of independent non-dimensional groups is a general one. The same solution may be applied for jet pumps having great differences in individual design or operating parameters so long as the independent non-dimensional groups are identical. For example, one such group is the primary-to-secondary flow area ratio, A_{th} C_w/A_m; if all other non-dimensional groups are the same, a large-scale and small-scale jet pump having identical area ratios will have identical non-dimensional performance characteristics.
- The non-dimensional formulation permits identification of the minimum number of independent non-dimensional groups

which are required to completely specify a jet pump design and its operating characteristics.

In the derivation which follows, unit conversion factors are not included, and all values of the specific heats (C_p, C_s, C_m) are assumed to be constant and equal.

The equations which apply in the mixing region are given below.

- Continuity Equation: $W_m = W_p (1 + m)$ (12)
- Mass Flow Rate: $W_m = \rho_m V_m A_m$ (40)

Equation of State:
$$P_m = \rho_m R T_m$$
 (41)

Momentum Equation:

Y

$$g_{0} A_{m} (P_{s2} - P_{m}) = W_{m} V_{m} - W_{p} V_{p2} - m W_{p} V_{s2}$$
 (42)

Energy Equation:

$$T_{po} + m T_{so} = (m + 1) C \left[T_m + \frac{V_m^2}{2g_o C} \right]$$
 (43)

The dimensionless variables to be used are defined as follows.

$$V^* = V_m / V_{p2}; P^* = P_m / P_{so}, \overline{P} = P_{po} / P_{so},$$

$$\overline{T} = T_{po} / T_{so}, \overline{A} = A_{th} C_w / A_m$$
(44)

Using these variables, equations (12), (40), (41), (42), and (43) can be combined as follows.

$$P^* = (P_{s2}/P_{s0}) + \gamma \left[1 + m (V_{s2}/V_{p2}) - (m+1) V^*\right]$$
(45)

$$V^{*2} + \frac{k}{(k-1)} \frac{2 P^{*} V^{*}}{\gamma (m+1)} - \frac{\delta}{(m+1)} (1 + \frac{m}{T}) = 0$$
(46)

where

$$\gamma = W_{\rm p} V_{\rm p2} / g_{\rm o} A_{\rm m} P_{\rm so}$$
⁽⁴⁷⁾

and

$$\delta = 2 g_0 C T_{p0} / V_{p2}^{2}$$
(48)

Equations (45) and (46) show that P* and V* are related through γ , δ , $P_{s2}/P_{so},~V_{s2}/V_{p2},$ m, k, and $\overline{T}.$

The equations which govern the flow in the accommodation region are developed next. For an isentropic primary stream,

$$T_{p2}/T_{po} = (P_{p2}/P_{po})^{\frac{k-1}{k}} = \begin{bmatrix} \frac{1}{\bar{P}} & \frac{P_{s2}}{P_{so}} \end{bmatrix}^{\frac{k-1}{k}}$$
 (49)

$$W_{p} = \sqrt{\frac{k g_{o}}{R} \left(\frac{2}{k+1}\right)} \sqrt{\frac{p_{o}}{\sqrt{T_{po}}}} (A_{th} C_{w})$$
(50)

and

$$V_{p2} = \sqrt{\frac{2 g_{o} k R}{k-1}} T_{po} (1 - T_{p2}/T_{po})$$
(51)

Equations (49), (50), and (51) can be combined to give

$$\gamma = \overline{P} \sqrt{\frac{2k^2}{k-1}} \left(\frac{2}{k+1}\right) \qquad \overline{A} \sqrt{1 - \left[\frac{1}{\overline{P}} \left(\frac{P_{s2}}{P_{s0}}\right)\right]} \qquad (52)$$

Similarly, equations (44) and (51) can be used to obtain a value for δ :

$$\delta = \frac{1}{2} \left\{ 1 - \left[\frac{1}{\overline{P}} \left(\frac{P_{s2}}{P_{s0}} \right) \right]^{\frac{k-1}{k}} \right\}$$
(53)

Thus, γ and δ are shown to be functions of $\bar{P},\,\bar{A},\,k$ and $P_{s2}^{}/P_{so}^{}$

The secondary stream velocity is given by equation (54):

$$V_{s2} = \sqrt{\frac{2 g_{o} k R}{k-1}} T_{so} \left(1 - \frac{T_{s2}}{T_{so}}\right)$$
(54)

Equations (49), (51), and (54) can be combined to yield

$$V_{s2}/V_{p2} = \sqrt{\frac{1}{\overline{T}}} \sqrt{\frac{1 - \left(\frac{P_{s2}}{P_{s0}}\right)^{\frac{k-1}{k}}}{1 - \left[\frac{1}{\overline{P}}\left(\frac{P_{s2}}{P_{s0}}\right)\right]^{\frac{k-1}{k}}}}$$
(55)

From the definition of total or stagnation pressure,

$$P_{s2}/P_{so} = 1 / (1 + \frac{k-1}{2} M_{s2}^2)^{\frac{k}{k-1}}$$
 (56)

Considering equations (51), (53), (55), and (56) together, it can be seen that the parameters γ , δ , V_{s2}/V_{p2} , and P_{s2}/P_{s0} are functions of M_{s2} , \overline{P} , \overline{T} , \overline{A} , m and k only. Equation (5) of section 3.1.2 can be written for the secondary stream as follows.

$$\frac{M_{s2}}{\left(1 + \frac{k-1}{2} M_{s2}^2\right)^{\frac{k+1}{2(k-1)}}} = \frac{W_s \sqrt{T_{so}}}{P_{so} \sqrt{\frac{k g_o}{R}} (A_m - A_p)}$$
(57)

Equation (57) can be combined with equation (50) and written in terms of dimensionless variables as given below.

$$\frac{M_{s2}}{(1+\frac{k-1}{2}M_{s2}^{2})} = \frac{m (\frac{2}{k+1})}{\sqrt{T} (1-A_{p}/A_{m})}$$
(58)

1_ 1 -1

For high entrainment ratio jet pumps, the term A_p/A_m is of order 0.01. Thus, the term $(1 - A_p/A_m)$ can be approximated as unity. Equation (58) can now be written as follows.

$$\frac{M_{s2}}{(1+\frac{k-1}{2}M_{s2}^2)} = m \left(\frac{2}{k+1}\right) \frac{\frac{k+1}{2(k-1)}}{\sqrt{T}}$$
(59)

Equation (59) shows that M_{s2} can be determined from m, \overline{P} , \overline{A} , \overline{T} , and k. This indicates that γ , δ , V_{s2}/V_{p2} , and P_{s2}/P_{so} are functions of \overline{P} , \overline{T} , \overline{A} , k and m. Returning to equations (45) and (46), it can be seen that the performance of the jet pump depends upon the parameters \overline{P} , \overline{T} , \overline{A} , k and m. For given values of \overline{P} , \overline{T} , \overline{A} and k, a complete dimensionless solution can be obtained for each specified value of entrainment ratio.

The jet pump performance parameters can be experssed in the form of dimensionless groups using the fundamental dimensionless parameters. The momentum augmentation, τ , is already a dimensionless group:

$$\tau = \frac{W_{m} V_{m}}{W_{p} V_{p2}} = (m+1) V^{*}$$
(60)

The dimensionless stagnation pressure rise parameter is defined as follows:

$$\Delta P_t^* = \frac{\Delta P_t}{P_{so}} = P^* (1 + \frac{k-1}{2} M_m^2) - 1$$
 (61)

To evaluate ΔP_t^* , it is necessary to know M_m^2 . Using the dimensionless variables, M_m^2 is given by equation (62).

$$M_{m}^{2} = \frac{V_{m}^{2}}{k g_{o} R T_{m}} = \frac{V^{*}}{P^{*}} (\frac{m+1}{k}) \gamma$$
 (62)

Thus, ΔP_t^* is a function of the fundamental dimensionless variables which determine V*, P*, and γ . The static pressure rise parameter, ΔP_s , can be expressed in dimensionless form as follows:

$$\Delta P_{s}^{*} = \frac{P_{m}^{-} P_{so}^{-}}{P_{so}^{-}} = P^{*} - 1$$
 (63)

Equations (60) through (63) show that the dimensionless jet pump performance parameters are functions of the fundamental independent dimensionless variables \overline{P} , \overline{T} , \overline{A} , k, and m. Five such independent variables and only five have to be specified in order to determine the jet pump performance characteristics in dimensionless terms. (This conclusion is restricted to jet pumps which satisfy the assumptions listed in section 3. 1. 1 and the additional assumption that the specific heats of all of the streams are equal, i.e., $C = C_{p2} = C_{s2} = C_m$.)

It is possible to use a different set of five independent dimensionless variables. For example, a velocity ratio $\overline{V} = V_{s2}/V_{p2}$ can be used in place of \overline{T} to complete an alternative set of five independent variables, \overline{P} , \overline{V} , \overline{A} , k and m. Another possible set is \overline{P} , \overline{T} , \overline{A} , k, and \overline{V} . The velocity ratio, \overline{V} , was one of the basic design parameters used to select the jet pump geometry for the boundary layer control system tested under contract No. NAS 2-2518.

These remarks can be summarized by the expressions below:

Jet Pump Performance	Design and Operating
Characteristics	Conditions
(dependent variables)	(independent variables)
ΔP_t^* , ΔP_s^* , and τ are functions of	$\left\{\begin{array}{c} \overline{P}, \overline{T}, \overline{A}, k, m\\ or\\ \overline{P}, \overline{V}, \overline{A}, k, m\\ \overline{P}, \overline{T}, \overline{A}, k, \overline{V}\end{array}\right.$

3.1.7 Maximum Entrainment Ratio

Equation (59) can be used to determine the maximum possible entrainment ratio. This maximum occurs when the secondary stream Mach number reaches a value of unity. Setting $M_{s,2} = 1$, equation (59) can be written as follows.

$$m_{\max} = \frac{\sqrt{\overline{T}'}}{\overline{P}\,\overline{A}} \tag{64}$$

Thus, the maximum entrainment ratio corresponding to the choking of the secondary stream can be determined directly from the dimensionless initial conditions.

3.2 The Computer Program

A computer program was prepared to predict the performance characteristics of constant area jet pumps using the analytical concepts formulated in the preceding sections. The program was written to develop both dimensional and dimensionless solutions. Values of \overline{P} , \overline{T} , \overline{A} , P_{so} , T_{so} , and A_m are read in as initial conditions. Values of k, R, and m are included within the program. For each value of m, values of τ , ΔP_t , ΔP_s , ΔP_t^* , and ΔP_s^* are calculated. The values of τ , ΔP_t^* , and ΔP_s^* depend only on the dimensionless data, while ΔP_t and ΔP_s depend also upon P_{so} , T_{so} , and A_m .

The program is written in Fortran IV language. The machine used was an IBM System 360/65 with an SC4020 plotter. Automatic plotting of the performance characteristics was obtained by using the subroutine EZPLOT developed by the Missile Systems Division of Avco Corporation, Burlington, Massachusetts.

A block diagram of the computer program is shown in figure 4. For each set of initial conditions, solutions are obtained for values of entrainment ratio between 10.0 and 40.0 in steps of 3.0. The results are printed as each solution corresponding to a particular value of entrainment ratio is determined. The results are also stored in arrays for plotting and for presentation in tabular form. A printout of the entire program is presented in Appendix A. A discussion of the program by blocks is given in Appendix B. Appendix C provides typical computer solutions which indicate the form of the output data. Certain of the blocks shown in figure 4 and described in Appendix B are denoted as optional, indicating that they can be removed from the program without interfering with the operation of the remaining blocks. Instructions for removing these blocks are given in Appendix B, section B-2.

When frictional effects in the suction duct and mixing tube are to be taken into account in the performance predictions, values of the loss coefficients K_{sd} (defined in Appendix B) and K_{MT} must be provided as input data for the computer solution. These loss coefficients are functions of the flow Reynolds numbers. If values of K_{sd} and K_{MT} have been established for ducts of one size and the computer performance predictions are to be used for ducts of much larger or smaller sizes, it may be advisable to adjust the values of the loss coefficients used by the computer to account for the Reynolds number change.

3.3 Solutions for a Range of Jet Pump Designs

1Ę

The computer program was applied to develop jet pump performance plots for a broad range of geometries and operating conditions. The range of solutions was selected to encompass all of the test conditions used in this investigation (section 4) and also the range of conditions of interest to NASA for boundary layer control systems and momentum augmentation. The performance plots were developed for use in preliminary design of jet pump systems for matching the jet pump to a duct system and for predicting the resulting system performance characteristics. Techniques for applying the solutions to system design are described in section 3.4.

The range of conditions used to obtain the performance plots were initially defined in dimensional form as follows:

po	0 f 10 3500 f
P _{po} primary flow stagnation pressure 10	0 psia to 400 psia
T _{so} secondary flow stagnation temperature 20	° F to 120° F
P _{so} secondary flow stagnation temperature 15	00 psfa to 2116 psfa

The range of values selected for the corresponding dimensionless parameters were as follows:

$$\overline{T} = 1.5 \text{ to } 8.0$$

 $\overline{P} = 5 \text{ to } 40$

The nozzle and mixing tube geometries available for the test program had area ratio values (\overline{A}) ranging from 0.00125 to 0.0067. The range of values selected for the performance plots is given below:

$$\overline{A} = 0.001 \text{ to } 0.007$$

The ranges of values of \overline{P} , \overline{T} , and \overline{A} given above were used to prepare 9 sets of performance plots showing ΔP_t^* vs. m and τ vs. m for various values of \overline{P} with \overline{T} and \overline{A} fixed; these plots are indexed in table 1 and given in figures 6 through 23. Table 1 includes the values of maximum entrainment ratio attainable for each combination of \overline{P} , \overline{T} , and \overline{A} values. This maximum entrainment ratio is set by choking of the secondary stream as given by equation (64) of section 3.1.7.

Typical computer output sheets for one of the solutions are reproduced in Appendix C. The printed output includes values of jet pump parameters not shown in the plots but required for the jet pump-duct matching techniques described in section 3.4. These parameters are given in dimensional form based upon standard secondary stream inlet conditions, $P_{so} = 2102 \text{ psfa}$, $T_{so} = 70 \text{ }^{\circ}\text{F}$. and $A_m = 0.08726 \text{ ft}^2$.

In the last six cases, the higher values of \overline{P} cannot be attained because of choking of the flow in the mixing tube. For the cases with $\overline{T} = 1.5$ and 3.5, choking occurs in the secondary flow ($M_{s2} = 1$) as discussed in section 3.1.7. When \overline{T} was set at 8.0, choking was predicted to occur first at the mixing tube exit, i.e., $M_m = 1$.

Cross-plots showing ΔP_t^* vs. m and τ vs. m for various values of \overline{A} with \overline{T} and \overline{P} fixed are presented in figures 24 and 25. Lines of constant mixing tube exit Mach number are also shown. These cross-plots provide additional insight on the effect of the area ratio upon jet pump performance.

3.4 Jet Pump-Duct Matching Considerations

Ż

The previous sections have developed analytical techniques and a computer program which allow prediction of the performance characteristics of highentrainment-ratio jet pumps. The performance characteristics take the form of plots of jet pump pressure rise and momentum ratio as functions of entrainment ratio for a number of values of primary jet pressures and temperatures and for various area ratios. These performance characteristics are analogous to head vs. capacity curves or performance maps which are commonly used for pumps and compressors. The actual point (i.e., entrainment ratio) at which a jet pump will operate when connected to a particular system of inlet and discharge ducts is dictated by the geometry of the duct system.

The resistance curve of a duct system is roughly parabolic as shown in figure 26. A typical jet pump characteristic is also shown on the figure . The actual operating point of the jet pump-duct combination is defined by the intersection of the two curves. The duct characteristic curve is set by the duct geometry and is essentially independent of the jet pump operating conditions. Therefore, if the duct geometry is not changed, the operating point of the system for any jet pump primary flow condition must be located on the parabola. Figure 27 shows how the operating points for a system can be determined if the jet pump performance at various pressure levels is known.

This section establishes a procedure for use to determine the operating points of a jet pump in a duct system when the loss characteristics of the duct system are known. This procedure can be employed as shown by example to match the jet pump design to the duct system so as to achieve maximum entrainment ratio or maximum thrust augmentation for given primary flow conditions.

3.4.1 Representation of Duct Loss Characteristics

When the analytical model developed in sections 3.1 and 3.2 is supplied with values of \overline{P} , \overline{T} and \overline{A} , the performance parameters such as stagnation pressure rise and momentum ratio (thrust augmentation) can be calculated as a function of entrainment ratio. In order to determine the specific value of entrainment ratio which will

be obtained during operation of a given jet pump of fixed geometry, the associated duct system flow characteristics must be taken into account. Using the notation shown in figure 2, the stagnation pressure rise may be written as follows:

$$\Delta P_{t} = P_{mo} - P_{so} = (P_{ao} - P_{so}) + (P_{mo} - P_{bo}) + P_{bo} - P_{ao}$$
(65)

The two bracketed terms together represent the stagnation pressure loss of the duct system including the inlet duct (first term) and blowing duct (second term).

At low Mach numbers, the stagnation pressure loss due to friction in a duct is proportional to the kinetic energy of the flow. For high entrainment ratio jet pumps, the mass flow rates in the suction and blowing ducts are nearly equal. Therefore, the total pressure loss of the entire ducting system may be related as a first approximation to the kinetic energy of the blowing duct inlet flow by equation (66):

$$(P_{mo} - P_{bo}) + (P_{ao} - P_{so}) = K_{\ell} \frac{\rho_m V_m^2}{2 g_0}$$
 (66)

This type of expression has been shown to be accurate for representing frictional losses in duct systems of various shapes.

3.4.2 Evaluation of the Loss Coefficient, K

The loss coefficient K_{ℓ} depends on the geometry of the particular ducts being used and the Mach number level (ref.13). At sufficiently low Mach numbers, (i.e., under 0.3), compressibility effects can be neglected. For flows at higher Mach numbers, the value of K_{ℓ} can be corrected for compressibility effects.

Loss coefficients have been presented for a number of duct configurations in references 13 through 26. The configurations reported include ducts of rectangular and circular cross section with varying amounts of diffusion or acceleration. Bends and elbows having a number of different angles of turn are included in these references. The loss coefficients reported were measured for subsonic flow covering a range of Mach numbers up to 1.0.
To provide an example of typical loss data, the influence of Mach number level upon the loss coefficient for straight, conical diffusers is shown in figure 28. The variation of K_{l} with inlet Mach number is significant for a diffuser of specified geometry.

Because the value of K_{l} is closely related to the duct configuration and the Mach number level, and values of K_{l} are readily available for only a few simple duct shapes, the designer of a jet pump and duct system generally will not be able to look up an accurate value for K_{l} for a new duct design. If optimum matching of the jet pump and ducting is required, the loss coefficient of a new duct geometry will have to be determined experimentally. Testing can be done by using either a fullscale or reduced scale model of the duct. The tests must cover the Mach number range which will be encountered by the actual duct when operating with the jet pump. Flow tests of ducts sometimes have additional value; regions of flow separation or undesirable velocity profiles may be revealed. When the duct geometry is modified to eliminate these problems, the loss coefficient is usually reduced.

3.4.3 Development of System Performance Equations

At the outlet of the blowing duct, the static pressure in the flow must be equal to the local "atmospheric" pressure. The use of a blowing duct having the same cross-sectional area as the constant-area mixing tube will limit the entrainment ratio which can be achieved in the jet pump system. Higher entrainment ratios can be obtained with the same jet pump if a diffuser is added to the blowing duct. The diffuser allows higher velocities and flow rates in the mixing tube. The mixing tube pressures can be sub-atmospheric; the diffuser decelerates the flow to increase its static pressure up to the atmospheric pressure level at the blowing duct exit.

A calculation method can be developed for use to determine the actual operating point (i.e., entrainment ratio) for a jet pump system as a function of the area ratio selected for the blowing duct diffuser. The calculation method makes use of the generalized jet pump performance characteristics developed by the computer program described in section 3.2 and Appendix B.

The loss coefficient defined in equation (66) is inserted in equation (65) with the following result:

$$\Delta P_t = (P_{bo} - P_{ao}) + K_{\ell} q_m$$
(67)

where

$$q_{\rm m} = \frac{\rho_{\rm m} V_{\rm m}^2}{2 g_{\rm o}^2}$$

Equation (67)may be rewritten as follows:

$$\Delta P_{t} = (P_{b} - P_{ao}) + K_{\ell} q_{m} + (P_{bo} - P_{b})$$
(68)

The term $(P_b - P_{ao})$ represents the external ambient pressure difference imposed upon the jet pump system. This term will be called ΔP_{ext} :

$$\Delta P_{ext} = (P_b - P_{ao}) \tag{69}$$

The value of ΔP_{ext} was zero for the experimental jet pump since both the discharge static pressure (P_b) and inlet stagnation pressure (P_{ao}) for the jet pump system were equal to atmospheric pressure. This term is not necessarily zero for jet pump systems which operate in the presence of an external velocity field. For example, a jet pump used for boundary layer control at the trailing edge of a wing will have its inlet pressure (P_{ao}) established by the flow behavior in the suction slot entry passages and by the local pressure acting on the wing. The discharge pressure (P_b) will be set by the local pressure field on the wing and by the flow behavior from the slot to the deflected flap.

The term $(P_{bo} - P_b)$ in equation (68) represents the dynamic head of the flow at the blowing duct exit. This term is related to the blowing duct exit Mach number as shown in equation (70): k

$$(P_{bo} - P_{b}) = P_{b} \left(\frac{P_{bo}}{P_{b}} - 1 \right) = P_{b} \left[\left(1 + \frac{k-1}{2} M_{b}^{2} \right)^{k-1} - 1 \right]$$
(70)

In order to evaluate this term, the value of M_b must be calculated using the selected blowing slot area A_b , the exit pressure P_b , and the flow conditions at the discharge of the jet pump mixing tube:

$$M_{b} \sqrt{1 + \frac{k-1}{2} M_{b}^{2}} = \frac{W_{m}}{A_{b}} \sqrt{\frac{T_{mo}}{P_{b}}} \sqrt{\frac{R}{k g_{o}}}$$
 (71)

This equation is based upon the assumption that the flow in the blowing duct is adiabatic. The values of T_{mo} and W_m are output values from the jet pump performance calculations described in the previous section. The evaluation of M_b can be made conveniently by using figure 29.

Equations (70) and (71) can be replaced by equation (72) when the blowing duct exit Mach number, $M_{\rm b}$, is less than 0.3.

$$P_{bo} - P_{b} = \frac{\rho_{b} V_{b}^{2}}{2 g_{o}}$$
 (72)

The error in using equation (72) in place of equations (70) and (71) is less than 2% of the true ($P_{bo} - P_{b}$) difference when M_{b} is less than 0.3.

A particularly simple jet pump-duct system matching equation can be derived when equation(72) is used. Equation(73) is the continuity relation for the blowing duct:

$$\rho_{\rm m} V_{\rm m} A_{\rm m} = \rho_{\rm b} V_{\rm b} A_{\rm b}$$
(73)

The influence of the blowing duct area ratio is introduced when equation(73) is combined with equation (72):

$$P_{bo} - P_{b} = q_{m} \left(\frac{A_{m}}{A_{b}}\right)^{2} \frac{\rho_{m}}{\rho_{b}}$$
(74)

The jet pump-duct matching equation, equation (75), is derived by combining equations (68), (69) and (74):

$$\Delta \mathbf{P}_{t} = \Delta \mathbf{P}_{ext} + q_{m} \left[K_{\ell} + \left(\frac{A_{m}}{A_{b}} \right)^{2} \frac{\rho_{m}}{\rho_{b}} \right]$$
(75)

~

1

 \mathbf{for}

 $M_b \leq 0.3$

For preliminary design purposes, the value of ρ_m/ρ_b can be taken as 1.0. A more accurate value can be determined as follows:

For a perfect gas,

$$\frac{\rho_{\rm m}}{\rho_{\rm b}} = \frac{P_{\rm m}}{P_{\rm b}} \frac{T_{\rm b}}{T_{\rm m}} \tag{76}$$

With M_b less than 0.3, equation(77) holds with an error of less than 2%:

$$T_b \cong T_{bo}$$
 (77)

Since the flow in the blowing duct is adiabatic, its stagnation temperature remains constant;

$$T_{bo} = T_{mo}$$
(78)

The last relation required is equation (79):

$$\frac{T_{mo}}{T_{m}} = 1 + \frac{k-1}{2} M_{m}^{2}$$
(79)

When equations (76) through (79) are combined, an equation for calculating $\rho_{\rm m}/\rho_{\rm b}$ is derived:

$$\frac{\rho_{\rm m}}{\rho_{\rm b}} \cong \frac{P_{\rm m}}{P_{\rm b}} \left(1 + \frac{k-1}{2} M_{\rm m}^2 \right) \tag{80}$$

Equations (74), (75), and (80) can be used with small error only if M_b is less than 0.3. If M_b is greater than 0.3, the jet pump-duct matching equation, equation (81), is derived by combining equations (68), (69), and (70):

$$\Delta \mathbf{P}_{t} = \Delta \mathbf{P}_{ext} + \mathbf{K}_{\ell} \mathbf{q}_{m} + \mathbf{P}_{b} \left[\left(1 + \frac{k-1}{2} \mathbf{M}_{b}^{2} \right)^{\frac{k}{k-1}} - 1 \right]$$
(81)

for all values of M_b

where M_{b} is computed by using equation(71) and figure 29.

There are two figures of merit which are of interest in the evaluation of jet pump systems for boundary layer control or thrust augmentation applications. These figures of merit are the entrainment ratio and the thrust augmentation. The equations developed above can be used to determine the entrainment ratio at which a jet pump system will operate. Several additional equations are required in order to calculate the thrust augmentation obtained from a jet pump system.

The thrust augmentation obtained with the complete system is defined in equation (82):

$$\Pi = \text{system thrust augmentation}$$

$$\Pi = \frac{W_m V_b}{W_p V_{p2}}$$
(82)

The thrust augmentation produced by the jet pump alone was designated as τ in section 3.1.4 (equation 22). The computerized jet pump performance analysis of section 3.2 provides as output data values of τ as a function of entrainment ratio. Thus, once the entrainment ratio is known for a jet pump system, the value of τ is known and the system thrust augmentation can be calculated as follows:

$$\tau = \frac{W_{m} V_{m}}{W_{p} V_{p2}}$$

$$\Pi = \tau \frac{V_{b}}{V_{m}}$$
(83)

 \mathbf{so}

The value of V_b/V_m can be related to the blowing duct area ratio by using equation (73):

$$\frac{V_{b}}{V_{m}} = \frac{\rho_{m}}{\rho_{b}} \frac{A_{m}}{A_{b}}$$
(84)

The value of ρ_m / ρ_b is given by equation (80) when $M_b \leq 0.3$. When M_b exceeds 0.3, the value of ρ_m / ρ_b is given by equation (85):

$$\frac{\rho_{\rm m}}{\rho_{\rm b}} = \frac{p_{\rm m}}{p_{\rm b}} - \frac{\left(1 + \frac{k-1}{2} M_{\rm m}^2\right)}{\left(1 + \frac{k-1}{2} M_{\rm b}^2\right)} , \qquad (85)$$

where the blowing duct flow has been assumed to be adiabatic and the value of M_b is determined by using equation(71) and figure 29.

The equations given above can be used to compute the thrust augmentation parameter once the operating point of the jet **p**ump is known. The next section establishes a procedure for determining the operating point.

3.4.4 Calculation Procedure for Determining the Operating Point of a Jet Pump in a Duct System

The operating point of a given jet pump and duct combination can be determined as follows:

Required Initial Data:

Given: Jet pump design and performance characteristics:



$$P_{so}$$

$$P_{po}$$

$$T_{so}$$

$$T_{po}$$

$$A_{th}$$

$$C_{w}$$

$$A_{m}$$

$$k$$

	ſ	ΔP_t	vs	m
jet pump performance	}	^q m	vs	m
computer program	<u>۲</u>	^M m	\mathbf{vs}	m
		^{P}m	vs	m
values characterizing the particular duct system		K _ℓ ΔP _{ex} P _b A _b	ct	

Solution technique if $M_h \le 0.3$:

9

The specified values of K_l , ΔP_{ext} , P_b , and A_b are inserted into equations (80) and (75). The jet pump performance curves are used to find associated values of ΔP_t , q_m , M_m , and P_m which satisfy equation (75). This is a trial-and-error process which is begun by assuming a value for entrainment ratio, m. The corresponding values of M_m and P_m are determined from the jet pump performance curves and entered into equation (80). The resulting value of ρ_m / ρ_b is entered, together with the value of q_m from the jet pump curves, into the right-hand side of equation (75). If the resulting value of ΔP_t does not agree with the curve value, a new value of m is assumed and the process is repeated. The iteration process is simplified by graphical solution techniques which are described in the section entitled "Sample Calculation". This calculation process finds the value of m at which the jet pump system will operate with the selected value of A_b .

Solution technique if $M_h > 0.3$:

The specified values of K_l , ΔP_{ext} , P_b , and A_b are inserted into equations (81) and (71). The jet pump discharge flow rate, W_m , and the stagnation temperature of the discharge flow, T_{mo} , are plotted as functions of entrainment ratio, m. Values of M_b can be determined as a function of m using equation (71) and figure 29. The solution technique is a trial-and-error process which is begun by assuming a value for m. The jet pump performance curves are used to find values of q_m and

 ΔP_t for each value of m assumed. Corresponding values of M_b are determined as above and entered with the q_m values into the right-hand-side of equation (81). If the resulting value of ΔP_t does not agree with the curve value, a new value of m is assumed and the process is repeated. Graphical solution techniques, described in the section below, can reduce the number of iterations required.

Evaluation of thrust augmentation:

The solution techniques described above yield the value of entrainment ratio at which a jet pump will operate in a selected duct system. The performance data provided by the jet pump computer program allows determination of the values of the following jet pump performance parameters at the operating point: τ , P_m, M_m, W_m, and T_{mo}. These values allow calculation of the thrust augmentation, TI, by using the equations presented at the end of section 3.4.3.

Additional Comments:

The values of K_{ℓ} and P_{so} are not constant for all values of entrainment ratio. At high entrainment ratios, the Mach number levels within the ducts may become sufficiently high that the influence of compressibility upon K_{ℓ} must be taken into account. Similarly, the value of P_{so} , which is a non-dimensionalizing parameter in the jet pump performance analysis, varies slightly as shown in equation (86) when the entrainment ratio (and secondary stream flow rate) changes.

$$P_{so} = P_{ao} - K_i q_m$$
 where q_m varies with m (86)

The variations of K_{ℓ} and P_{so} with entrainment ratio are generally second-order in magnitude. These variations can be neglected in preliminary design calculations, then included for final design if K_{ℓ} and K_{i} are known as functions of the Mach number $M_{m} \cong M_{i}$.

3.4.5 Sample Calculation

The use of the procedure described above to determine the operating point of a jet pump-duct system is illustrated by the sample calculation which follows:

The jet pump design data is:

ð

$$\begin{array}{l} \mathbf{P}_{po} &= & 300 \text{ psia} \\ \mathbf{P}_{so} &= & 14.9 \text{ psia} \end{array} \right\} \quad \overline{\mathbf{P}} &= & 20.13 \\ \begin{array}{l} \mathbf{T}_{po} &= & 1200^{\circ} \text{ F} \\ \mathbf{T}_{so} &= & 80^{\circ} \text{ F} \end{array} \right\} \quad \overline{\mathbf{T}} &= & 3.074 \\ \begin{array}{l} \mathbf{A}_{m} &= & 0.08726 \text{ ft}^{2} \\ \mathbf{A}_{th}\mathbf{C}_{w} &= & 0.000165 \text{ ft}^{2} \end{array} \right\} \quad \overline{\mathbf{A}} &= & 545.4 \qquad \text{(Test value for the Case 4 nozzle)} \end{array}$$

The computer solution for the jet pump performance yielded the values given in table 2. The values of ΔP_t , p_m , M_m , and q_m are plotted against entrainment ratio in figure 30.

The duct system design conditions were assumed to be as follows:

$$\Delta P_{ext} = 0 \quad (i.e., P_b = P_{ao})$$
$$K_{\ell} = 0.1$$
$$P_{ao} = 14.9 \text{ psia}$$

The blowing slot discharge Mach number, M_b , was assumed to be less than 0.3. Using these values in equation (75) equation (87) was derived.

$$\Delta \mathbf{P}_{t} = \mathbf{q}_{m} \left[0.1 + \left(\frac{\mathbf{A}_{m}}{\mathbf{A}_{b}}\right)^{2} \frac{\rho_{m}}{\rho_{b}} \right]$$
(87)

The calculations were begun for a blowing duct diffuser area ratio, (A_b/A_m) , equal to 1.0. Three values of entrainment ratio, m = 13, 15, and 17, were selected arbitrarily. The corresponding values of ΔP_t , M_m , q_m , and P_m were read off from figure 30. These values were used to compute ρ_m/ρ_b from

equation (80) and then to compute the right-hand side of equation (87); the right-hand side will be called ΔP_t) trial. The results are given in the table below.

For $A_b = 1.0$:						
m	${\scriptstyle \Delta \ P_t}$ (psf)	M _m	q _m (psf)	P m psia	$\frac{\rho_{\rm m}}{\rho_{\rm b}}$	$ \Delta P_t $ trial (psf)
13	99.6	. 205	65.3	15.15	1.02	74.9
15	98.5	. 235	87	15.00	1.01	96.8
17	97.5	. 270	109	14.80	1.01	121

The ΔP_t trial values can be plotted against entrainment ratio as shown in figure 31. The intersection of the ΔP_t trial curve with the ΔP_t jet pump performance curve represents the solution of equation (87) for the selected value of A_b/A_m . This intersection is the operating point of the jet pump in the specified duct system.

3.4.6 Influence of Blowing Duct Area Ratio and Duct Losses Upon Entrainment

Similar calculations were carried out for values of A_b/A_m equal to 2, 3, and 4. The results are shown in figure 31. The entrainment ratio increases as the blowing duct area ratio is increased; the trend is more clearly shown when the results are replotted as in figure 32. For the particular jet pump and system design conditions assumed for this sample calculation, the maximum entrainment ratio is achieved when the mixing tube is choked, i.e., when $M_{s2} = 1.0$.

The influence of the duct loss coefficient was explored by setting $K_{l} = 0.2$ instead of 0.1 as previously assumed. The duct matching calculations were repeated using equation (75); the results are shown in figure 32. Only a small increase in entrainment ratio can be obtained by increasing the area ratio from 4 to 5. This is a consequence of the fact that, by using a sufficiently large area ratio in the blow-ing duct diffuser, the term ($P_{bo} - P_{b}$) in equation(68) can be reduced to almost zero. In that case, equation (75) takes the following form:

$$\Delta P_{t} \cong \Delta P_{ext} + K_{\ell} q_{m}$$
(88)

for small values of A_m/A_h

 \mathbf{or}

The maximum value of entrainment ratio is the value for which equation (89) holds:

$$K_{\ell} q_{m} = \Delta P_{t} - \Delta P_{ext}$$
(89)

In the present example, with $K_{\ell} = 0.2$ and $\Delta P_{ext} = 0$, the limiting value of entrainment ratio is found by use of figure 30 to be 34. The corresponding mixing tube exit Mach number, M_{m} , is 0.615.

The results show that the maximum entrainment ratio which can be achieved in a duct system driven by a particular constant-area jet pump is set by one of two conditions:

- by choking at the mixing tube outlet or the suction duct inlet (i.e., $M_m = 1.0$ or $M_{s2} = 1.0$) if the duct losses are sufficiently low
- by the duct loss limit which is represented by equation (89) if the value of M_m remains below 1.0.

The form of equation (75) is such that, along curves representing constant values of K_{l} , an increase in blowing duct diffusion always yields an increase in entrainment ratio until the limiting value is reached. In practice, K_{l} is a variable which depends upon the diffuser area ratio. In jet pump systems with low-loss inlets, the value of K_{l} is determined primarily by the blowing duct loss coefficient which increases as the area ratio increases. This trend is shown for conical diffusers in figure 28. Examples of the effect of the variation of K_{l} are shown by the dashed curves in figure 32; these curves represent the loss characteristics of 15° and 20° conical diffusers. The peak entrainment ratio for the 20° diffuser is achieved by using an area ratio of 3; higher area ratios lead to reduced entrainment because of increased losses.

3.4.7 <u>Influence of Blowing Duct Area Ratio and Duct Losses Upon Thrust</u> <u>Augmentation</u>

Figure 33 shows the thrust augmentation parameter, τ , for the jet pump itself. This curve is taken directly from the computer calculations for the jet pump selected in section 3.3.5. In order to determine the values of the system thrust augmentation parameter TT in relation to the blowing duct diffuser area ratio A_b/A_m and the loss coefficient K_{ℓ} , figure 32 was used to determine the entrainment ratio corresponding to selected values of A_b/A_m and K_{ℓ} . Then figure 33 was used to find the associated values of τ . The equations of section 3.4.3 permitted calculation of TT.

The variation of system thrust augmentation with blowing duct diffuser area ratio is shown in figure 34. The curve for $K_{l} = 0$ yields maximum thrust augmentation when the mixing tube is choked, i.e, for $M_{m} = 1.0$. Even with a very low loss in the duct system ($K_{l} = 0.1$), the thrust augmentation reaches a maximum value at a mixing tube Mach number less than 1.0. The curves for $K_{l} = 0.1$ and 0.2 show that the thrust augmentation does not fall off rapidly if the diffuser area ratio is made larger than optimum. This suggests that, when designing a duct system without complete data on duct losses, it is preferable to err on the side of increased diffusion.

The relationship of system thrust augmentation to the entrainment ratio is shown in figure 35. The thrust augmentation peaks on the curve for $K_{l} = 0.1$ and 0.2 and then falls off with increasing entrainment ratio. This is a consequence of the fact that the thrust augmentation is proportional to the product of entrainment ratio and blowing duct exit velocity as follows:

$$\Pi = (m+1) \frac{V_b}{V_{p2}}$$

where $(m + 1) \approx m$ for high entrainment ratio jet pumps and $V_{p2} \approx constant$

so

$$\Pi \approx m V_{b}$$
(90)

In order to achieve entrainment ratios higher than the value at the peak of the TT curve, the diffuser area ratio must be increased. This has the effect of reducing the exit velocity V_b faster than entrainment increases. The net effect is a reduction of the product m V_b and thus TT.

The maximum entrainment ratic attainable is set by the choking limit for the $K_{l} = 0$ and $K_{l} = 0.1$ cases. For the $K_{l} = 0.2$ case, the maximum entrainment value is set by the duct loss limit as represented by equation (84). At this limit, the diffuser area ratio is very large and the duct exit velocity is zero. Consequently, Π must be zero as shown by equation (85). This illustrates the general rule that the thrust augmentation in a jet pump system is always zero at maximum entrainment unless the jet pump mixing tube is choked.

3.4.8 Conclusions

The two previous sections have shown the influence of duct losses and blowing duct diffuser area ratio upon the entrainment ratio and thrust augmentation obtained in a jet pump-duct system. The results shown in figures 32, 34, and 35 are quantitatively valid only for the particular jet pump geometry and operating conditions which were chosen in the section 3.4.5. However, the figures illustrate trends which are qualitatively correct for high entrainment compressible flow jet pumps as a general class.

The results show that the design goals of maximum entrainment and maximum thrust augmentation may require different duct geometries; a system designed for maximum entrainment may have a low value of the thrust augmentation parameter and vice versa. The influence of duct losses is shown to be very strong. Entrainment ratios and thrust augmentation both can be improved significantly by making only minor reductions in the duct loss coefficient. This provides considerable incentive for testing flow models of proposed new duct designs in order to adjust their geometry to achieve minimum losses. Accurate estimates of duct loss coefficients can be obtained from these tests; such estimates are required in order to predict the performance of a new jet pump system and to allow selection of the best diffuser area ratio. The design problem for a jet pump system often takes the following form:

Given: Primary flow pressure, temperature, and flow rate Duct system inlet and discharge pressure levels and inlet pressure

Problem: What is the proper mixing tube area and blowing duct diffuser area ratio to be used to achieve the design goal, e.g., maximum entrainment or thrust augmentation ?

The information provided in plots like figures 32, 34, and 35, together with duct loss estimates, will allow the designer to evaluate the effect of diffuser area ratio upon entrainment ratio and thrust augmentation for a selected mixing tube area. By preparing similar sets of curves for several other values of mixing tube area, the designer can chose the best combination of mixing tube area and diffuser area ratio to meet the design goals. New jet pump performance curves analogous to figures 30 and 32 will be required for each value of mixing tube area to be considered. Data for these performance curves can be obtained by using the computer program described in section 3.2 A series of computer solutions covering a broad range of jet pump geometries and operating conditions is provided in section 3.3 for use in preliminary design calculations.

Section 4

TEST PROGRAM

The test program had two major objectives:

- to provide data for use to evaluate the analytical model
- to determine whether new, reduced-blockage nozzle clusters could be used to improve the performance of the original jet pump

The tests covered a range of operating conditions and nozzle geometries as indicated below:

Primary Flow

pressure range55 psia to 400 psiatemperature range 200° F to 1200° Fnozzle throat area range 1.1×10^{-4} ft² to 6.0×10^{-4} ft²nozzle clusterthree designsnozzle geometryfour designs

Secondary Flow

inlet pressure	laboratory ambient
inlet temperature	laboratory ambient
mixing tube geometry	constant area = $.087 \text{ ft}^2$, two lengths
pressure rise	regulated by discharge throttling device

This section of the report describes the jet pump test arrangement, the test program, and the results which were obtained.

4.1 Test Arrangement

The jet pump test arrangement with its instrumentation is shown schematically in figure 36. The primary flow supply system employed a 2-stage reciprocating compressor capable of supplying 7 lbm/min of air at 400 psia. Electrical heaters were used to achieve temperatures up to 1200° F. The primary flow was delivered to a multiple-nozzle cluster directed along the axis of a constant-area circular mixing tube.

The momentum of the primary flow entrains a secondary air flow from the room into the bellmouth inlet and then into the mixing tube. Here, the two streams mix together and the stagnation pressure of the secondary stream is increased. The flow from the mixing tube passes through a conical diffuser and exhausts to the atmosphere through an adjustable throttling cone.

The individual components of the experimental jet pump are described below:

1. Calibrated bellmouth inlet section.

This component consists of a wooden bellmouth, metal connecting tube, and fiberglass primary flow inlet section. The bellmouth differential pressure was calibrated in terms of flow rate by using an orifice and blower available in the laboratory. The calibrated bellmouth permitted direct measurement of secondary mass flow rate for all jet pump tests.

A ceramic insert was used to protect the fiberglass duct from the hot primary flow pipe and flange.

2. Mixing Tube

The original variable-area mixing tube from the previous investigation (ref. 12) was used for the first tests in order to provide baseline performance data. This mixing tube had a length of 6.87" (figure 37). After the initial tests were completed, the mixing tube was bored out to a constant inner diameter of 4.000". A mixing tube extension of the same diameter was also fabricated. The remainder of the test program was completed using both the original mixing tube length of 6.87" and the extended mixing tube length of 18.87".

3. Conical Diffuser

The initial section of the conical diffuser had a length of 10.98" and an area ratio of 1.79. This diffuser section was previously used during the Wagner BLC system tests. An additional section was added to this diffuser to obtain an overall area ratio of 5.0. The purpose of the exhaust diffuser was to maximize the static pressure recovery so that the highest possible system entrainment ratio could be achieved. Changes in the axial positioning of the throttle cone in the diffuser discharge produced a variable system resistance. The jet pump performance characteristic (pressure rise versus entrainment ratio) was generated by varying the system resistance in this manner.

4. Nozzle Cluster Geometry

The nozzle cluster geometry used in the previous investigation (ref. 12) was believed to cause excessive blockage of the secondary flow at the mixing tube inlet, thus causing reduced performance of the jet pump system. Two "reduced-blockage" nozzle cluster configurations were tested in order to determine whether improved performance could be achieved.

The first reduced blockage nozzle cluster was made by placing 2.0" long nozzle extensions between the original nozzle cluster body and each of the nine nozzles. This change moved the cluster body back away from the mixing tube inlet in order to reduce the velocity level around the cluster by increasing the adjacent flow area. This change is shown on figure 38.

The second reduced blockage nozzle cluster was a completely new design consisting of 7 nozzles at the end of small diameter tubes (1/4" dia.). This low drag cluster is shown on figure 39.

5. Primary Nozzle Geometry

Five sets of primary flow nozzles were used in the test program. The throat size, nozzle type, and design conditions are listed in table 3.

The first four nozzle sets listed in table 3 were used with the original nozzle cluster and also the first reduced-blockage configuration (original cluster with 2" extension tubes). The fifth set, Case LD#2-4, consisted of seven nozzles for the second reduced-blockage cluster. The Case LD#2-4 nozzles were designed for the same operating conditions as the Case 4 nozzles of the original cluster.

The nozzle flow coefficients listed in the table were calculated from test results according to the definition below:

$$C_{w} = \frac{W_{p}}{W_{p}}$$
 ideal

where

- W_p = measured nozzle flow rate at design pressure and temperature
- W = isentropic flow rate through nozzle throat at design pressure and temperature; based upon one-dimensional flow assumption

4.2 Instrumentation and Data Reduction Procedures

4.2.1 Instrumentation

The instrumentation used to determine the performance of the experimental jet pump is shown on figure 40 and described in table 4.

The jet pump inlet bellmouth was calibrated for use as a flowmeter. The calibration was accomplished by connecting the bellmouth and the suction duct to the inlet of a blower. An orifice and a throttling arrangement were included in the blower system. The blower permitted calibration of the bellmouth up to a flow rate of 200 lbm/min. The resulting bellmouth flow equation is given on the following page:

$$W_{\rm s} = 229.5 \sqrt{\rho_{\rm b} \Delta h_{\rm b}} \tag{91}$$

where

 $\Delta h_{\rm h}$ P_b differential pressure, inches of water gage = inlet density, lbm/ft^3

Stagnation pressure traverses were made in the mixing tube and diffuser with a Kiel probe. When the short mixing tube configuration was used, traverses were taken only in the diffuser at the location shown in figure 40. When the mixing tube extension was used, stagnation pressure traverses were taken in the tube 16.4 inches downstream from the primary nozzle exit plane. Additional traverses were made in the diffuser 26.4 inches downstream from the primary nozzle exit plane.

The angular orientation of the traverse locations is shown on figure 40. The same numbering system was used for all traverse locations.

4.2.2 Data Reduction Procedures

=

 $\rho_{\rm h}$

The measured data were used to calculate the following jet pump performance parameters.

$$m = \frac{W_s}{W_p} - jet pump entrainment ratio$$

$$\Delta P_s = \frac{P_m - P_{so}}{P_{so}} - jet pump static pressure parameter$$

$$\Delta P_t^* = \frac{P_mo - P_{so}}{P_{so}} - jet pump stagnation pressure rise parameter$$

$$V vs. \left(\frac{r}{r_o}\right)^2 - velocity profiles$$

$$P vs. distance - jet pump static pressures$$

$$P vs. m and P_{po}$$

The static pressure parameter was calculated using the wall static pressure measured at the discharge of the short mixing tube or the maximum static pressure reached within the extended mixing tube. The secondary flow stagnation pressure at the mixing tube inlet (P_{so}) was determined by subtracting the measured bellmouth inlet and cluster loss (section 4.2.3) from the barometric pressure.

The jet pump stagnation pressure rise parameter, ΔP_t^* , is based upon the secondary flow total pressure P_{so} and upon P_{mo} . Our first tests showed that it was not possible to measure P_{mo} accurately with the "short" mixing tube, i.e., the original 6.87" mixing tube length. Complete mixing was not achieved by the end of this short tube; sharp velocity peaks corresponding to the primary flow jets were observed at the exit section. These peaks were too sharp to be accurately measured by stagnation pressure probes of reasonable size. Therefore, the stagnation probe measurements could not be used to determine P_{mo} with the necessary accuracy.

To obtain an approximate value of P_{mo} for the short mixing tube tests we used the following procedure:

Stagnation pressure probe traverse data was taken at the conical diffuser exit section, P_{do} in figure 40. This data was used to determine an average stagnation pressure (P_{do}) by procedures described below. Then the value of P_{mo} was computed by using equation (92) which accounts for the diffuser stagnation pressure loss.

$$P_{mo} = P_{do} + K \frac{\rho_m V_m^2}{2 g_o}$$
 (92)

The value of K was selected to be 0.05, a value representative of the loss coefficient for a good diffuser.

The procedure used to determine an average stagnation pressure from the stagnation pressure probe readings was the "mass-momentum integral method". The stagnation pressure was measured along a diameter at the diffuser traverse location. A wall static pressure measurement was obtained at that cross-section. Using the ratio of the local stagnation pressure and the wall static pressure at each point along the diameter, the local Mach number was determined. Using a plot of $(1 + kM^2)$ vs. cross-section area (i.e., r^2/r_0^2), the impulse function was determined by graphical integration:

$$\varphi$$
 = impulse function = P $\int (1 + kM^2) dA$ (93)

The following equations were used to determine the "mass momentum" averaged properties of this non-uniform compressible flow:

- calculate
$$\overline{M}$$
 from: $\frac{\overline{M}\sqrt{1+\left(\frac{k-1}{2}\right)} \ \overline{M}^2}{1+k \ \overline{M}^2} = \frac{W_m\sqrt{\frac{R T_{eo}}{kg_o}}}{\varphi}$ (94)

- calculate P (static) from:
$$\overline{P} = \frac{\varphi}{(1 + k\overline{M}^2) A}$$
 (95)

- calculate P_0 (stagnation) $\overline{P}_0 = \overline{P} \left[1 + \left(\frac{k-1}{2}\right) \overline{M}^2 \right]^{\frac{K}{K-1}}$ (96) from:

These "mass-momentum average" values of pressure satisfy the measured mass flow rate and integrated momentum of the flow. They correspond to values that would be obtained if the actual nonuniform flow was mixed to a uniform flow in a frictionless, constant area duct.

To calculate P_{mo} for the extended mixing tube, the "mass momentum integral method" was applied to the traverse data taken in the mixing tube 16.4" downstream from the primary nozzle exit plane. No correction for diffuser losses was needed in this case.

The velocity profiles were calculated from the local values of Mach number and the measured jet pump exhaust temperature (T_{eo}) .

4.2.3 Suction Duct and Nozzle Cluster Losses

1

In order to calculate the jet pump pressure rise parameters ΔP_s and ΔP_t^* , the mixing tube inlet stagnation pressure P_{so} is required. The value of P_{so} was determined by subtracting the appropriate suction duct and nozzle cluster

stagnation pressure losses from barometric pressure (the suction duct inlet stagnation pressure). The loss data used for the determination of P_{so} is given below.

Bellmouth Inlet

The stagnation pressure losses in the bellmouth, suction duct, and nozzle cluster configuration used in the test rig (figure 36) were measured for the following four configurations:

- 1. Original nozzle cluster
- 2. Original nozzle cluster with 2" nozzle extensions (figure 38)
- 3. "Low-drag" nozzle cluster (figure 39)
- 4. No nozzle cluster

The loss measurements were made by connecting the jet pump including the initial section of the conical diffuser (area ratio 1.79) to the suction line of a blower and orifice installation. Air was drawn through the bellmouth and jet pump system by the blower. The stagnation pressure at the end of the constant area mixing tube was calculated using the measured wall static pressure and the Mach number (computed on a one-dimensional basis) at the same measuring station. The stagnation pressure loss was set equal to the difference between this value and the atmospheric pressure. The loss values for the four nozzle configurations are presented on figure 41. These loss values include the suction duct loss, the nozzle cluster loss, and the short mixing tube wall friction loss ($L/d_m = 1.35$).

The results show that the losses are identical for the two reduced blockage clusters #2 and 3. A comparison of the losses for the low blockage clusters to the losses for the original cluster and the minimum possible losses (the "no cluster" curve) shows that the reduced blockage clusters cut the cluster losses by about 42%.

These results were used for two purposes:

1. to guide the selection of the "optimum" nozzle cluster

2. to permit calculation of P_{so} at the jet pump inlet by subtracting the suction tube and nozzle cluster losses from the atmospheric pressure. The P_{so} value was used to calculate the static pressure parameter ΔP_s^* and the stagnation pressure rise parameter ΔP_t^* .

BLC Suction Duct

The data obtained during tests of the Wagner BLC system under the previous contract NAS 2-2518 indicated that the aerodynamic drag of the present nozzle cluster was much higher than had been expected. This drag was considered to be one of the principal reasons for the difference between the predicted and actual performance of the jet pump BLC system. To check this point, the NAS 2-2518 suction duct (figure 42), the short mixing tube, and the initial section of the conical diffuser were connected to the inlet of the blower and orifice system. The original nozzle cluster geometry was installed in the suction duct. The tests results are shown on figure 41. The BLC suction duct produced higher losses than the bellmouth configurations.

Still using the NAS 2-2518 suction duct, the nozzle cluster was dismantled in 3 steps with a loss test made between each step. The nine nozzles only were removed first but no decrease in loss was measured. Next, the cluster body was removed but the elbow was left in place. The measured losses dropped by 13% to the "no cluster" curve shown on figure 41. This amounts to a 5 psf loss reduction for a jet pump entrainment ratio of 20 ($W_s = 128 \text{ lb/min}$). When the elbow was removed, no further reduction of losses was observed. Therefore, the cluster body with its extensions produced the nozzle cluster losses.

This series of loss tests shows that the nozzle cluster losses are much smaller than had been deduced from the experimental data obtained on the jet pump BLC system. The NAS 2-2518 final report showed suction duct losses to be three times the values shown on figure 41. At that time, a large percentage of these losses were attributed to the nozzle cluster. The results in figure 41 show that this conclusion was incorrect for two reasons:

- 1. The calculation of suction duct stagnation pressure loss from the BLC system data is sensitive to the choice of the static pressure value at the mixing tube inlet to be used for the calculation. The static pressure data recorded by the first 2 or 3 pressure taps in the first half-inch of the mixing tube always showed a sharp spike of low static pressure at the mixing tube inlet. The static pressure variation becomes more gradual from about the fourth tap onward in the mixing tube. Testing of the mixing tube with the BLC suction duct during the series of loss tests of figure 41 showed this same variation of static pressure even without the nozzle cluster in place. The spike of static pressure must be caused by the local curvature of the streamlines as the flow reaches the mixing tube throat. If suction duct losses are calculated from the third or fourth static pressure tap instead of the first tap measurement, the calculated BLC suction duct losses would compare favorably to the measured losses shown on figure 41.
- 2. The loss tests on the BLC suction duct showed that the duct itself causes most of the suction duct losses. The nozzle cluster accounts for about 13% of the suction duct loss which may amount to 4% to 8% of the total system resistance.

4.2.4 Pressure Loss Due to Wall Friction in the Constant-Area Mixing Tube

The mixing tube wall friction loss levels must be taken into account when comparing the experimental pressure parameters to the analytical predictions which do not include these losses. Therefore, the pressure loss in the extended mixing tube was measured during the nozzle cluster loss tests which were described in section 4.2.3. The wall static pressure was measured at two points 10.88 inches apart in the constant 4" diameter mixing tube while air was being drawn through the tube by the blower. The total pressure change was calculated using the local Mach numbers (determined on a one-dimensional flow basis) at the two cross sections. The static pressure change and the stagnation pressure loss as a function of flow rate are given in figure 43. The static pressure change is larger than the stagnation pressure loss because of compressibility effects. The measured loss levels agree well with predictions based upon pipe friction factors for fully-developed turbulent flow.

When the jet pump is operating, the mixing tube velocity profiles differ from the profiles for fully-developed turbulent flow in pipes because of the primary flow - secondary flow interactions. Therefore, wall friction losses in a jet pump mixing tube can be expected to differ somewhat from the losses predicted for fullydeveloped turbulent flow. No data correlations or analytical procedures are available to allow accurate prediction of wall friction losses in mixing tubes. Thus, pipe friction factor correlations or equivalent test results as in figure 43 must be used as a first approximation in order to estimate mixing tube wall friction losses so that the analytical predictions of jet pump performance can be compared to the experimental results.

4.3 Tabulation of Test Conditions

The jet pump testing was carried out in three series of tests. The configurations used and the test objectives are described below.

- Series 1: The jet pump included the calibrated bellmouth inlet, the original short NAS 2-2518 mixing tube with varying area, and the original nozzle cluster. The purpose of this test series was to determine the performance of the jet pump as used in the NAS 2-2518 BLC system tests.
- Series 2: The jet pump included the calibrated bellmouth inlet, the short constant-area mixing tube, and three nozzle clusters; the original NAS 2-2518 cluster, the same cluster with 2" nozzle extensions and moved back from the mixing tube throat (i.e., low-drag cluster #1), and the low-drag nozzle cluster #2 as in figure 39. The objectives of this test series were to determine the effect of nozzle cluster design upon jet pump performance and to determine which cluster would be best for subsequent testing.

Series 3: The jet pump configurations tested included the calibrated bellmouth inlet, both the short and the extended constant-area mixing tubes, and the optimum nozzle cluster selected from the results of test series #2 (the LD#1 cluster). The purpose of this test series was to provide experimental verification of the jet pump performance analysis over a broad range of operating contions. Each of the four nozzle sets described in table 3 were used; they were operated not only at their design points but also at off-design pressures and temperatures. É

A description of the individual tests, operating conditions and configurations is given in table 5.

4.3.1 Presentation of Data

Tabulated data and graphical results are presented for each of the run numbers listed in table 5. An index to the tables and figures is given in table 6. The test data is presented in tables 8 to 19 and figures 44 to 84.

4.4 Discussion of Test Results

4.4.1 Comparison of Constant-Area and NAS 2-2518 Varying-Area Mixing Tubes

The original NAS 2-2518 short mixing tube was designed to have a 5.5% contraction in area from inlet to outlet. The tube was intended to have constant-pressure mixing at its design point. Since the jet pump has not been able to produce a flow rate as high as the mixing tube design flow rate, the constant static pressure condition was never achieved. However, a nearly-constant static pressure distribution has been approached near the discharge end of the mixing tube at the higher flow rates obtained during the test program.

The short constant-area mixing tube was made by boring out the NAS 2-2518 mixing tube to a constant internal diameter of 4.00".

Entrainment Ratio Results

Tables 8 and 9 present the primary and secondary flow rates and the entrainment ratio for each test point recorded for the two short mixing tube configurations. For each primary stagnation pressure, the maximum entrainment ratio represents the test where the throttle cone was moved out of the exhaust diffuser as far as it would go. The two test configurations were then completely identical except for the mixing tube. The data shows that use of the NAS 2-2518 mixing tube resulted in maximum entrainment ratios about 6% to 8% higher than the constant-area mixing tube configuration. A basic difference was observed between the measured velocity profiles at the diffuser discharge (see below).

Jet Pump Stagnation and Static Pressure Parameter Results

The experimental results for both mixing tube configurations are compared to the analytical predictions in figure 48. Both the stagnation and static pressure parameters are plotted for four primary pressures.

The experimental stagnation pressure rise values calculated from a single traverse were found to be 13% to 32% below the analytical predictions. The one experimental value calculated from traverses in two perpendicular directions was 8% below the analytical prediction. The differences between the experimental and analytical values are caused by a combination of effects listed below.

- The experimental value is very sensitive to the accuracy and thoroughness of the traversing of the diffuser discharge to determine P_{do}.
- The estimated diffuser losses between the mixing duct exit and the traverse station are added to P_{do} to determine P_{mo}. Since the diffuser inlet flow is highly distorted (the mixing tube is too short and primary nozzle jets persist into the diffuser inlet) the estimated losses may be too low.
- The suction duct and nozzle cluster losses must be subtracted from the barometric pressure in order to determine P_{so} . The

loss characteristics used were obtained by drawing air past the nozzle cluster using a blower at the discharge of the duct system (section 4.2.3). The flow around the cluster may be different when the primary nozzles are in operation. Higher inlet losses may be the result.

• Wall friction losses in the mixing tube (not taken into account in the analytical predictions) reduce the jet pump stagnation pressure rise (section 4.2.4).

The experimental static pressure parameter curves are similar in slope to the curves for the analytical predictions. The experimental curves are shifted to higher negative values of the static pressure parameter. This means that the measured static pressure values are too low just as the integrated stagnation pressure values were too low. There are two reasons why this occurred.

- The data shows that the static pressure is still increasing at the end of the constant-area mixing tube. This means that considerable mixing is still taking place. A longer constant-area mixing tube improves the agreement of the experimental and analytical static pressure parameters as shown in section 4.4.3.
- The suction pipe and nozzle cluster losses, which are used to determine P_{so}, may not be accurately represented by our measured loss characteristics. This problem was discussed above for the stagnation pressure loss parameter.

To improve the agreement between the experimental and analytical results, the length of the mixing tube was increased using the mixing tube extension piece. The stagnation pressure traverses were then repeated at the end of the extension tube.

Velocity Profiles at the Diffuser Discharge

Velocity profiles were calculated from the stagnation pressure traverse data used to calculate the jet pump discharge stagnation pressure. The velocity profiles measured for the two mixing tubes are presented in figures 44, 45, and 49. Figure 44 shows velocity profiles taken at four throttle settings at a primary flow pressure of 350 psia using the NAS 2-2518 mixing tube. Three observations can be made from this figure.

- The profiles are generally not symmetrical,
- a sharp dip in velocity occurs at the center of the flow for the higher entrainment ratios,
- the velocity dip becomes less pronounced as entrainment ratio decreases.

Figure 45 shows two velocity profiles for the NAS 2-2518 mixing tube at reduced pressures and one velocity profile measured with an unheated primary flow. In all profiles, a sharp dip in velocity occurs in the center of the tube. The unheated primary flow velocity profile is nearly symmetrical. The asymmetry of all of the heated primary flow velocity profiles is probably caused by a radial shift of the nozzle cluster relative to the mixing tube due to thermal expansion of the elbow which feeds hot air to the cluster. This shift has been measured to be about 1/16 inch. This situation was corrected in some of the subsequent tests by setting the nozzle cluster off-center at room temperature to compensate for thermal expansion at operating temperature.

Figure 49 presents velocity profiles measured for the short constant-area mixing tube configuration. The velocity profiles are not symmetrical for this mixing tube either. For run 24, two perpendicular velocity traverses were obtained. Both traverses were asymmetrical and both were shifted in the direction which would be expected if caused by elbow thermal expansion (the location of the traverse planes relative to the elbow is shown in figure 40). The dip in velocity at the center of the tube was much smaller for the constant area mixing tube configuration than for the NAS 2-2518 mixing tube. Further discussion of the dip in the velocity profile is presented in section 4.4.4. The presence of the exhaust cone was **sh**own to have no effect on the velocity profile and a wedge probe traverse revealed that a small amount of swirl was present in the flow.

Mixing Tube Static Pressure Distribution

Static pressure data is presented in figures 46, 47, and 50. Figure 47 compares the variation of static pressure along the mixing tube and diffuser for three configurations:

- 1. NAS 2-2518 mixing tube in the BLC duct system.
- 2. NAS 2-2518 mixing tube in the Dynatech Test Rig.
- 3. Constant area mixing tube in the Dynatech Test Rig.

The test conditions for each of the three configurations are listed below. Slight differences in primary temperature and entrainment ratio existed for the three conditions.

Configuration	P po psia	T po °F	W _p lbm/min	W _s lbm/min	W _m lbm/min	m
1	350	1200°	6.4	118.5	124.9	18.5
2	350	1085°	6.70	118.20	124.9	17.65
3	350	1130°	6.65	118.25	124.9	17.8

The static pressures for configuration 2 are about 2" of water larger than for configuration 1. This shift in the static pressure level is a result of reduced losses in the suction duct and bellmouth as compared to the BLC suction duct. The 2" of water shift is equivalent to a 10.5 psf decrease in suction duct losses. Comparison of this number to figure 41 shows that 10.5 psf is about half of the difference between the measured losses for the two inlet geometries.

The effect of changing from the contracting NAS 2-2518 mixing tube to a constant area tube is shown by configurations 2 and 3 in figure 47. The static pressure rises more rapidly in the constant area tube because of the lower velocity levels However, the static pressure recovery obtained in the diffuser is substantially larger

for the NAS 2-2518 mixing tube case. This effect may be the result of lower mixing tube stagnation pressure rise or reduced diffuser effectiveness for the constant-area mixing tube case.

Figures 46 and 50 show the variation of static pressure at two locations in the mixing tube for configurations 2 and 3 as a function of the primary pressure and the entrainment ratio. The increase in static pressure for the constant area tube above that of the contracting tube was found to exist for all primary pressure levels and entrainment ratios tested.

4.4.2 Reduced Blockage Nozzle Clusters

Two reduced drag nozzle cluster configurations were tested to determine the influence of the cluster drag upon jet pump performance and to allow selection of the best cluster for further testing. Section 4.1 presents the dimensions of each of the cluster configurations. Table 5 lists the test conditions and table 6 provides an index to the results obtained.

The jet pump performance with the three nozzle cluster configurations (original, original cluster with 2-inch nozzle extension, and the low-drag cluster #2) were compared in the following ways:

- 1. System performance at minimum throttling (wide open throttle cone)
- 2. Static pressure parameter
- 3. Static pressure variation along the mixing tube at a selected total flow rate
- 4. Velocity profiles
- 5. Cluster pressure loss characteristics

Entrainment Ratio Results

The system performance at minimum throttling is presented in figure 51 where secondary flow rate is plotted as a function of primary flow rate. Except for the loss characteristics of the nozzle clusters themselves, the system loss characteristics are identical for all three nozzle configurations when the cone is in the wide open position. Both of the modified nozzle clusters show equal improvement with respect to the performance of the original cluster. Both have an increased secondary flow rate for the same primary flow rate.

The LD #2-4 cluster nozzles were designed for a slightly lower primary flow rate at a given pressure and temperature than the Case 4 Nozzles. This difference in design accounts for the shift in data points along the curve for the two reduced drag clusters.

Jet Pump Static Pressure Parameter Results

The analytical and experimental static pressure parameters are compared for the three nozzle clusters in the following figures:

> Original Nozzle Cluster: figure 48 LD#1 - Low-Drag Nozzle Cluster: figure 52 LD#2 - Low-Drag Nozzle Cluster: figure 56

The best agreement between analytical and experimental results was obtained for the LD#1 cluster (figure 52). The higher nozzle cluster losses in the original cluster (figure 48) and the less-complete mixing obtained with the 7-nozzle arrangement of the LD#2 cluster (figure 56) are the causes of the poorer agreement between analysis and test results for these clusters.

Velocity Profiles at the Diffuser Discharge

The velocity profiles for the original cluster without and with nozzle extensions are presented in figures 49 and 53. The profiles had two velocity peaks with a dip in the center amounting to 30 to 60 fps. The velocity profiles for the LD#2cluster are presented in figure 57. Three nearly-equal velocity peaks were obtained with the LD #2 cluster when traversing along a line passing through three nozzles, and one velocity peak was obtained when traversing along a line passing between nozzles. The difference between the profiles for the original cluster and for the LD #2 cluster appears to be related to the number of nozzles used. The LD #2 nozzle cluster has seven equally-spaced nozzles, each having an equal share of the mixing tube flow cross section to energize. The original cluster has two additional nozzles around the outside, leaving the center nozzle with a larger percentage of the mixing tube flow area to energize. The velocity dip probably was not caused by nozzle cluster losses (i.e., a wake effect) because the LD#1 and LD#2 clusters apparently had similar loss characteristics (figure 51).

Several tests were completed with and without the throttle cone in place. No change in the velocity profile was detected. Therefore, the presence of the throttle cone at the diffuser exit does not seem to be the cause of the dip at the center of the velocity profile when the original nozzle cluster was used. No satisfactory explanation for the dip was developed during this program.

Mixing Tube Static Pressure Distribution

The static pressure variation along the mixing tube is shown for all three clusters in figure 55. The static pressure levels are dictated primarily by the total flow rate, but they are also slightly affected by changes in the mixing process which accompany modifications of the nozzle cluster geometry. The original nozzle cluster produced similar pressure distributions with and without the nozzle extensions. The pressure distribution for the LD #2 cluster shows higher static pressures near the mixing tube inlet because of reduced blockage, and lower static pressures at the mixing tube exit because of less complete mixing (the result of the reduction in number of nozzles from 9 to 7).

Cluster Drag

The cluster loss tests were discussed in section 4.2.3. Both of the lowdrag clusters showed equal reduction in loss characteristics relative to the original cluster (figure 41).

Selection of the Optimum Nozzle Cluster

The measured nozzle cluster losses (figure 41) and the jet pump performance characteristics with minimum throttling (figure 51) show that the two lowdrag nozzle clusters yield similar results. The comparison of experimental and analytical static pressure parameters and the comparison of static pressure distributions along the mixing tube for both low-drag clusters show that the original cluster with 2 inch nozzle extensions (LD #1) is slightly preferable to the LD #2 cluster.

The original cluster with 2 inch nozzle extensions (LD # 1) was selected as the best cluster for the remainder of the test program for the following reasons:

- 1. The nozzles for the remainder of the test program were already available for this cluster.
- 2. This cluster gives better agreement between experimental and analytical static pressure values because of the more complete mixing upstream of the measuring station.
- 3. The LD #2 cluster offers no advantages in comparison to the original cluster with extensions; the measured losses are equal for both clusters and the flow rate curves at the wide-open throttle position are the same.

4.4.3 Comparison of Short and Extended Mixing Tubes

Performance data was obtained for both the short and the extended constant-area mixing tubes using each of the four nozzle sets with the LD#1 nozzle cluster. The test results are indexed in table 6 .

The extended mixing tube was found to change the jet pump performance in the following ways:

1. The static and stagnation pressures both reached a maximum value within the extended mixing tube.

This indicates that complete mixing was achieved. The extended mixing tube was superior to the short mixing tube, particularly with respect to static pressure recovery at low entrainment ratios.

- 2. The wall friction losses in the extended mixing tube were significant in comparison to the stagnation pressure rise developed by the jet pump.
- 3. The velocity profile at the diffuser inlet was improved by extending the mixing tube.

Identical duct loss characteristics existed for the maximum-entrainment ratio runs for each test number. The throttle cone was withdrawn to a fixed location for these runs. Thus, the performance characteristics of the jet pump as influenced by nozzle design, mixing tube length, and primary flow pressure and temperature can be determined by comparing these maximum-entrainment runs.

Entrainment Ratio Results

A Comparison of the maximum entrainment ratio achieved with the long and short mixing tubes operated with the same nozzles and the same primary flow conditions shows that the short mixing tube configuration achieves a slightly larger entrainment ratio in all cases. The mixing tube extension section allows more complete mixing and improves the diffuser inlet velocity profile. However, the extended tube introduces an additional frictional loss which becomes significant at large flow rates (figure 43). The reduction in entrainment for the extended mixing tube shows that the extra friction losses in the longer tube cancel the effects of improved mixing. The use of a mixing tube length longer than the short tube and shorter than the extended tube would probably lead to a higher entrainment ratio than was developed by either of the tested lengths.

Jet Pump Static Pressure Parameter Results

Table 6 lists the figure numbers which show the static pressure parameters for the eight nozzle and duct configurations tested. The experimental values shown are based upon measured static pressures (P_m or P_{max}) and upon secondary flow stagnation pressures (P_{so}) calculated from the measured inlet losses, cluster losses, and mixing tube losses (figures 41 and 43). The analytical values shown in the figures were calculated by the computer using the ideal jet pump analytical model which neglected inlet, cluster, and mixing tube losses.

The results show that the extended mixing tube test data corresponds more closely to the analtycal predictions. The biggest difference between the short and long mixing tubes occurs at low entrainment ratios because the extended mixing tube significantly increases the static and stagnation pressure recovery at low flow rates. At high entrainment ratios, no improvement is produced by the extended mixing tube because the increased frictional losses cancel the potential gains from more thorough mixing.

Inclusion of the inlet and mixing tube losses in the analytical model would give lower (more negative) values of the static pressure parameter which would agree more closely with the experimental values. The biggest changes would occur at high entrainment ratios where the curves are presently furthest apart.

Jet Pump Stagnation Pressure Parameter Results

Table 7 lists all of the experimental stagnation pressure rise parameters which were calculated from the traverse data taken. Traverses were made either at the mixing tube discharge or in the conical diffuser at a station where the area is 1.61 x the inlet area. The stagnation pressure at the traverse station was calculated by the mass-momentum method presented in section 4.2.2. Each result represents the integration of one or two traverses as indicated in the table. The value of P_{mo} was determined from the diffuser exit traverse values (P_{do}) by correcting for the diffuser losses using equation (92) of section 4.2.2.
Figure 59 compares the analytical predictions and experimental values of the stagnation pressure rise parameter for the extended mixing tube tested with the Case 4 nozzles at 300 psia and 1150° F primary flow conditions. The experimental values were obtained as follows:

- The stagnation pressure value P_{mo} was calculated using the diffuser exit stagnation pressure traverse data and correcting for diffuser losses by using equatior (92) of section 4.2.2.
- The stagnation pressure P_{mo} was calculated by using stagnation pressure traverse data obtained at the mixing tube exit.
- An additional correction was made to the O i data to account for mixing tube wall friction effects. The wall friction pressure losses are taken from figure 43 to make this correction.

The correction of the stagnation pressure rise parameter to account for duct losses places three of the four experimental points within 6 psf of the analytical value. The differences which remain may be due to the factors discussed in section 4.4.1 where similar results were presented for the short mixing tube test.

Additional comparisons of the analytical and experimental stagnation pressure values are made in section 5.1 of this report.

Velocity Profiles at the Diffuser Discharge

Table 6 lists the figure numbers showing the velocity profiles obtained with both the short and long mixing tubes. The effect of extending the mixing tube can be seen clearly on figure 73 where results for both mixing tubes are plotted for the Case 2 nozzles. The centerline depression at the diffuser discharge is much smaller when the extended mixing tube is used since the flow has a longer flow path for mixing. The tests with the Case 3 and Case 4 nozzles gave the same result. The longer mixing tube improves the symmetry of the flow entering the diffuser and reduces the likelihood of flow separation in the diffuser.

For the Case 3 nozzles with the extended mixing tube, two traverses each were taken at the mixing tube exit and at the diffuser exit. One traverse at each location passed along a diameter through the wake of three nozzles. The second traverse in each location passed through the wake of only the centerline nozzle (figure 79). The four traverses show the flow to be reasonably symmetrical. A reduced velocity exists along the centerline while two peaks of velocity appear on either side. The velocity profiles in the two planes at one station were nearly identical. Further discussion of the velocity profiles is included in section 4.4.4.

Mixing Tube Static Pressure Distribution

Table 6 lists the figures showing the static pressure variation along the mixing tube. These variations are plotted only for the extended mixing tube tests. The static pressure at each location in the duct is a result of the interaction of the following factors:

- 1. The local Mach number of the flow,
- 2. The stagnation pressure rise achieved by mixing of the two streams,
- 3. The frictional losses on the walls.

The data in figures 71, 75, and 81 show that the static pressure reaches a maximum at the middle or near the end of the mixing tube extension. In the duct upstream of the location of maximum static pressure, the increase in stagnation pressure due to mixing is larger than the reduction in stagnation pressure due to wall friction. Beyond the maximum point, the added wall friction loss becomes dominant. The location of the maximum static pressure point is closer to the mixing tube inlet for the high flow rate test points because of the increased magnitude of frictional pressure loss which accompanies the increased velocity levels. These results show that an optimum mixing tube length exists which will produce the maximum static pressure. This optimum length is a function of flow rate.

4.4.4 Velocity Profile Investigations

As mentioned in other portions of section 4.4, the velocity profiles measured at the mixing tube exit and the discharge of the initial diffuser section have shown various degrees of distortion. This distortion is due to the following causes:

- 1. Angular and/or radial misalignment of the nozzle cluster and mixing tube centerlines.
- 2. Non-uniform pumping.

The alignment of the nozzle cluster elbow with the mixing tube centerline was accomplished with a special centering plug which was inserted snugly into both the elbow and the mixing tube. The original centering plug aligned the elbow and tube concentrically with the elbow at room temperature. Traverses taken with low primary air temperatures (from ambient to 200° F) gave velocity profiles which were quite symmetrical about the centerline, thus indicating good alignment. However, when the primary flow was heated to 1150° F, the elbow flange-to-centerline dimension increased by about $1/16^{"}$ due to thermal expansion. The high velocity region in the velocity profile shifted noticeably in the same direction (an example is given by figure 49).

To compensate for the thermal expansion, an offset plug was made to position the two centerlines 1/16 inch apart when the elbow was at room temperature. The offset plug was used to position the cluster for all of the high temperature tests performed subsequent to test No. 16. Even with the offset plug, some of the short constant area mixing tube tests showed distorted velocity profiles. indicating that the mixing process in the short tube is highly sensitive to slight misalignments. All of the velocity profiles for the extended mixing tube tests were reasonably symmetrical showing that the mixing process in the longer tube is relatively insensitive to misalignment. Most of the velocity profiles exhibited a slight depression of the centerline velocity below the velocity of the surrounding flow. This depression was less pronounced when the extended mixing tube was used (an example is given by figure 73). To determine the reason for the slight depression of velocity along the centerline, the following series of tests were run:

- 1. Velocity profiles were obtained with the Kiel probe with and without the throttle cone in place. The same primary flow conditions were maintained for both tests.
- 2. A wedge probe was used to measure local static pressure and flow direction for comparison with a Kiel probe measurement.
- 3. A nozzle arrangement using 8 Case 2 nozzles around the circumference of the cluster and 1 Case 4 nozzle located in the center was tested to explore the effect of increasing the momentum in the center of the mixing tube.

The first test described above showed conclusively that the throttle cone does not influence the flow at the traverse station. There was no detectable difference in the traverse results with and without the cone.

The traverse results for tests 2 and 3 above are shown on figure 85. The wedge probe results, like the Kiel probe results, show the depression in velocity along the centerline. Associated with this depression in velocity was a reduction in static pressure of about 1 inch of water and a departure of the velocity from the axial direction by roughly $\pm 2^{\circ}$ all along the diameter. These results indicate that a slight swirl exists in the flow.

The third test above employed a center nozzle with a throat area about 50% larger than that of the surrounding eight nozzles. The velocity profile still shows a centerline depression. Thus, the depression cannot be accounted for as only a primary flow momentum deficiency in the center of the mixing tube.

The small amount of swirl that appears to be present in the flow is not likely to have a significant effect on the test results. This small amount of swirl could be caused by one or more of the primary nozzles being bent at a small angle to the axis, by the presence of the primary nozzle elbow, or by wakes shed off of objects in the laboratory outside the jet pump.

No satisfactory explanation for the centerline depression in the velocity profile remained after these tests were completed. The cause of the depression is unknown.

Section 5

6a

COMPARISON OF ANALYTICAL AND EXPERIMENTAL RESULTS

5.1 Jet Pump Stagnation Pressure Rise

The experimental measurements of the jet pump stagnation pressure rise parameter, ΔP_t^* , generally fell below the analytical predictions. Examples are given in section 4.4.1 (figure 48) and section 4.4.3 (figure 59). Additional values of ΔP_t^* were measured in other tests and the results are given in table 7.

The analytical values of ΔP_t^* were predicted by neglecting wall friction losses in the mixing tube, by assuming complete mixing, and by neglecting suction duct losses and conical diffuser losses. Thus, the analytical values represent "ideal" jet pump performance. The measured performance fell below the "ideal" values for the following reasons:

- the experimental value is very sensitive to traversing thoroughness and accuracy
- mixing tube wall friction losses were not included in the analysis; these losses are not negligible
- in many tests, the stagnation pressure traverses were made in the conical diffuser. The test results were corrected to account for diffuser losses between the measuring section and the mixing tube exit. These corrections may be inaccurate.
- to determine the measured value of $\Delta P_t = P_{mo} P_{so}$, the value of P_{so} had to be estimated by subtracting suction duct and nozzle cluster losses from the atmospheric pressure. These losses cannot be measured under the conditions which exist during jet pump operation so the corrections may be inaccurate.

• the adjustment of the supersonic primary flow to match the secondary flow static pressure was assumed to be isentropic in the analysis. When the primary flow at the nozzle exit is appreciably over- or under-expanded, the flow is non-isentropic and stagnation pressure losses occur in the primary flow adjustment process.

The effects of mixing tube wall friction can be estimated by using the loss measurements of figure 43. The analytical values of ΔP_t^* were reduced by the appropriate loss taken from figure 43 to derive the column in table 7 entitled "Corrected for Duct Friction". The corrected analytical values were used to prepare the right-hand column which shows the percentage error between the corrected analytical value and the test measurement of ΔP_t^* , i.e.:



The table shows that the difference between the corrected analytical value and the test value of ΔP_t^* is 10% or less when traverses in two directions were made during the test. The difference can be 20% or more if only one traverse was made, particularly when the short mixing tube was used or distorted velocity profiles were observed.

The remaining differences between the corrected analytical value and the test value are due to incomplete traversing and possible inaccuracies in the conical diffuser and suction duct loss corrections. Furthermore, the pressure loss measurements shown in figure 43 were made by drawing air through the mixing tube with a blower. The wall friction losses may be different when the jet pump is operating because the mixing action changes the velocity profiles considerably. Thus, even the mixing tube wall friction corrections may not be exact.

5.2 Jet Pump Static Pressure Rise

The experimental measurements of the jet pump static pressure rise parameter, ΔP_s^* , fell consistently below the analytical predictions. The measured static pressure values at the mixing tube exit were lower than predicted. The reasons for this discrepancy were as follows:

- mixing tube wall friction losses were not included in the analysis; these losses are not negligible
- in the case of the short mixing tube, the tests showed that mixing was not completed within the tube. Thus, the static pressure_did not reach its mixed-out value, the value which the computer program seeks to predict.
- the distorted velocity profiles existing in the actual mixing tube lead to lower static pressures than the uniform velocity profiles assumed in the computer analysis.
- to determine the measured value of $\Delta P_s = P_m P_{so}$, the value of P_{so} was estimated by subtracting suction duct and nozzle cluster losses from the atmospheric pressure. These loss corrections may not be exact.
- the primary flow is not isentropic in the accommodation region when the primary flow is over- or under-expanded at the primary nozzle exit. Isentropic accommodation is assumed in the analysis.

The differences between the experimental measurements and the analytical predictions are most pronounced at low entrainment ratios when poor mixing occurred, when the short mixing tube was used, and particularly with the 7-nozzle LD#2 nozzle cluster. The extended mixing tube improved the recovery of static and stagnation pressure at low flow rates. At high flow rates (high entrainment ratios), the increased frictional losses in the extended mixing tube cancel the potential gains from more thorough mixing. The computer program was modified as described in sections 3.1.5 and appendix B.1 to allow inclusion of mixing tube wall friction and suction duct losses in the analytical performance predictions. Preliminary values of $K_{\rm MT}$ and $K_{\rm sd}$ can be obtained by loss tests as infigures 41 and 43, or by estimates using duct loss correlations available in the literature. More accurate values of $K_{\rm MT}$ and $K_{\rm sd}$ must be obtained empirically from jet pump testing.

An example of the empirical approach is shown in figure 86. Test values of ΔP_s^* are replotted for the 260 psia operating condition from figure 72. A number of alternative values of K_{MT} were assumed and used as input data for the computer performance analysis. The value of K_{sd} was assumed to be 0.006 for all calculations. The results show that selection of $K_{MT} = 0.055$ makes the analytical prediction agree closely with the test results. The loss coefficient K_{MT} as used here includes not only the effect of mixing tube wall friction, but also the effects of the other sources of discrepancies mentioned above. The mixing tube loss coefficient measured by drawing air through the mixing tube with a blower (figure 43) was $K_{MT} = 0.053$.

This good agreement between the value of $K_{\rm MT}$ measured in the blower test and the value of $K_{\rm MT}$ deduced from the computer calculations suggests that the blower test method may afford a simple and accurate way to determine $K_{\rm MT}$ for a new jet pump design. However, the mixing action which occurs when the jet pump is operating may cause the mixing tube wall friction characteristics to vary when the primary flow pressure and temperature are changed. To determine whether such variations of $K_{\rm MT}$ are significant, more of the test results of section 4 could be analyzed by the computer to determine the appropriate values of $K_{\rm MT}$. These "jet-pump-derived" values of $K_{\rm MT}$ could be compared to $K_{\rm MT}$ as measured by the blower test method to determine whether serious discrepancies can occur. This report includes sufficient data to make such comparisons.

ł

Section 6

CONCLUSIONS

The conclusions which may be reached as a result of this investigation are listed below:

1. Validity of the Analysis:

The analytical model developed in section 3.1 for high-entrainment compressible-flow jet pumps with constant area mixing tubes is based upon the simplifying assumption that the supersonic primary nozzle flow adjusts isentropically to match the secondary flow static pressure. The analytical predictions of jet pump pressure rise and thrust augmentation based upon this assumption agree closely with test results when mixing tube wall friction effects are taken into account. The agreement is good over a very broad range of operating conditions.

2. Mixing Tube Wall Friction:

The stagnation pressure losses which occur in the mixing tube due to wall friction are significant in comparison to the stagnation pressure rise developed by a high-entrainment jet pump. Measurement of the wall friction losses by drawing air through the mixing tube with a blower may yield values of the friction loss coefficient which are accurate enough for design purposes. Further analysis of the data in this report is required to check this point.

3. Jet Pump-Duct System Matching:

A technique for selecting the optimum design for a jet pump to match given operating conditions was presented in section 3.4. For the same primary flow conditions, the jet pump geometry to achieve maximum entrainment was shown to be different from the geometry required to achieve maximum thrust augmentation. In order to match a jet pump to its associated duct system to obtain peak performance, it is essential that the loss coefficients of the duct components be estimated as accurately as possible.

4. Optimum Mixing Tube Length:

The mixing tube length must be selected to balance the increased pressure recovery resulting from more complete mixing in a longer tube against the increased wall friction losses in the longer tube. The results of this investigation suggest that no simple length-to-diameter rule is applicable to mixing tube design. The optimum mixing tube length is a function of the primary flow conditions and the entrainment ratio at the operating point. This entrainment ratio is set by the loss characteristics of the jet pump duct system.

5. Nozzle Cluster Design:

The original position of the nozzle cluster close to the mixing tube inlet led to increased losses and inferior jet pump performance. When the nozzle cluster was moved upstream away from the mixing tube inlet, its pressure loss and blockage effects were minimized and the performance of the jet pump was measurably improved. A special "low-drag" nozzle cluster design was no better than the original cluster when both were-positioned away from the mixing tube inlet.

A reduced centerline velocity appeared in most of the velocity profiles measured at the mixing tube exit and in the conical diffuser. The cause of the reduced centerline velocity is unknown.

6. Performance of the Wagner Jet Induced Lift System:

The loss characteristics of the suction duct and nozzle cluster arrangement used in the NAS 2-2518 test program were measured during this program. The high suction duct and nozzle cluster losses, incomplete mixing in the too-short mixing tube, and non-optimum matching of the jet pump to the duct system all acted to reduce the entrainment ratios achieved with the system substantially below the peak values attainable.

7. Mixing Tube Design:

The original mixing tube geometry, which had a small reduction in cross-sectional area along its length, produced a slightly higher entrainment ratio than the constant-area mixing tube under similar test conditions. Further analytical and experimental work should be carried out to determine the performance characteristics of a variety of mixing tube shapes. Significant performance improvements may be possible if mixing tubes other than the conveniently-analyzed constant area and constant pressure designs are used.

APPENDIX A

Listing of the Computer Program

An "*" before a line indicates that it can be removed if there is to be no plotting with the EZPLOT subroutine.

DIMENSION PRISE(11,15), THRUS(11,15), ENT(11,15), ARAY(11,15) 0001 0002 DIMENSION COUNT(15), X(15), Y(15), PTITLE(15), TTITLE(15), NC(9) 0003 DIMENSION CHAR1(11,6), CHAR2(11,5) 0004 DATA NC/38,63,16,55,44,19,24,52/ С CLEAR ARRAYS 0005 DO 26 K=1.15 0006 COUNT(K)=0X(K) = 00007 0008 Y(K) = 00009 PTITLE(K)=00010 TTITLE(K)=00011 DO 25 J=1,11 0012 PRISE(J,K)=0 0013 THRUS(J,K)=0 ENT(J,K)=00014 0015 ARAY(J,K)=00016 25 CONTINUE 0017 26 CONTINUE CALL IDFRMV REMOVED FROM PROGRAM HERE С Line 1 0018 READ(5,350)NTP,(PTITLE(I),I=1,15) Line 2 0019 READ(5,450)NTT,(TTITLE(I),I=1,15) Line 3 0020 READ (5,52) N 0021 G=1.4 0022 G0=32.2 0023 R=53.35 0024 CONV=778.16 0025 K=0 С BEGIN OUTER LOOP, EACH LOOP USES A NEW SET OF INITIAL CONDITIONS 0026 1 WRITE(6,100) 0027 DO 761 J=1,11 0028 DO 760 M=1.5 0029 CHAR1(J,M)=00030 CHAR2(J,M)=00031 760 CONTINUE 0032 761 CONTINUE 0033 CHAR1(11,6)=00034 K = K + 1READ DIMENSIONLESS INITIAL CONDITIONS С 0035 READ(5,50,END=300)PBAR,TBAR,ABAR1,ABAR2 Line 4 С READ DIMENSIONAL INITIAL CONDITIONS 0036 READ(5,51)PSOI,TSO,AM Line 5 0037 Line 6 READ (5+455) FDUCT+ FTUBE 0038 WRITE(6,101)PBAR, TBAR 0039 WRITE(6,102)ABAR1,ABAR2 0040 WRITE(6,103)PSDI,TSD,AM 0041 WRITE(6,456)FDUCT,FTUBE С CALCULATE OTHER DIMENSIONAL VALUES FROM DIMENSIONLESS VALUES 0042 PPO=PBAR*PSOI

11.1

.

. .

Ar.,

0043 0044		TPO=TBAR*TSO ATH=ABAR1*AM
0045		AP=ABAR2*AM
	C C	BEGIN INNER LOOP, EACH TIME THROUGH GIVES A SET OF SOLUTIONS FOR A DIFFERENT VALUE OF ENTRAINMENT RATIO.
0046	Line 7	DO 13 J=1,11
0047		ENTR=7•0+J*3•0
0048		WRITE(6,105)J,ENTR
	C	CALCULATE PRIMARY AND SECONDARY MASS FLOW RATES
0049		$WP=G*GO*(1/R)*(2 \cdot O/(G+1))**((G+1)/(G-1))$
0050		WP = SQRT(WP) + 144 + 0 + 60 + 0 + PPO + ATH/SQRT(TPO)
0051		WS=ENTR∻WP
0052	Line 8	PSD=PSDI-FDUCT*WS*WS*R*TSD
		1/(3600.0*2.0*G0*PSOI*AM*AM*144.0*144.0)
	Line 9 C	ITERATE TO FIND SECONDARY MACH NUMBER
0053		820 I=0
0054		GUESS=0.2
0055		Z = W S + SQRT(TSD) / (60, 0*) + 4, 0* PSO + (AM - AP) + SQRT(G+G0/R))
0056		2 PARAM= $1+(G-1)/2 \cdot 0$ *GUESS*GUESS
0057		ZCALC=GUESS/PARAM**((G+1)/(2.0*(G-1)))
0058	Line 10	IF (GUESS-1, 0) 14, 14, 15
0059		14 IF(I-100)3,3,5
0060		3 IF (ABS(ZCALC-Z) 0005)6.6.4
0061		4 DERIV=PARAM**(-(G+1)/(2.0*(G-1)))-(G+1)/2.0*GUESS*GUESS
		1*PARAM**((1-3.0*G)/(2.0*(G-1)))
0062		GUE SS=GUE SS-(ZCALC-Z)/DER IV
0063		I = I + 1
0064		GO TO 2
0065		5 WRITE(6,66)
0066		GD TO 405
0067		15 WRITE(6,115)
0068		GO TO 13
0069		6 CONTINUE
0070		SMD K=GUESS
0071		PS=PSO/(1+(G-1)/2.0*SMOK*SMOK)**(G/(G-1))
0072		TS=TSO/(1+(G-1)/2•0*SMOK*SMOK)
0073		VS=2•0*G*G0*R*(TSO-TS)/(G-1)
0074		VS=SQRT(VS)
0075		PSOR=PSOI-FDUCT*VS*VS*PS/(2.0*GO*R*TS)
0076		IF(ABS((PSOR-PSO)/PSOR)0005) 800,800,810
0077		810 PSO=PSOR
0078		GO TO 820
0079		800 PSO=PSOR
0080		WRITE(6,457)PSO
0081		WRITE(6,600)I
	С	MAKE CORRECTION FOR UNDER OR OVER EXPANSION
0082		I = 0
0083		16 PMOK=(1+(G-1)/2.0*SMOK*SMOK)*(PPO/PSO)**((G-1)/G)-1.0
0084		$PMOK = SQRT(2 \cdot O * PMOK/(G-1))$
0085		AM2=₩S*SQRT(TSD)*(1+(G-1)/2•O*SMOK*SMOK)**((G+1)/(2•O*(G-1)))
		1/(P SO*SMOK)
0086		AM2=AM2+WP*SQRT(TPD)*(1+(G-1)/2.0*PMOK*PMOK)**((G+1)/(2.0*(G-1))) 1/(PPD*PMOK)

0087 0088 0089 0090 0091 0092 0093	17	AM2=AM2*SORT(R/(G*G0))/(144.0*60.0) IF(ABS(AM2-AM)00005) 19,19,17 7 SMOK=SMOK+((AM2-AM)/ABS(AM2-AM))*FXP(5.2*SMOK*SMOK)*5.E-5 IF(I-200)18,5,5 3 CONTINUE I=I+1 GO TO 16
0094 0095 0096 0097	19	<pre>> CONTINUE WRITE(6,601)I AP=(1+(G-1)/2.0*PMOK*PMOK)**((G+1)/(2.0*(G-1))) AP=AP*WP*SQRT(TPO)/(144.0*60.0*PPO*PMOK*SQRT(G*G0/R))</pre>
0098 0099 0100	С	CALCULATE CONDITIONS AT END OF ACCOMODATION REGION PS2=PS0/(1+(G-1)/2.0*SMOK*SMOK)**(G/(G-1)) TS2=TS0/(1+(G-1)/2.0*SMOK*SMOK) VS2=2.0*G*G0*R*(TS0-TS2)/(G-1)
0101 0102 0103 0104		VS2=SORT(VS2) PP2=PS2 TP2=TPO*(PP2/PPO)**((G-1)/G) VP2=2_0*G*G0*R*(TPO-TP2)/(G-1)
0105 0106 0107	~	VP2=SQRT(VP2) C2=SQRT(G*G0*R*TP2) PM0K=VP2/C2
0108 0109 0110	L	WM=WP $*(1+ENTR)$ CP2=.24914 CS2=.24914
0111 0112 0113 0114		I=0 HP2=CP2*TP2+VP2*VP2/(2.0*G0*C0NV) HS2=CS2*TS2+VS2*VS2/(2.0*G0*C0NV) INT=0
0115 0116 0117	7	PMG=PS2+.4 VM=VP2+ENTR*VS2+60.0*144.0*G0*AM*(PS2-PMG)/WP VM=VM*WP/WM
0118 0119 0120 0121		TM=60.0*144.0*PMG*VM*AM/(R*WM) CM=.24914 PM=(HP2+ENTR*HS2)*R*WP/(60.0*144.0*VM*AM*CM) PM=PM~R*WM*VM/(2.0*144.0*60.0*G0*CONV*CM*AM)
0122 0123 0124 0125	8	IF(I-800) 8,8,15 IF(ABS(PMG-PM)01)11,11,9 IF(PM .GT. PMG) GO TO 20 IF(INT FO. 1.) GO TO 11
0126 0127 0128		PMG=PMG+(PMG-PM)/PMG I=I+1 GG TO 7
0129 0130 0131 0132	20	INT=1 PMG=PMG001 I=I+1 GD_TO_7
0133 0134 0135	10 11	WRITE(6,66) GO TO 405 CONTINUE
0136		PM=PMG

•

[A

0138					EMDK=VM/SQRT(G*GO*R*TM)
	Line	12	С		CALCULATE THE EFFECT OF WALL FRICTION
0139					GAM1=1.0+(G-1)/2.0*EMOK*EMOK
0140					ALPH=144.0*PM*(GAM1**(G/(G-1))-FTUBE*EMOK*EMOK*G/2.0)
0141					BETA=ALPH/SQRT(144.0*PM*144.0*PM*EMOK*EMOK*GAM1)
0142					
0143					GMUK=EMUK
0144				700	$GAM 2= I \bullet O + (G - I) / 2 \bullet O * GMUK * GMUK$
0145					$A = \{1/GMUK\} \neq GAM2 \neq \neq \{(G+1)/(2, O\neq (G-1))\}$
0146					UA=(6+1)/2•0*6AM2**((3=6)/(2•0*(6=1)))
0147					UA=UA=(1/(GMUK*GMUK))*GAM2**((G+1)/(2.0*(G=1)))
0148					1F(1 .61. 200)60 10 10
0149				-	IF (GMUK-1.0)/01,/01,15
0150				701	IF(ABS(A-BEIA)0001)/03,/03,/02
0151				702	1=1+1
0152					GMOK=GMOK-(A-BETA)/DA
0153					GU TU 700
0154				703	CONTINUE
0155					WRITE(6,604)I
0156					P2=(1/144.0)*ALPH/GAM2**(G/(G-1))
0157	- .		_		TM=TM*(P2/PM)**2.0*(GMUK/EMUK)**2.0
	Line	13	С		CALCULATE OUTLET PARAMETERS
0158					VM=GMDK*SQRT(G*GO*R*IM)
0159					PM=P2
0160					
0161					PM U = 144.0*PM*(1.0+(G-1)/2.0*EMUK*EMUK)**(G/(G-1))
0162					DELP=PMIUI-144.0*PSU
0163					$DDELP=DELP/(144 \cdot 0 \neq PSU)$
0164					$DSTAT = I44 \bullet 0 \% (PM - PSU)$
0165					DDSTA=(PM-PSU)/PSU
0166					
0167					ENER=144.0*PM*VM*VM/(2.0*G0*R*IM)
0168					DENER=ENER/(I44.0∓PSU)
0169					NDELP=DELP
0170			_		IF (NDELP .LE. 0) GU IU I
			£		STORE SOLUTIONS FOR TABLE PRESENTATION
0171					SUM=J
0172					CHAR1(J, 1) = ENTR
0173					CHAR1(J,2)=DDELP
0174					CHAR1(J,3)=DDSTA
0175					$CHAR1(J_{4}4) = EMUK$
0176					CHARI(J,5)=UENER
0177					$LHAR\left[\left(\mathbf{J},6\right)=AUG\right]$
0178					CHAR2(J, I) = ENIR
0179					
0180					
0181					
0182			~		CHAKZ(J))=ENEK
0105			U		STUKE SULUTIONS IN AKKAIS TO BE PLUTTED LATER
0183					
0184					
0185					ENILJ,KJ=ENIK
0186					COUNF(K)=J

- --

l

.

0187 0188 0189 0190 0191 0192 0193 0194 0195 0196 0197 0198					WRITE(6,106) WRITE(6,107) WRITE(6,108)PP2,PS2,PM WRITE(6,109)TP2,TS2,TM WRITE(6,110)VP2,VS2,VM WRITE(6,111)PMOK,SMOK,EMOK WRITE(6,112)WP,WS,WM WRITE(6,112)WP,WS,WM WRITE(6,113)DELP WRITE(6,120)DDELP WRITE(6,120)DDELP WRITE(6,120)DDELP WRITE(6,114)AUG
			С		END OF INNER LOOP
0199	Line	14		13	CONTINUE
			С		PRINT OUT TABLE OF RESULTS
0200				765	WRITE(6,707)
0201					WRITE(6,710)
0202					WRITE(6,751)PBAR,TBAR,ABAR1
0203					WRITE(6,707)
0204					WRITE(6,720)
0205					WRITE(6,730)
0206					M= SUM
0207					DO 725 J=1,M
0208					WRITE(6,704)CHAR1(J,1),CHAR1(J,2),CHAR1(J,3),CHAR1(J,4),CHAR1(J,5)
				1	L+CHAR1(J+6)
0209				725	CONTINUE
0210					WRITE(6,707)
0211					WRITE(6,750)
0212					WRITE(6,751)PBAR, TBAR, ABAR1
0213					WRITE(6,752)PS0,TS0,AM
0214					WRITE(6,707)
0215					WRITE(6,753)
0216					WRITE(6,754)
0217					DO 726 J=1,M
0218					WRITE(6,755)CHAR2(J,1),CHAR2(J,2),CHAR2(J,3),CHAR2(J,4),CHAR2(J,5)
0219				726	CONTINUE
			С		END OF OUTER LOOP
0220	Line	15			IF(K .GE. N) GO TO 300
0221					GO TO 1
0222				300	CONTINUE
			С		FIND LARGEST VALUES IN ARRAYS PRISE AND THRUS
0223					PMAX=TOP(PRISE)
0224					TMAX=TOP(THRUS)
			С		ENTER PLOTTING SECTION OF THE PROGRAM
0225					κ=1
0226					NA=17
0227					N0=27
0228					NF=1
0229					GO TO 303
0230				301	NTP=0
0231					NA=0
0232					N0=0
0233					NF=2

•

Š.

86

_

0234		302 K=K+1
0235		303 M=COUNT(K)
0236		00 305 J=1,11
0237		0=(L)X
0238		V (1) = 0
0239		305 CONTINUE
0240		00 304 J=1.M
0241		$X(J) = FNT(J \cdot K)$
0241		$Y(1) = PRISE(1) \cdot K$
0242		
0245		
0244	c	CALL FZELOT REMOVED FROM DECK HERE
0245	U.	F(k) = GE = N) GO TO 400
0245		I = (K - 2) = 202 = 302
0240		
0241		
0240		
0247		
0250		
0251		
0252		
0253		
0254		
0255		
0256		402 K = K + 1
0257		
0258		DU = 406 J = 1 + 11
0259		X (J) = 0
0260		Y(J)=0
0261		406 CONTINUE
0262		DD 404 J=1,M
0263		X(J) = ENT(J,K)
0264		Y(J)=THRUS(J,K)
0265		404 CONTINUE
0266		LAST=1
0267		IF(K •GE• N)LAST=2
	С	CALL EZPLOT REMOVED FROM DECK HERE
0268	_	IF(K •GE• N) GD TD 405
0269		IF(K-2)+01+402+402
0207	C	CALL PLIND REMOVED FROM DECK HERE
0270	Ť	405 CONTINUE
0271		50 EDRMAT(4E10.5)
0272		51 FORMAT(3F10.5)
0273		52 EORMAT(110)
0274		100 FORMAT(//25X-65H**** THE FOLLOWING CASES WILL USE THESE INITIAL C
		INDITIONS ****
0275		101 CODMAT/22, ISUDECCUDE DATIO-, ELO A LOY LOUTEMDEDATURE DATIO
0215		101 FURMALLYZAY LIMPRESSUKE KALLU-YELU-YYLUAYLUAHIMEEMPEKALUKE KALLU-Y
0.27/		102 CORMAT/ /2X 120/ATU#CU/ANA E10 / 12X /040/ANA E10 / 1
0276		102 rukmai(/2X,12H(A)H π UW/AM)=, $E10-4$, $13X$, $OHAP/AM=$, $E10-4$, 102 , $COPATING ADV - COPATING ADV -$
0211		103 FURMATI/2X,19HSECUNDARY PRESSURE=,EIU.4,3X,22HSECUNDARY TEMPERATUR

	1E=,E10.4,3X,17HMIXING TUBE AREA=,E10.4)
0278	105 FORMAT(/40X,10H**** CASE ,I2,2X,18HENTRAINMENT RATIO=,F10.5,
	15H ****)
0279	66 FORMAT(/2X,19HTOO MANY ITERATIONS)
0280	106 FORMAT(/39X,19HACCOMODATION REGION,10X,1H*,10X,6HOUTLET)
0281	107 FORMAT(34X,7HPRIMARY,11X,9HSECONDARY,7X,1H*)
0282	108 FORMAT(/2X, 13HPRESSURE, PSIA, 15X, E10.4, 10X, E10.4, 8X, 1H*, 6X, E10.4)
0283	109 FORMAT(/2X,11HTEMPERATURE,17X,E10,4,10X,E10,4,8X,1H*,6X,E10,4)
0284	110 FORMAT (/2X · 15HVFLOCITY · FT/SEC · 13X · F10 · 4 · 10X · F10 · 4 · 8X · 1H * • 6X · E10 · 4)
0285	111 FORMAT(/2X,11HMACH NUMBER,17X,F10,4,10X,E10,4,8X,1H*,6X,F10,4)
0286	112 FORMAT (/2X-24HMASS FLOW RATE POUND/MIN-4X-F10-4-10X-F10-4-8X-1H*-
0200	16X • E10.4)
0287	113 EDRMAT(/5X-20HTDTAL PRESSURE RISE=.F10.4.2X.14HP0UND/S0. EDDT)
0288	114 FORMAT($/5x_15$ HOMENTIM RATIO=.F10_4)
0289	115 FORMAT(/5X,47HTHIS VALUE OF ENTRAINMENT RATIO IS INACCESSIBLE)
0200	116 FORMAT($/5$, 434 THE DETMARY STREAM AREA AFTER ACCOMODATIONS FIG. 4.
0230	114.04CO EETI
0201	12A JOHSQUILLET J
0291	120 FURMAT(7)A;20001MENSIUNLESS PRESSURE RISE-;EIU.44) 250 CODMAT(7)E 1544)
0292	500 FURMAT(115,1544)
0293	490 FURMAI (115,1544)
0294	600 FURMAI (/5X, 23HMACH NUMBER ITERATIONS=,13)
0295	601 FURMAI(/5X,24HPRIMARY AREA ITERATIONS=,13)
0296	602 FORMAT(75X,20HPRESSURE ITERATIONS=,13)
0297	603 FORMAT(/5X,37HDIMENSIONLESS STATIC PRESSURE CHANGE=,E10.4)
0298	707 FORMAT(/1X,116H* * * * * * * * * * * * * * * * * * *
	1 * * * * * * * * * * * * * * * * * * *
0299	710 FORMAT(15X,54HDIMENSIONLESS SOLUTION USING THESE INITIAL CONDITION
	15,)
0300	720 FORMAT(/3X,11HENTRAINMENT,4X,1H*,2X,19HDIMENSIONLESS TOTAL,3X,1H*,
	12X,20HDIMENSIONLESS STATIC,2X,1H*,4X,4HMACH,6X,1H*,2X,13HDIMENSION
	2LESS,4X,1H*,3X,8HMOMENTUM)
0301	730 FORMAT(3X,5HRATIO,10X,1H*,2X,13HPRESSURE RISE,9X,1H*,2X,15HPRESSUR
	1E CHANGE,7X,1H*,4X,6HNUMBER,4X,1H*,2X,14HKINETIC ENERGY,3X,1H*,3X,
	25HRATIO)
0302	704 FORMAT(/8X,F4.1,6X,1H*,4X,F10.5,10X,1H*,5X,F10.5,9X,1H*,4X,F5.3,
	15x • 1H * • 3X • F7 • 4 • 9X • 1H * • 3X • F7 • 3)
0303	750 FORMAT(/5X-52HDIMENSIONAL SOLUTION USING THESE INITIAL CONDITIONS.
0505	
0304	751 FORMAT(5%, 15HORESSURE RATIO=, 55, 2, 3%, 18HTEMPERATURE RATIO=, 55, 2, 3%
0504	1. 120(ATUER) (AN)- EQ ()
0305	TY LENGATING WART-TY CONDADY DESCRIPE-LES 2 1Y 20051 2Y 2205CCONDADY TE
0505	INDEDATION ET 2. 1V IANDERDE DANVINE 2V IZMAVINE THE ADEAL ED L
	TMPERATURE - FF / 29 IA 1 4 DEGREE RANKINE 93 A 11 MMIAING TUDE AREA = 7 8 0 0 9
0204	21A) (HOWS FIS) 752 FORMAT (29) INTENTRATIMENT OF 100 OF INTENTS STATIC SY 100 OF INTENTS
0300	133 FURMALLY 3A FILMEN RAINMENT $f(2A)$ Int $f(2A)$ ILLEAL STATLE $f(2A)$ ILLEAL STATLEAL
	TIAL PRESSURE, 2A, TH*, 2A, TOHSTATIC PRESSURE, 2A, TH*, 2A, T4HKINETTC
0207	ZKOT) 764 CONNATION EURATIO ON 114 ON 14100555105 (DST) ON 114 ON 1010555 (D
0507	124 FURMAITSAJORATIUJSAJIM+JZAJIMHRESSUKE (PSIJJZAJIM+JZAJIUHKISE (P
0200	$13\Gamma I J O A J I T T Z Z Z I L T U H A N G E (Y S F I J Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z$
0508	()) FUKMAI 1/88,574.1,48,1H*,48,F5.2,98,1H*,38,57.4.2,88,1H*,38,59.4,78,
0200	$11\pi\tau_{1}2\lambda_{1}\tau_{1}(\epsilon)$
0210	OUT FURMATIONAL CONTRICTION ITERATIONS=,13)
0211	400 FURMALLET 10.01
1160	430 FURMAII/2X;23HINLEI LUSS CUEFFICIENI=;EIU.4;3X;24HMIXING LUSS CUEF
	1010101-9010041

۰.

• •

--- · ··

···· - ·· ···

0312	457	FORMAT(/5X, 39HSECONDARY TOTAL PRESSURE BEFORE MIXING=, E10.4)
0313		END
0001		FUNCTION TOP(ARAY)
0002		DIMENSION ARAY(11,15)
0003		TOP=ARAY(1,1)
0004		DO 501 L=1,11
0005		DD 500 M=1,15
0006		IF(TOP .LT. ARAY(L,M)) TOP=ARAY(L,M)
0007	500	CONTINUE
0008	501	CONTINUE
0009		RETURN
0010		END

_

APPENDIX B

1

B.1 Discussion of Computer Program by Blocks

Block 1 - Declare and Clear Arrays, Set Values of Constants:

To make sure that the storage arrays used in the program are all empty, zeroes are placed in every location by this block of the program.

Numerical values of G, GO, R, and CONV are also defined in this section of the program. The numerical values are not read in as input; they are defined within the program. To change them, the appropriate cards in the deck must be changed.

Block 2 - Read Initial Conditions and Initialize Parameters:

In the segment of the program between line 1 and 6, the initial conditions are read in and prepared for further calculations. These initial conditions include data required by the plotting routine.

Titles for plots of dimensionless pressure rise and momentum ratio as functions of entrainment ratio (PTITLE, TTITLE) and the number of cases to be solved (N) are introduced in lines 1, 2 and 3 respectively. The number preceding each title is the number of characters and spaces in the title. Dimensionless initial conditions (PBAR, TBAR, ABAR1 and ABAR2) are read in on line 4. Although ABAR2 has not been used in the formulation, it is used as an initial guess for an iteration loop as described in more detail in the discussion of block 5. Since a value of k has been specified in block 1, the initial conditions for a dimensionless solution are complete at this point.

Values of PSOI, TSO and AM are read in on line 5, and values of friction coefficients (FDUCT, FTUBE) are introduced at line 6. This completes the information required to begin calculations. Specific values of PPO, TOI, ATH and AM are obtained by multiplying the dimensional initial conditions by the nondimensional initial conditions.

Block 3 - Calculation of Suction Duct Pressure Loss:

The performance of a jet pump is dependent upon the stagnation pressure of the secondary flow at the exit of the primary nozzles, PSO. It is difficult to measure this pressure in an experimental jet pump. Instead, the suction duct inlet stagnation pressure can be measured and the loss between the inlet section and the primary nozzle exit section can be accounted for by the following equation:

$$\Delta P_{t} \right) \text{ suction duct} = K_{sd} \frac{\rho_{s} V_{s}^{2}}{2 g_{o}}$$
(97)

The loss coefficient K_{sd} can be evaluated by drawing air through the suction duct with a blower and plotting (ΔP_t) suction duct $vs. \rho_s V_s^{2/2} g_o$ on a Cartesian graph. The slope of the resulting curve is equal to K_{sd} .

The correction for suction duct pressure loss begins on line 8 of the program. The equation for this correction is given below.

$$P_{so} = P_{soi} - K_{sd} \frac{\rho_s V_s^2}{2g_o}$$
(98)

where P_{soi} is the secondary flow stagnation pressure at the suction duct inlet.

Blocks 4 and 10 - Specification of Entrainment Ratio:

Line 7 initiates a DO loop which encloses all statements down to line 14 (statement number 13). The entrainment ratio (ENTR) is initially set equal to 10.0 and the loop is repeated, increasing ENTR by 3.0 each time through, until a value of 40.0 has been reached. The integer J is used to count the number of loops completed.

٥.

Block 5 - Solution of Equations for the Accommodation Region:

Block 4 extends between line 9 and line 11. Primary and secondary mass flow rates (WP, WS) are calculated using equations (1) and (2). The value of the secondary flow Mach number (SMOK) is obtained by solving equation (57). Equation (57) is solved using Newton's method. A series of values of M_{s2} are tried until one is found which makes the left side of equation (57) sufficiently close in value to the right side. The numerical criterion for acceptable convergence is agreement within 0.0005, which gives a secondary flow rate accurate to within 0.5 lbm/min.

The primary mass flow rate and the entrainment ratio are specified in this program before the secondary Mach number is evaluated. There exists a maximum possible secondary mass flow rate corresponding to a secondary Mach number of 1.0. Thus, for given initial conditions, the jet pump will have a maximum permissible value of entrainment ratio, which can be calculated from equation (64). The statement on line 10 was inserted to recognize trial values of M_{s2} which are greater than 1.0. If the program tries to obtain a solution for an entrainment ratio greater than the maximum possible for the system, a message "This value of entrainment ratio is inaccessible" is printed and the program proceeds to block 9 (line number 15) to look for another set of initial conditions.

When the secondary Mach number has been determined, the static pressure at the end of the accommodation region ($P_{p2} = P_{s2}$) is calculated using P_{s0} and M_{s2} in equation (3). Then equation (3) is used with P_{p0} and P_{p2} to calculate M_{p2} . This procedure neglects the effect of primary stream expansion or contraction upon the secondary stream area. Equation (6) is used to correct for primary flow area

changes as follows. The calculated values of M_{p2} and M_{s2} are substituted into the left side of equation (6) and the result is compared with the actual mixing tube area. If the values are not sufficiently close, the value of M_{s2} is changed slightly, a new value of M_{p2} is determined from equation (3), and the test to see if equation (6) is satisfied is repeated. When values of M_{s2} and M_{p2} which satisfy equation (6) are determined, equations (7), (8), (9), (10), and (11) are employed to calculate the pressure, temperature and velocity for each stream at the end of the accommodation region.

The solution of equation (57) for M_{s2} requires a value of A_p , the area of the primary nozzle flow, which is introduced as an initial condition through the dimensionless area ratio ABAR2. However, this value of M_{s2} is used only temporarily; it is eventually modified as the effect of primary area changes is considered. Therefore, even though the analytical formulation did not make use of ABAR2, the program is made more efficient by using it to obtain a preliminary value of M_{s2} .

Block 6 - Solution of Equations Representing the Mixing Region:

The segment of the program which solves the mixing region equation extends from line 11 to line 12. The solution is obtained by an iteration technique in which the value of one variable is assumed and the equations are solved to obtain a calculated value of that variable. When the trial value is sufficiently close to the calculated value, an acceptable solution has been determined.

Values of stagnation enthalpy (HP2, HS2) are calculated for the primary and secondary streams using their properties at the end of the accommodation region. A trial value of PM (PMG = PSO + 0.4) is chosen and equations (19), (18), (17) and (20) are solved sequentially to determine a calculated value of PM. If the trial and calculated values satisfy the criterion below, the trial value is accepted as the solution.

 $|PMG - PM| \leq 0.01$

If this convergence criterion is not satisfied, a new trial value is chosen and the procedure is repeated. The convergence criterion corresponds to an error of less than 0.1%.

The integer I keeps track of the number of iterations performed in order to limit their number to 400. Convergence is generally obtained in less than 100 iterations. In some cases, the trial value will oscillate about the calculated value without quite converging. When this occurs, the trial value is within a region very close to convergence. Therefore, the logical variable INT is used to detect this condition, stop the iteration, and accept the current trial value as a satisfactory solution.

In the course of the development of a satisfactory value of PM, corresponding values of VM and TM are determined. These three values completely specify the conditions at the end of the mixing region. If no frictional effects are included, all the desired jet pump performance parameters such as stagnation pressure rise, static pressure change and momentum ratio can be determined from PM, VM, and TM.

For the temperature ranges of interest, the variation of specific heat with temperature has been found to have a negligible effect on the results (figure 5). This computer program, therefore, has been written to treat specific heat and k as constants. As a result, equation (17) has not been used in the program. If it is desired to include a variable specific heat, an equation such as (17) can be added to the existing program without altering its basic structure.

Block 7 - Solution of Equations Representing Mixing Region Friction Effects:

Block 7, from lines 12 to 13, corrects the values of M_m , P_m , T_m and V_m for frictional effects. Equation (38) is used to accomplish this. For a specified value of mixing duct friction coefficient, K_{MT} , both α and β can be determined from the results of block 6. Equation (38) is then solved using Newton's method to give a new value of Mach number at location m. A new value of pressure is obtained from equation (34). Corrected values of temperature and velocity are then determined using equations (39) and (33).

Block 8 - Print Solutions for One Value of Entrainment Ratio:

When a solution for a particular value of entrainment ratio is obtained, all variables are printed together with appropriate titles. Data internal to the program

is also printed, e.g., the number of iterations required for convergence in each section in which an iterative method is used. When a new set of initial conditions are used, this is indicated by an appropriate statement and a list of the new initial conditions.

Block 9 - Store Solutions for Plots and Tabular Presentation:

The results which are printed as each solution is obtained are also summarized in tabular form. To allow the tabular form to be printed, arrays are filled with the numbers from each solution.

Arrays of the same type are used to store solutions for the plotting routine. These arrays are larger because the plotter is used for several sets of initial conditions whereas the tabular results are printed for each individual set of initial conditions.

Block 11 - Print Solutions in Tabular Form:

After a set of solutions for one set of initial conditions has been obtained and stored in the arrays of block 9, the results are printed in a table. Values of the desired performance parameters are printed vs. entrainment ratio. Two tables are printed; one presents dimensionless variables and the other presents dimensional variables. The dimensionless solutions are independent of the dimensional initial conditions unless frictional effects have been included in the solution.

Block 13 - Plot Dimensionless Pressure Rise and Momentum Ratio:

Values which have been stored in the arrays PRISE and THRUS in block 9 are plotted as functions of entrainment ratio. Two plots are obtained; dimensionless pressure rise and momentum ratio vs. entrainment ratio. If it is desired to plot variables other than these, this may be done by storing the desired variables in the PRISE and THRUS arrays in place of the dimensionless pressure rise and momentum ratio values.

The plotting system is too complex to discuss in detail here. Unless the user has access to the EZPLOT routine used in this program, it is unlikely that he

will be able to use the plotting section directly. However, the storage arrays developed within the program should be useful for providing data to any other plotting device which may be available.

B.2 Using the Computer Program

This section discusses the various options which may be exercised when using the program and the tasks which must be performed to set them up.

The Complete Program:

The complete program includes suction duct and mixing tube frictional effects and plotting of the results. Values of PBAR, TBAR, ABAR1, ABAR2, PSOI, TSO, and AM must be provided as input data together with empirically-determined values of FDUCT and FTUBE. A value of N equal to the number of cases to be solved must be included to maintain control within the plotting block. It is best to vary only one parameter such as \overline{A} in a single set of solutions. The plot titles (which are provided as input data) can then denote this parameter on the plots as illustrated in the sample solutions provided in section 3.3.

If variables other than DDELP and AUG are to be plotted, this may be done by storing them in the PRISE and THRUS arrays.

Omission of Frictional Effects:

Frictional effects may be omitted by inputting zero values for FDUCT and/or FTUBE, depending on which frictional effect is to be eliminated.

Omission of the Plotting Section:

The plotting function may be temporarily omitted by punching a character in the first column of the "CALL EZPLOT" cards. The necessary input data for plotting must still be read-in or else the input format will not function correctly. If the plotting section is to be removed permanently, it is best to remove all cards associated with the plotting function. This includes READ statements, logic for data manipulation, storage arrays, and the function TOP (ARRAY). The cards which may be removed are noted by an asterisk on the program listing of Appendix A.

- -

APPENDIX C

Typical Computer Solutions

_

MOMENTIA PATIO=0.1302F DL

DIMENSIONLESS STATIC PRESSURE CHANGE=0.1102F-01

DINESSIONESS PRESSURE RISES. 20256-01

TOTAL PRESSURE PISE=0.4251F 07 POUND/SQ. FOOT

THE PRIMARY STREAM AREA AFTER ACCOMMONATION=0.3303E-03 SQ. FEET

	PRIMARY	SECONDARY	•	
PRESSURE.PSIA	0.1424F 0?	0.1474F 02	•	0.1476F 07
TEMPERATURE	0.29975 03	0.5262F 03	•	0.5433E 0?
VFLOCITY+FT/SEC	0.74435 04	0.21255 03	•	0.2272F 03
MACH MIJMBER	0.29926 01	0.1889F 00	•	0.1987F 00
MASS FLOW CATE, POUND/MIN	0.6230E 01	0.8099F 07	•	0.P777F 07

ACCOMODATION	PEGION

PRESSURE ITERATIONS= 4

PRIMARY AREA ITERATIONS= 0

MACH NUMBER ITERATIONS= 1

**** CASE 2 ENTRAINMENT RATIO= 13.00000 ****

٠

OUTLET

.

MOMENTUM RATID=0.0049E 00

DIMENSIONLESS STATIC PRESSURE CHANGE=0.7251F-01

DIMENSIONLESS PRESSURE RISE=0.3988E-01

TOTAL PRESSURF RISE=0.8384E 02 POUND/SQ. FOOT

THE PRIMARY STREAM AREA AFTER ACCOMODATION=0.3282E-03 SQ. FEFT

	ACCIMODATI	ON REGION	•	OUTLET	
	PRIMARY	SECONDARY	٠		
PRESSUPE+PSTA	0.1439E 02	0.1439E 0?	•	0.1493E 07	
TEMPERATURE	0.2996E 03	0.5278F 03	٠	0.549RE 03	
VELOCITY, FT/SEC	0.2441E 04	0.1624F 03	•	0.1786F 03	
MACH NUMBER	0.28758 01	0.1441F 00	•	0.1553F 00	
MASS FLOW RATE, POUND/MIN	0.6230E 01	0.6230E 02	•	0.6853F 07	

PRIMARY AREA ITERATIONS= 10 PRESSURE ITERATIONS= 73

MACH NUMBER ITERATIONS= 2

**** CASE ? ENTRAINMENT PATIO= 10.00000 ****

PRESSURE RATIO=0.3000F 02	TEMPERATURE RATIO=0.1500E 01	
(ATH*CW/A4)=0.1000F-07	AP/AM=0.9000E-02	
SECONDARY PRESSURF=0.1460F 0?	SECUNDARY TEMPERATURE=0.5300F 03	MIXING TUBE APPA=0.8726F-0

***** THE ENLINWING CASES WILL USE THESE INITIAL CONDITIONS *****

MACH NUMBER ITERATIONS= 1

PRIMARY AREA ITERATIONS= 6

PRESSURE ITERATIONS= 4

	ACCOMO	٠	OUTLET	
	PRIMARY	SECONDARY	٠	
PRESSURE + PSIA	0.1405E 02	0+1405E 02	٠	0.1455E 02
TEMPERATURE	0.29755 03	0.5242F 03	٠	0.5383E 03
VELOCITY,FT/SEC	0.2446E 04	0.2641E 03	٠	0.2773E 03
MACH NUMBER	0.2891F 01	0.2352E 00	٠	0.2437E 00
MASS FLOW RATE, POUND/MIN	0.6230E 01	0.9968E 02	٠	0.1059E Q3

THE PRIMARY STREAM AREA AFTER ACCOMODATION=0.3331E-03 SQ. FEET

TOTAL PRESSURE RISE=0.8131E 02 POUND/SQ. FOOT

DIMENSIONLESS PRESSURE RISE=0.3867E-01

DIMENSIONLESS STATIC PRESSURE CHANGE=-.3367E-02

MOMENTUM RATIO=0.1927E 01

**** CASE 4 ENTRAINMENT RATIO= 19.00000 ****

200

MACH NUMBER ITERATIONS= 2

PRIMARY AREA ITERATIONS= 0

PRESSURE ITERATIONS= 6

	ACCOMODATION REGION		٠	OUTLET
	PRIMARY	SECONDARY	٠	
PRESSURE, PSTA	0.1381F 02	0.1381E 07	٠	0.1429E 02
TEMPERATURE	0.2961E 03	0.5216F 03	٠	0.5336E 03
VELOCITY,FT/SEC	0.2449E 04	0.3177F 03	٠	0.3291E 03
MACH NUMBER	0.2903E 01	0.2836E 00	٠	0.2905E 00
MASS FLOW RATE+POUND/MIN	0.6230E 01	0.1184F 03	٠	0.1246F 03

THE PRIMARY STREAM AREA AFTER ACCOMUDATION≠0+3368F-03 SQ. FEET

TOTAL PRESSURE RISE=0.P030E 02 POUND/SQ. FPOT

DIMENSIONLESS PRESSURE RISE=0.3819E-01

DIMENSIONLESS STATIC PRESSURE CHANGE=-.2089E-01

MOMENTUM PATTO=0.2688F 01

OUTLFT

0.13628 02

0.5237E 03 0.4408F 0?

0.39275 00

0.1620F 02

8

MACH NUMBER ITERATIONS= 2

PRIMARY AREA ITERATIONS= 0

PRESSURE ITERATIONS= A

	ACCOMMONATION REGION			CUTLFT
	PRIMARY	SECONDARY	*	
PRESSURE, PSIA	0.1351E 02	0.1351F 07	*	0.1399F 02
TEMPERATURE	0.2943F 03	0.5184F 03		0.5289F 03
VFLOCITY,FT/SEC	0.2454E 04	0.37355 03		0.3834F 03
MACH NUMBER	0.29175 01	0.3345F 00		0.3399E 00
MASS FLOW RATE POUND/MIN	0.6230F 01	0.1371E 03	*	0-1433F 03

ACCOMODATION REGION

SECONDARY

0.1315F 02

0.5144F 03

0.4325F 03

0.3889F 00

0.1558F 03

DIMENSIONLESS STATIC PRESSURE CHANGE=-.4197E-01 MOMENTUM RATIO=0.3593E 01 **** CASE 6 ENTRAINMENT RATIO= 25.00000 ****

DIMENSIONLESS PRESSURE RISE=0.3780E-01

MACH NUMBER ITERATIONS= 2 PRIMARY AREA ITERATIONS= 0 PRESSURE ITERATIONS= 10

MASS FLOW RATE, POUND/MIN 0.6230F 01

MOMENTUM PATIO=0.4660F 01

TOTAL PRESSURE RISE=0.7884E 02 POUND/SQ. FOOT

DIMENSIONLESS STATIC PRESSURE CHANGE=-.6717E-01

DIMENSIONLESS PRESSURE RISE=0.3750F-01

PRESSURE PSIA TEMPERATURE

VELOCITY, FT/SEC

MACH NUMBER

TOTAL PRESSURE RISE=0.7946E 02 POUND/SQ. FOOT

THE PRIMARY STREAM AREA AFTER ACCOMODATION=0.3414E-03 SQ- FEET

PRIMARY

0.1315F 02

0.2970E 03

0.2459F 04

0.29355 01

THE PRIMARY STREAM AREA AFTER ACCOMODATION=0.3472E-03 SQ. FEFT

(E22046+b214	0.13-16 02	0.13512 02	•	0.1999
EMPERATURE	0.2943F 03	0.5184F 03		0. 5 ? 891
FLOCITY,FT/SEC	0.2454E 04	0.37355 03		0.3824
ACH NUMBER	0.29175 01	0.3345F 00	•	0.3399
ASS FLOW RATE, POUND/MIN	0.6230F 01	0.1371E 03	*	0.1433

MOMENTIN PATIS=0.7373F 0]

DIMENSIONLESS STATIC PRESSURE CHANGE=-.1340E CO

DIMENSIONLESS PRESSURE RISE=0.3729F-01

TOTAL PRESSURE PISE=0.7839E 0? POUND/SQ. FOOT

THE PRIMARY STREAM APEA AFTER ACCOMODATION=0.3545F-03 SQ. FEFT

	ACCOMODATION REGION		*	OUTLET
	PRIMARY	SECONDARY	*	
PRESSURE, PSTA	0.1219E 02	0.1218F 02	•	0.1264F 02
TEMPERATURE	0.28575 07	0.5033E 03	•	0.51115 03
VELOCITY, FT/SEC	0.2475F 04	0.5669F 03	*	0.570?F 03
MACH NIJMBER	0.29965 01	0.5153F 00	•	0.5143F 00
WASS FLOW RATE, POUND/MIN	0.6230F 01	0.1931F 03	•	0.10945 03

PRESSURE ITERATIONS= 16

PRIMARY AREA ITERATIONS= 0

MACH NUMBER ITERATIONS= 3

**** CASE B ENTRAINMENT RATID= 31.00000 ****

MOMENTUM RATIO=0.590RE 01

DIMENSIONLESS STATIC PRESSURE CHANGE=-.9734E-01

DIMENSIONLESS PRESSURE RISE=0.3733E-01

TOTAL PRESSURE RISE=0.7848E 02 POUND/SQ. FOOT

THE PRIMARY STREAM AREA AFTER ACCOMODATION=0.3547E-03 SQ. FEET

	PRIMARY	SECONDARY	•	
PRESSURE, PSTA	0.1272E 02	0.1272F 07		0.1318E 02
TEMPERATURE	0.28925 03	0.5095F 03	*	0.5179F 03
VELOCITY, FT/SEC	0.2466F 04	0.49655 03	*	0.5024E 0?
MACH NUMBER	0.2957F 01	0.4486E 00	•	0.45025 00
MASS FLOW RATE, POUND/MIN	0.6230F 01	0.1744E 03	*	0.1807F 03

ACCOMODATION REGION

PRESSURE ITERATIONS= 13

PRIMARY AREA ITERATIONS= 0

MACH NUMBER ITERATIONS= 3

**** CASE 7 ENTRAINMENT RATIO= 28.00000 ****

٠

CUTLET

MOMENTUM PATINED. 1176F 07

DIMENSIONLESS STATIC PRESSURE CHANGES-. 23976 00

DIMENSION ESS PEESSINE DISCEN. 2401E-01

TOTAL PRESSURE PISE=0.70026 02 POUND/SO. FOUT

THE PRIMARY STREAM AREA AFTER ACCUMODATEDN=0.3097F-03 SQ. FEET

	PQIMARY	<u> </u>	*	
PRESSIRE, PSTA	0.1059F 02	0.1058F 07	*	0.1110F 07
TEMPERATURE	0.2744F 03	0.4834F 03	•	0.401?F 03
VELOCITY, FINSEC	0.25075 04	0.7483F 03	•	0.7413F 03
MACH MUMBER	0.30805 0:	0.4029F 00	•	0.4871F 00
MASS FLOW PATE, POUND/MIN	0.62305 01	0.2305F 03	•	0.77676 03

ACCOMODATION REGION

PRESSURE ITERATIONS= 30

PRIMARY APPA ITEPATIONS= 1

MACH NUMBER ITERATIONS= 3

**** CASE 10 ENTRAINMENT PATIN= 27.00000 ****

*

MOMENTUM PATIO=0.9115F 01

DIMENSIONLESS STATIC PRESSURE CHANGE=-.1796E 00

DIMENSIONLESS PPESSURE RISE=0.3746E-01

TOTAL PRESSURE RISE=0.7875E 02 POUND/SQ. FOOT

THE PRIMARY STREAM AREA AFTER ACCOMODATION=0.3780F-03 SQ. FEFT

	PRIMARY	SECONDARY	*	
PRFSSURF, PSIA	0.11505 02	0.11505 02	*	0.11985 02
TEMPERATURE	0.2810F 03	0.4951= 03	*	0.5026F 03
VELOCITY, FT/SEC	0.74965 04	0.5478F 03	*	0.6474F 03
MACH NUMBER	0.3024F 01	0.5937E 00	٠	0880- 00
MASS FLOW RATE, POUND/MIN	0.6230F 01	0.2119F 03	*	0.21815 03

ΑΓΓΩΜΟΊΑΤΙΟΝ ΘΕGION

PRESSURE ITERATIONS= 21

PRIMARY APEA ITERATIONS = 0

MACH NUMBER ITERATIONS - 3

*

OUTLET

OUTLET

MACH NUMBER ITERATIONS= 4

PRIMARY AREA ITERATIONS = 1

PRESSURE ITERATIONS= 80

ACCOMODATION	REGION	*	OUTLET
PRIMARY	SECONDARY	*	

PRESSURE, PSIA	0.8932E 01	0.8932E 01	*	0.9727E 01	
TEMPERATURE	0.26145 03	0.4606F 03	*	0.4723E 03	
VELOCITY, FT/SEC	0.2533E 04	0.9137E 03	*	0-8776E 03	
MACH NUMBER	0.3194F 01	0.8681F 00	*	0.8234E 00	
MASS FLOW RATE, POUND/MIN	0.6230E 01	0.2492E 03	*	0.2554F 03	

THE PRIMARY STREAM AREA AFTER ACCOMODATION=0.4445F-03 SQ. FEET

TOTAL PRESSURE RISE=0.8365E 02 POUND/SQ. FOOT

DIMENSIONLESS PRESSURE RISE=0.3979E-01

DIMENSIONLESS STATIC PRESSURE CHANGE=-.3337E 00

MOMENTUM RATIO=0.1420E 02
TRAINMENT TIO	* •	DIMENSIONLESS TOTAL PRESSURE RISE	*	DIMENSIONLESS STATIC PRESSURE CHANGE	* •	MACH Number	• *	DIMENSIONLESS KINFTIC ENFRGY	* *	MOMENTU Ratio
10.0	٠	0.03988	٠	0.02251	٠	0.155	٠	0.0173	*	0.805
13.0	٠	0.03925	٠	0.01102	*	0.199	٠	.0.0280	٠	1.302
16.0	٠	0.03867	٠	-0.00337	٠	0.244	٠	0.0414	٠	1.927
19.0	٠	0.03819	٠	-0.02089	٠	0.291	*	0.0579	٠	2.688
22.0	*	0.03780	*	-0.04197	٠	0.340	*	0.0775	٠	3,593
25.0	*	0.03750	٠	-0.06717	٠	0.393	٠	0.1007	*	4.660
28.0	*	0.03733	٠	-0.09734	٠	0.450	*	0.1281	*	5.908
·1•0	*	0.03729	٠	-0.13397	٠	0.514	*	0.1604	٠	7.373
34.0	*	0.03746	٠	-0.17957	٠	0.589	٠	0.1997	٠	9.115
27.0	٠	0.03801	٠	-0.23973	٠	0.682	*	0.2476	+	11.259
40.0	*	0.03979	٠	-0.33374	٠	0.823	٠	0.316?	*	14.205

.

PRFSSURE PATID=30.00 TEMPERATURE RATID= 1.50 (ATH+CW/AM)=0.001000

SECONDARY PPFSSURE=14.60 PSI SECONDARY TEMPERATURE= 530.00 DEGREE RANKINE MIXING TURE AREA=0.087260 SQ. FT. * * * * * * * * *

ENTRAINMENT RATIO	* *	EXIT STATIC PRESSURE (PSI)	*	TOTAL PRESSURF RISF (PSF)	*	CHANGE (PSF)	*	(PSF)
10.0	*	14.93	*	87.84	*	47.3240	*	36.30
13.0	*	14.76	*	82.51	*	23.1685	*	59.76
16.0	*	14.55	*	R1.31	*	-7.0784	*	87.09
19.0	*	14.29	*	P0.30	*	-43.9244	*	121.64
22.0	*	13.99	*	79.46	*	-88.2290	*	162.93
25.0	*	13.62	*	78.84	*	-141.2129	*	211.76
28.0	*	13.18	*	78.48	*	-204.6518	*	269.21
31.0	*	12.64	*	78.39	*	-281.6621	*	337.16
34.0	*	11.98	*	78.75	*	-377.5278	*	418.71
37.0	*	11.10	*	79.0?	*	-504.0132	*	520.54
40.0	*	9.73	*	83.65	*	-701 .6533	*	664 . 84

REFERENCES

- 1. United States Letter Patent #3,093,349, Fred G. Wagner, Inventor: "Duct Arrangement for Aircraft Boundary Layer Control."
- 2. United States Letter Patent #3,085,740, Fred G. Wagner, Inventor: "End Inlet Jet Pump for Boundary Layer Control System."
- 3. Mc Clintock, F. A., and Hood, J. Hall: "Aircraft Ejector Performance," Journal of the Aeronautical Sciences, November, 1946.
- Keenan, J. H., Neumann, E. P., and Lustwerk, F.: "An Investigation of Jet Pump Design by Analysis and Experiment", Journal of Applied Mechanics, Volume 17 (1950).
- 5. Fabri, J. and Paulon, J.: "Theory and Experiments on Supersonic Air-to-Air Ejectors", NACA TM 1410, September, 1958.
- 6. Fabri, J. and Siestrunck, R.: "Supersonic Air Ejectors", in <u>Advances in</u> Applied Mechanics, Volume V., Academic Press, N.Y., 1958.
- De Leo, Richard V. and Wood, Richard D." "An Experimental Investigation of the Use of Hot Gas Ejectors for Boundary Layer Control", WADC TR 52-128, Part 1 (April, 1952), Part 2 (December, 1953), Part 3 (June, 1958).
- De Leo, Richard V., and Rose, R. E.: "An Experimental Investigation of the Use of Supersonic Driving Jets for Ejector Pumps", WADC TR 57-357, December, 1958.
- 9. De Leo, Richard V., Rose, R. E., and Dart, R. S.: "An Experimental Investigation of the Use of Supersonic Driving Jets for Ejector Pumps", Transactions of the ASME, J.E.P., 61-HTD-13.
- 10. Traksal, J.: "Simplified Ejector and Jet Thrust Augmentor Theory", Report No. 14898, Lockheed Aircraft Corporation, California Division, January, 1963.

- 11. Van der Lingen, T. W.: "A Jet Pump Design Theory", Transactions of the ASME, J.B.E., December, 1960.
- Seaver, R. J., "Development of a Hot Jet Pump BLC System (Jet Induced Lift System), Phase I: Bench Testing of An Experimental Jet Pump", final report, NASA Contract No. NAS 2-2518, prepared by Sunrise Aircraft Corporation of America, November, 1966.
- Society of Automotive Engineers, <u>Aero-Space Applied Thermodynamics Manual</u>, 1960.
- 14. Wirt, L., "New Data for the Design of Elbows in Duct Systems", G. E. Review, Volume 30, 1927, p. 286.
- 15. American Society of Heating, Refrigerating and Air Conditioning Engineers, Guide and Data Book, Systems and Equipment Volume, 1967.
- Lamb, O.P., and Holdhusen, J. S., "Investigation of Aircraft Ducting Components at High Subsonic Speeds," WADC TR 56-187, September, 1957.
- Dynatech Report No. 261, "Pressure Loss Calculation Procedures for High Speed Gas Flow in Ducts", prepared for U.S. Navy Bureau of Ships, Contract No. Nobs-84282, June, 1961.
- 18. Henry, J. R., "Design of Power Plant Installations; Pressure-Loss Characteristics of Duct Components," NACA WR L-208, 1944.
- Hall, W. B., and Orme, E. M., "Flow of a Compressible Fluid Through a Sudden Enlargement in a Pipe," Proceedings of the Institute of Mechanical Engineers, Volume 169, No. 49, 1955.
- Cole, B. N. and Mills, B., "Theory of Sudden Enlargements Applied to the Poppet Exhaust-Pulse Scavenging," Proceedings of the Institute of Mechanical Engineers, Vol. 1B, p. 364.

No.

- Patterson, G. N., "Modern Diffuser Design," Aircraft Engineering, September 1938.
- 22. Young, A. D., and Green, G. L., "Tests of High-Speed Flow in Diffusers of Rectangular Cross-Section," R & M No. 2201, British A.R.C., July, 1944.
- 23. Little, B.H., Jr., and Wilbur, S. W., "Performance and Boundary Layer Data from 12° and 23° Conical Diffusers of Area Ratio 2.0 at Mach Numbers up to Choking and Reynolds Numbers up to 7.5 x 10⁶, " NACA Report 1201, 1954.
- Young, A.D., Green, G.L., and Owen, P.R., "Tests of High-Speed Flow in Right-Angled Pipe Bends of Rectangular Cross-Section," R & M No. 2066, British A.R.C., October, 1943.
- 25. Higginbotham, J.T., Wood, C.C., and Valentine, E.F., "A Study of the High-Speed Performance Characteristics of 90° Bends in Circular Ducts," NACA TN 3696, June, 1956.
- 26. Friedman, D., and Westphal, W.R., "Experimental Investigation of a 90° Cascade Diffusing Bend with An Area Ratio of 1.45:1 and with Several Inlet Boundary Layers," NACA TN 2668, April, 1952.

Ā	Ŧ	Va	lues of E	P and Intrainn	Corresp nent Rai	onding tios, m	Maximu nax	ım	Fig. Nos.
. 001	1.5	P:	25	30	35	40	45		6,7
		m _{max} :	49	41	35	30.5	27		
	3.5	P:	40	45	50	55	60	65	8,9
		m _{max} :	47	41.5	37.5	34	31	29	
	8.0	P:	60	70	80	90			10,11
+		m _{max} :	47	40.5	35	31.5			
. 003	1.5	P :	9	10	11	12	13	14	12,13
		m _{max} :	45.5	41	37	34	31.5	29	
	3.5	P:	20	22	24	26	28		14,15
		m _{max} :	47	43	39	36	34	_	
	8.0	P :	20	22	24	26	28		16,17
+		m _{max} :	47	43	39	36	34		
.007	1.5	₽: ₽	4	4.5	5	5.5	6		18,19
		m _{max} :	44	39	35	32	29		
	3.5	P :	5	6	7	8	9		20,21
		m _{max} :	53.5	44.5	38	33.5	30.0		
	8.0	P :	9	10	11	12	13	14	22,23
+		m _{max} :	45	40.5	37	33.5	31	29	

 Table 1: Index to Analytical Jet Pump Performance Plots

Entrainment Ratio (m)	Jet Pump Total Pressure Rise psf (Δ P _t)	Mixing Tube. Outlet Static Pressure psia (P _m)	Mixing Tube Dynamic Head $psf \\ \left(\frac{\rho_m V_m^2}{2 g_0}\right)$	Mixing Tube Exit Mach No. ^M m
10.0	101.5	15.32	40.83	. 163
13.0	99.59	15.13	65.30	. 206
16.0	97.86	14.90	96.04	. 255
19.0	96.35	14.62	133.47	. 300
22.0	95.05	1 4.29	178.30	. 352
25.0	93.96	13.88	231.09	. 407
28.0	93. 12	13.40	293.75	. 469
31.0	92.61	12.81	368.10	. 535
34.0	92.60	12.06	458.39	. 615
37.0	93.52	11.03	573.92	. 722



Nozzle	Nozzle Nozzle Throat Throat Diameter Area inches ft ² x10 ⁴		Nozzle Type	Design Pressure psia	Design Temp. F	Flow Coefficient C _w	Flow Rate at Design W _p
Case 2	0.047	1.089	converging- diverging	350	200°F	. 938	6.39 lbm/min
Case 3	0.1109	6.05	converging- diverging	100	1200°F	.965	6.52 lbm/min
Case 4	0.0596	1.75	converging- diverging	350	1200°F	. 965.	6.60 lbm/min
Case 4A	0.063	1.952	converging	350	1200°F	. 935	7.11 lbm/min
Case LD-4	0.0658	1.650	converging- diverging	350	1200°F	.941	6.08lbm/min

 Table 3.
 Primary Nozzle Characteristics

.

	Flow Parameter	Instrumentation Used to Measure Parameter	How Recorded	Required for Determining	Data Reduction Procedure
Primary Flow	P _{po}	Bourdon Tube Gage	Manually and Photographically	Jet Pump Input Conditions	None needed
	T _{po}	Thermocouple and Bridge	Manually	Jet Pump Input Conditions	None needed
	w _p	Orifice Flow Meter and Panel Gage	Manually	Jet Pump Input Conditions	Standard calibration curves provided by flowmeter manu- facturer
Secondary Flow	T _{so}	Dial Gage In Suction Duct	Manually	Secondary Flow Temperature	None needed
	P _{atm}	Mercury Barometer	Manually	Atmospheric Pressure	None needed
	P _b	Manometers	Manually and Photographically	Secondary Flow Rate	See below
	w _s	Calibrated Bellmouth	Manually	Secondary Flow Rate in lb/min	Equation (91)
Mixing Tube and Diffuser	P vs. length	Manometer Board	Photographically	Mixing Tube and Diffuser Static Pressures	None needed
	P _{mo}) P _{do})	Kiel Probe Traverse	Manually	Discharge Stagnation Pres- sure for Jet Pump	"Mass-momentum" method for Compressible Flow (See Text)
	T _{eo}	Dial Gage Near End of Diffuser	Manually	Jet Pump Discharge Ten:perature	None needed

.

114	Test No.	Run No.	Nozzle Set	Primary Flow Pressure (psia)	Primary Flow Temperature (°F)	Secondary Flow Rate	Discharge Configuration	Cluster Configuration
	$\begin{pmatrix} 1\\2\\3\\\end{pmatrix}$	1-5 6-9 10-14	4	350 300 260	1085° 1160° 1185°	4 values 4 values 5 values	NAS2-2518 mixing tube	Original
	۲4 Test Seri	15-18 ies #2:	V Nozzle Cluster	225 r Performance Compari	1145 ison	4 values	v	·
	5 6 7 8 9	19-22 23-26 27-31 32-35 36-39 40-43	4	350 300 260 225 350 300	1130° 1160° 1165° 1158° 1160° 1180°	4 values 4 values	Constant-area short mixing tube	Original
	$ \begin{array}{c} 11\\ 12\\ 13\\ 14\\ 15\\ 16\\ \end{array} $	$\begin{array}{r} 44-47\\ 48-51\\ 52-55\\ 56-59\\ 60-63\\ 64-67\end{array}$	LD#2-4	260 225 350 300 260 225	1200° 1200° 1130° 1195° 1190° 1180°	4 values		LD#2
	Test Ser	ies #3:	Final Perform	ance Tests				
	$\begin{cases} 17\\18\\19\\20\\21\\22\\23\\24\\25\\26\\27\\\end{cases}$	$\begin{array}{r} 96-99\\ 100-103\\ 104-107\\ 108-111\\ 80-83\\ 84-87\\ 88-91\\ 92-95\\ 68-71\\ 72-75\\ 76-79\\ \end{array}$	4	350 300 260 225 300 260 225 175 260 225 175 175	$ \begin{array}{c} 1110^{\circ} \\ 1150^{\circ} \\ 1200^{\circ} \\ 1200^{\circ} \\ 750^{\circ} \\ 455^{\circ} \\ 460^{\circ} \\ 460^{\circ} \\ \end{array} $	4 values	Constant-area long mixing tube	LD#1

Test Series #1: Reference Test

Table 5. Jet Pump Test Program

Test No	Run No.	Nozzle Set	Primary Flow Pressure	Primary Flow Temperature	Secondary Flow Rate	Discharge Configuration	Cluster Configuration
			(psia)	(* £`)			
128	242-245	4A	350	1160°	4 values	Constant-area	LD#1
29	246 - 249	(converging)	300	1150°-1100°	1	short mixing tube	1
30	250-253	1	260	1150°			
31	254 - 257		225	1150°		↓ ↓	
32	140-143		350	1160°		Constant-area	
33	144 - 147	[300	1150°		long mixing tube	1
34	148-151	1	260	$1095^{\circ} - 1031^{\circ}$	1		
35	152 - 155		225	1020°		ļ	
36	124 - 127		300	750°			
37	128 - 131		260	1	1		
38	132 - 135		225	. ↓		}	
39	136-139		175	745°			
40	112 - 115		260	455°	[1	
41	116-119		225	455°		1	
\ 42	120-123	*	175	460°	*	*	*
f 43	180-183	2	350	200°	4 values	Constant-area	LD#1
44	184 - 187	1	300			long mixing tube	
45	188-191	Í	260	1		Ű, Ű	
46	192-195		225	. ↓			
47	196-199		400	455°	ļ		
48	200-203		350	455°		1	
49	204-207		260	450°			
50	208-211		225	450°		÷	
51	212-215	ļ	400	460°		Constant-area	
52	216-219		350	455°	[short mixing tube	
53	220-223		260	450°	1		
L 54	224-227	*	225	450°	*	*	Y
55	168-171	3	100	1160°-1115°	4 values	Constant area	LD#1
56	172 - 175	1	85	1140°	1	long mixing tube	1
57	176 - 179	1	70	1145°	1		1
58	156-159	1	85	755°			
59	160 - 163		70	750°		ļ	
60	164 - 167		55	745°		. ↓	
61	228 - 231	1	100	1150°-1110°		Constant-area	
62	232-235		85	1155°		short mixing tube	
63	236 - 239	*	70	1155°	*	-	*

-

.

Table 5. Jet Pump Test Program (Continued)

Test Date	Test Nos.	Run Nos.	Constant Area Mixing Tube	Nozzle Case No.	Tabulated Data	Static Pressure Parameter <u>Pm - Pso</u> Pso	$\frac{\text{Stagnation}}{\text{Pressure}}$ Rise Parameter $\frac{P_{\text{mo}} - P_{\text{so}}}{P_{\text{so}}}$	Velocity Profiles	Static Pressures vs. P and m po	Static Pressures vs. Distance	Thrust Augmentation Parameter T
25 July 1968	1-4	1-18	short	4 no extension	Table 8		Table 7 Figure 48	Figure 44 Figure 45	Figure 46	Figure 47	
31 July 1968	5-8	19-35	short	4 no extension	Table 9	Figure 48	Table 7 Figure 48	Figure 49	Figure 50	Figure 47	
13 August 1968	9-12	36-51	short	4	Table 10	Figure 52	Table 7	Figure 53	Figure 54	Figure 55	
19 August 1968	13-16	52-67	short	LD#2-4	Table 11	Figure 56	Table 7	Figure 57	Figure 58	Figure 55	
28 August 1968	17-27	68-111	extended	4	Table 12	Figure 59 Figure 60	Table 7 Figure 59	Figure 61 Figure 62	Figure 63	Figure 64	Table 12
13 Sept. 1968	28-31	242-259	short	4A	Table 13	Figure 65	Table 7	Figure 66	Figure 67		
3 Sept. 1968	32-42	112-155	extended	4A	Table 14	Figure 68 Figure 69			Figure 70	Figure 71	Table 14
11 Sept. 1968	43-50	180-211	extended	2	Table 15	Figure 72	Table 7	Figure 73	Figure 74	Figure 75	Table 15
11 Sept. 1968	51-54	212-227	short	2	Table 16	Figure 76	Table 7	Figure 73	Figure 77		
10 Sept. 1968	55-60	156-179	extend e d	3	Table 17	Figure 78	Table 7	Figure 79	Figure 80	Figure 81	Table 17
12 Sept. 1968	61-63	228-241	short	3	Table 18	Figure 82	Table 7	Figure 83	Figure 84		

Table 6

Index to Test Results

Test No.	Run No.	Traverse Location and Number of Traverses	Nozzles	Clus Co figur	n- mation	Duc Co figura	rt n- ition	P po DB1a	T _{po}	n	P tlo psf	P mo ps[P _{so} psf	$\frac{\overline{P}_{mo} \cdot P_{so}}{P_{so}}$	Analy Pmo- P	tical P <u>so</u> in Corrected	¹ Pico - ¹ Pino nici nol ex	(¹ ^{niv} ana) ^{- P} ^{mo} ex (¹ ^P mo) - P ^{so} anal	Comments on Results	
1	1	Diffuser 1	Case 4	Orig Clu	inal ster	Sho Conve	rt rging Tube	350	1094	26.2	2169.5	2187.9	2093.3	. 0 1519	, 0538	Friction 050:1	10,70	+10.2	short mixing tube one traverse incomplete mixing	
1	2	Diffuser 1	Case 4					350	1086	22.5	2171.2	2184.0	2100.3	. 03985	. 0540	. 0514	24.26	+22.5		
1	4	Diffuser	C880 4					350	1084	18.4	2180.6	2189,0	2107.0	. 03891	. 0555	. 0536	30.95	+27.4		
1	5	Diffuser	Case 4		╞			350	1085	15.6	2196.9	2203.0	2110.7	. 04372	. 0561	. 0547	23, 18	+20.07		
2	7	Diffuser	Case 4	<u> </u>	┢─			300	1158	25.4	2169.4	2180.8	2102.4	. 03729	. 9459	. 0435	13.06	+14.28	· · · · · · · · · · · · · · · · · · · ·	
3	11	Diffuser	Case 4	┼──	┞	$\left[- \right]$		260	1186	29.7	2157 8	2169.1	2102 2	03182	0791	0287		.12.20	·	
		1 Diffuser		Оті	† rinal	Short	Con-						2108.8	. 03182	. 0391	. 0387	10.26	+13.30	· · · · · · · · · · · · · · · · · · ·	
		1 Diffuser	CADO 4	Clu	ster	Mixin	Area <u>rTube</u> I	350	1131	22.4	2214.6	2225.8	2128.0	. 04595	. 0541	.0515	11.81	+10.78	···········	
	24	2	Case 4	-		<u> </u>		300	1174	24.9	2216.4	2225.8	2129.4	. 04527	. 0460	.0437	-3.34	- 3.59		
T	27	1	Ca10 4	-		(Dearer	6	360	1166	29.13	2192.5	2201.8	2128.1	. 03463	. 0393	. 0371	5.26	+ 6.66		
10	41	Diffuner 1	Case 4	LI	# 1	stant	Area Tube	300	1187	25.6	2192.1	2202.4	2117.2	. 0402	. 0458	. 0434	6.96	+ 7.60		
14	57	Diffuser 2	Case LD2-4	LI	₩ 2	Short stant Mizin	Con- Area (Tube	300	1198	27.4	2216.3	2226.5	2105.8	. 0573	. 0427	. 0403	-35.80	-42.18	Not as well mixed as other clusters. Traverse through secules significantly larger than traverse set through secules.	
18	101	Mixing Tube 2	Case 4	LI	₩1	Consi Ari Mixio	ded ani 6 Tube	300	1154	26.5		2204.1	2126.7	. 03639	. 0455	. 0376	2.57	+ 3.22	extended mixing tube two traversee rood mixing	
18	101	Diffuser 2	Case 4					30 0	1154	26, 5	2190. 4	2202.1	2126.7	. 0355	. 0455	. 0376	4.47	+ 5. 59		
18	162	Mixing Tube	Case 4					300	1158	22.1		2206.9	2131.25	. 0355	. 0465	. 0409	11.51	+13.20		
18	102	Diffuser 2	Case 4					300	1158	22.1	2204.9	2212.9	2131.3	. 03828	. 0465	. 0409	5.58	÷ 6.40		
56	173	Mixing Tube 2	Case 3					85	1138	23.6		2188.7	2127.8	. 02862	. 0360	. 0298	2.51	+ 3.96		
56	173	Diffuser 2	Case 3					85	1138	26,6	2178.1	2187.8	2127.8	. 02782	. 0360	. 0298	4.21	+ 6.64		
48	200	Diffuser 1	Case 2					350	+44	28.5	2124.0	2136.1	2088.6	. 02274	. 0350	. 0269	8, 69	415,46	extended mixing tube one traverse	
52	217	Diffuser 1	Case 2			Short stant Mixing	Con- Area Lube	350	452	27.3	2156.1	2166.6	2089.2	. 03704	, 0349	0:12.5	9,48	-13.97	short mixing tube one traverse distorted velocity profile	
63	240	Diffuser 1	Case 3					85	1159	26	2142.5	2153,7	2088.7	. 03111	. 0354	0328	3, 53	+ 5 .15	A CONTRACTOR OF	
29	258	Diffuser 1	Case 4/			\square		300	1103	26.6	2143.5	2157.4	2093. 9	. 03032	. 0490	. 0159	32-62	+33 94	short muxing tube one traverse distorted velocity profile	

Table 7. Tabulation of Stagnation Pressure Results

.

Test No.	Run	P po psia	T po °F	W _p lbm/min	T so °F	W _s lbm/min	m
1	1	350	1094	6.69	95	175.5	26.2
	2		1086	6.70	93	150.8	22.5
	4		1084	6.66	95	122.8	18.4
ł	5	+	1085	6,70	95	104.4	15.6
2	6	300	1164	5.66	93	163.4	28.9
	7		1158	5.66	94	143.5	25.35
	8		1158	5.65	95	124.0	21.95
+	9	ŧ	1160	5.67	95	99.0	17.45
3	10	260	1185	4.91	91	153.5	31.3
	11	ļ	1186	4.87	93	144.8	29.7
	12		1185	4.87	93	128.1	26.3
	13		1184	4.89	92	108.3	22.15
+	14	ŧ	1185	4.90	92	92.0	18.8
4		225	1139	4.39	93	143.2	32.6
	16		1158	4.20	92	128.0	30.5
	17	1	1158	4.34	93	112.6	25.9
+	18	ŧ	1145	4.40	93	94.1	21.4

Table 8

Jet Pump Test Results NAS 2-2518 Mixing Tube Original Nozzle Cluster Case 4 Nozzles .

I

Test No.	Run	P po psia	T po °F	W _p lbm/min	T so ° F	W _s lbm/min	m
5	19	350	1162	6 57	89	162 5	24 74
1	20	l	1131	6 75	88	151 4	22.12
	20		1194	6 66	96	195 /	10 00
	41	ļ	1124	0.00	00	120.4	10.00
V	44	v	1130	0.03	82	104.6	19.18
6	23	300	1158	5.70	90	151.4	26.57
	24	1	1162	5.70	88	145.7	25.56
	25		1162	5.70	88	124.4	21.83
ł	26	Ļ	1160	5.70	88	99.9	17.52
7	27	260	1166	4.84	89	141.0	29.13
	29	I	1165	4.84	89	129.2	26.68
	30		1165	4.84	89	110.2	22.76
	31	Ļ	1166	4.84	89	91.8	18.97
0		2.95	1156	4 99		129 6	20 62
8	32	225	1156	4.33	90	132.6	30.63
	33		1158	4.29	89	128.1	29.85
	34		1158	4.29	89	111.5	25.98
+	35	*	1158	4.29	88	92.4	21.54

Table 9

Jet Pump Test Results Short Constant Area Mixing Tube Original Nozzle Cluster Case 4 Nozzles

Test No.	Run	P po psia	T °Po °F	W _p lbm/min	T so °F	W _s lbm/min	m
9	36	350	1164	6.82	88	182.0	26.7
	37		1158	6.82	87	149.5	21.95
	38		1158	6.82	86	125.0	18.35
ļļ	39	ţ	1158	6.82	87	107.4	15.80
10	40	300	1209	5.72	85	170.0	29.8
	41		1187	5.75	86	146.9	25.6
	42		1160	5.78	85	125.0	21.6
+	43	ł	1161	5.78	85	98.5	17.0
11	46	260	1207	4.95	86	160.0	32.3
	47		1201	4.98	86	139.7	28.1
	44		1200	4.98	86	112.5	22.6
l t	45	ł	1202	4.98	86	93.6	18.8
12	48	225	1188	4.32	86	151.0	35.0
	49		1196	4.32	86	129.0	29,9
	50		1201	4.27	85	113.1	26,5
ļ t	51	ŧ	1203	4.32	85	95.5	22.1

Ô

Table 10

Jet Pump Test Results Short Constant Area Mixing Tube LD#1 Nozzle Cluster Case 4 Nozzles

Test No.	Run	Ppo	Tpo	Wp	T _{so}	Ws	m
		psia	<u> </u>	lbm/min	<u> </u>	<u>lbm/min</u>	
13	52	350	1137	6.18	82	177.4	28.7
	53		1128	6.21	80	153.1	24.7
	54		1125	6.21	82	125.1	20.2
↓	55	ł	1122	6.20	83	105.0	16.9
14	56	300	1195	5.24	85	164.3	31.4
	57		1160	5.34	83	146.1	27.4
	58		1194	5.21	82	125.1	24.0
•	59	¥	1194	5.23	82	100.4	19.2
15	60	260	1192	4.63	84	154.7	33.4
	61		1191	4.61	84	129.7	28.1
	62		1190	4.58	84	109.8	24.0
•	63	ł	1190	4.59	84	93.3	20.3
16	64	225	1184	3.99	84	144.3	36.2
	65		1183	3.99	84	129.4	32.4
	66		1182	3.96	82	115.1	29.1
	67	¥	1182	3.96	84	95.3	24.1

Table 11

Jet Pump Test Results Short Constant Area Mixing Tube LD#2 Nozzle Cluster LD#2-4 Nozzles (7 Nozzles)

Test No.	Run	P po	T _{po}	w _p	T _{so}	ws	m	au
		psia	<u>°</u> F	lbm/min	<u>°F</u>	lbm/min		<u> </u>
25	68	260	453	6.48	7,8	163.4	25.2	5.06
1	69		452	6.48		140.4	21.7	3.64
	70		453	6.46	1	122.9	19.0	2.77
*	71	*	455	6.48	Ţ	110.5	17.1	2.24
26	72	225	454	5.67	78	153.2	27.0	5.05
}	73		456	5.67	79	139.3	24.6	4.11
l	74		460	5.65	79	124.7	22.1	3.28
Y	75	Y	462	5.64	77	104.2	18.5	2.28
27	76	175	459	4.43	78	133.8	30.2	4.92
	77		459	4.43	79	121.6	27.5	4.04
	78		463	4.43	79	106.8	24.1	3.09
V	79	¥	463	4.42	79	94.3	21.3	2.42
21	80	300	753	6.63	82	172.3	26.0	4.96
	81		753	6.60	1	153.2	23.2	3.81
	82	L	752	6.59	1	130.0	19.7	2.71
	83	Y	752	6.59	•	111.4	16.9	1.99
22	84	260	749	5.66	81	161.8	28.6	5.07
	85	1	745	5.65	80	142.2	25.2	3.81
	86	1	749	5.64	80	123.5	21.9	2.86
+	87	Ţ	752	5.62	80	107.6	19.2	2.19
23	88	225	753	5.03	82	150.4	29,9	4.89
1	89		750	5.02	80	134.3	26,7	3.83
	90	¥	749	5.01	81	117.8	23,5	2.94
*	91	,	748	5.02	81	103.9	20.7	2.28
24	92	175	744	4.0	82	132.2	33.1	4.77
	93		745	3.96	80	119.9	30,3	3.92
	94	L L	745	3.94	83	107.3	27.2	3.15
*	95	Y	748	3.94	80	95.7	24.3	2.50
17	96	350	1116	6.82	88	179.8	26.4	4.85
1	97		1115	6.81	88	153.0	22.5	3.35
	98	4	1112	6.76	88	129.7	19.2	2.39
Ť	99	T I	1105	6.77	89	117.3	17.3	1.96
18	100	300	1147	5.78	88	167.3	28.9	4.76
	101		1149	5.78	87	153.1	26.5	3.91
•	1 02	*	1154	5.77	86	127.7	22.1	2.68
1	103	•	1159	5.73	86	112.7	19.7	2.08
19	104	260	1195	5.01	83	156.7	31.3	4.67
	105	1	1198	4.98	83	137.4	27.6	3.54
Ļ	106		1199	4.98	83	115.1	23.1	2.47
Y	107	T	1203	4.96	80	106.7	21.5	2.13
20	108	225	1198	4.35	83	145.5	33.4	4.61
	109		1203	4.35	82	134.7	31.0	3.92
4	110	Ţ	1205	4.26	81	119.0	27.9	3.09
Ŧ	111	Ţ	1204	4.30	81	101.0	23.5	2.22

.

Table 12

Jet Pump Test Results Extended Constant Area Mixing Tube LD#1 Nozzle Cluster Case 4 Nozzles

•

Test No.	Run	P po psia	T po °F	Wp lbm/min	T so °F	W _s lbm/min	m
28	242	350	1160	7.25	86	175.3	24.17
	243		1139	7.24	86	156.4	21.60
	244		1156	7.25	86	134.3	18.53
+	245	+	1156	7.24	85	105.5	14.57
29	246	300	1165	6.31	85	166.0	26.30
	247		1154	6.24	85	148.0	23.71
	248		1143	6.23	85	127.3	20.43
	249		1126	6.23	84	101.0	16.21
	258		1103	6.26	87	166.8	26.64
•	259	+	1103	6.30	88	167.7	26.62
30	250	260	1149	5.44	85	157.7	28.99
	251		1149	5.41	85	142.5	26.35
	252		1149	5.38	85	122.0	22.68
•	253	ŧ	1150	5.38	84	100.5	18.68
31	254	225	1148	4.70	84	150.0	31.91
	255		1150	4.69	84	136.6	29 . 13
	256		1150	4.66	84	119.0	25.54
ļ	257	ł	1150	4.64	83	95.3	20.53

> Table 13 Jet Pump Test Results Short Constant Area Mixing Tube LD#1 Nozzle Cluster Case 4A Nozzles

Test	Run	Ppo	T _{po}	w _p	T _{so}	Ws	m	τ
NO.		psia	°F	lbm/min	°F	lbm/min		
40	112	260	453	6.96	78	165.6	23.8	4.93
1	113	1	457	6.92	78	142.8	20.6	3.57
	114		459	6.91	79	125.2	18.1	2.72
t i	115	+	462	6.89	79	108.3	15.7	2.05
41	116	225	456	6.05	79	155.4	25.7	4.96
1	117	1	455	6.02	80	138.5	23.0	3.89
	118		456	5.98	80	124.3	20.8	3.12
. ↓	119	1	457	5.98	80	103.2	17.3	2.16
42	120	175	458	4.81	80	140.2	29.1	5.06
	121		462	4.81	1	123.5	25.7	3.89
	122		463	4.78		107.3	22.5	2.94
ł	123	1	464	4.78	+	96.5	20.2	2.38
36	124	300	753	7.10	84	172.1	24.2	4.73
1	125		750	7.10		150.4	21.2	3.50
1	126		749	7.08	1	128.7	18.2	2.54
ŧ	127	ļ	750	7.07	Y	111.7	15.8	1.92
37	128	260	751	6.20	85	161.9	26.1	4.75
	129		752	6.19		140.5	22.7	3.49
1	130		754	6.17		123.7	20.1	2.70
ł	131	ļ	756	6.17	ŧ	106.8	17.3	2.01
38	132	225	745	5.35	86	152.6	28.5	4.86
	133	1	743	5.35	86	134.2	25.1	3.69
	134		744	5.36	85	117.8	22.0	2.82
. ↓	135	↓ ↓	746	5.35	85	103.2	19.3	2.17
39	136	175	739	4.26	86	137.0	32.2	4.94
1	137		743	4.24		119.7	28.2	3.73
	138		746	4.23	1	105.4	24.9	2.89
ŧ	139	ŧ	748	4.22	*	95.9	22.7	2.40
32	140	350	1175	7.26	92	174.6	24.1	4.34
1	141		1162	7.18	1	151.1	21.1	3.17
	142		1158	7.17		124.5	17.4	2.13
↓ ↓	143	. ↓	1140	7.18	ŧ	115.5	16.1	1.83
33	144	300	1146	6.26	94	165.1	26.4	4.43
1	145	1	1149	6.23	1	145.2	23.3	3.35
	146		1148	6.28		124.6	19.8	2.43
↓ I	147	ŧ	1124	6.27	ŧ	110.7	17.7	1.93
34	148	260	1094	5.52	94	156.8	28.4	4.60
Ŧ	149	1	1057	5,55	1	129.7	23.4	3.05
	150	I	1048	5,56	1	112.8	20.3	2.30
Ļ	151	. ↓	1031	5.58	+	105.1	18.8	1.99
35	152	225	1021	4.95	94	148.7	30.0	4.62
1	153		1021	4.94		129.7	26.3	3.47
	154		1021	4.94	L I	114.5	23.2	2.70
+	155	+	1021	4.93	v	101.9	20.7	2.14

<u>ې</u> 🚯

Table 14

Jet Pump Test Results Extended Constant Area Mixing Tube LD#1 Nozzle Cluster Case 4A Nozzles

Test No.	Run	P po psia	T po °F	W _p lbm/min	T so °F	W _s lbm/min	m	τ
<u> </u>								
43	180	350	201	6.35	78	158.7	25.0	5.55
	181	1	199	6.38		140.2	22.0	4.21
	182		199	6.39		123.3	19.3	3.21
*	183	Y	200	6.39	ŧ	106.4	16.7	2.38
44	184	300	200	5.51	79	148.8	27.0	5.53
	185		205	5.47	1	137.0	25.0	4.68
	186		204	5.45		123.6	22.7	3.79
+	187	ŧ	205	5.44	ŧ	101.3	18.6	2.53
45	188	260	201	4.80	79	138.4	28.8	5.52
l	189	1	203	4,79	1	126.3	26.4	4.56
	190		201	4.78		114.1	23.9	3.70
ŧ	191	+	201	4.78	+	96.4	20.2	2.63
46	192	225	197	4.14	79	128.4	31.0	5.52
l l	193	1	197	4.14	1	118.9	28.7	4.70
	194		196	4.14		108.1	26.1	3.86
+	195	ŧ	199	4.12	ŧ	93.9	22.8	2.92
47	196	400	456	6.30	81	167.7	26.6	5.43
Î	197	1	453	6.30	1	147.3	23.4	4.06
	198		455	6, 29	1	129.3	20.6	3.09
+	199	ŧ	454	6.29	+	110.5	17.6	2.24
48	200	350	444	5, 56	81	158.3	28.5	5.43
Ĩ	201	Ĩ	454	5.54	1	140.8	25.4	4.21
	202		454	5.54		126.4	22.8	3.36
+	203	ŧ	454	5,52	ŧ	104.8	19.0	2.31
49	204	260	449	4,07	82	137.0	33.7	5.50
Ī	205	l l	450	4.08	1	126.7	31.0	4.63
	206		451	4.08	1	115.4	28.3	3.82
ŧ	207	ŧ	452	4.08	+	97.2	23.8	2.70
50	208	225	447	3, 58	82	127.8	35,7	5.45
Ĩ	209		447	3,58	1	116.2	32.5	4.47
	210		449	3, 59		106.5	29.7	3.73
Ļ	211	1	450	3, 58	↓ I	93.2	26.0	2.85
Ŧ		T	100	v. vo	4			

_ - - - -

Table 15

Jet Pump Test Results Extended Constant Area Mixing Tube LD#1 Nozzle Cluster Case 2 Nozzles

Test No.	Run	P po psia	T °F	₩ _p lbm/min	T SO °F	₩ s lbm/min	m
51	212	400	456	6.25	85	171.4	27.4
	213		459	6.26	85	149.6	23.9
	214		458	6.25	85	127.4	20.4
+	215	+	458	6.24	86	23.7	15.8
52	216	350	451	5.52	86	161.3	29.2
	217		452	5.52	85	150.8	27.3
	218		456	5.50	84	128.3	23.3
+	219	+	457	5.49	83	96.6	17.6
53	220	260	451	4. 12	81	142.5	34.6
1	221		451	4.12	81	128.6	31.2
	222		451	4.09	82	109.5	26. 8
+	223	+	451	4.09	82	92.1	22.5
54	224	225	449	3.57	83	133.4	37.4
	225		448	3.58	83	120.9	33.8
	226		44 8	3.58	83	105.9	29.6
Ŧ	227	ŧ	448	3.58	83	88.9	24.8

Table 16

Jet Pump Test Results Short Constant Area Mixing Tube LD#1 Nozzle Cluster Case 2 Nozzles

Test No.	Run	P po psia	T po °F	W _p lbm/min	T SO °F	W _s lbm/min	m	τ
58	156	85	755	6.64	540	158.3	23.8	4.95
	157		758	6.59	541	138.6	21.0	3.75
	158		758	6.59	542	124.3	18.9	3.02
ŧ	159	ţ	758	6.58	543	105.5	16.0	2.18
5 9	160	70	751	5.44	542	141.9	26.1	4.97
	161		748	5.42	542	128.9	23.8	4.07
	162		748	5.43	542	113.4	20.9	3.13
ŧ	163	ł	749	5.41	542	98.0	18.1	2.35
60	164	55	743	4.40	542	121.6	27.6	4.68
	165		743	4.40	543	114.2	26.0	4.16
Ì	166		744	4.40	543	105.0	23.9	3.52
ŧ	167	ŧ	746	4.40	543	90.2	20.5	2.59
55	168	100	1158	6.61	550	161.2	24.4	4.62
ĺ	169		1140	6.66	550	146.1	21.9	3.68
	170		1128	6.68	550	128.3	19.2	2.81
ŧ	171	ŧ	1116	6.67	550	107.9	16.2	2.00
56	172	85	1141	5.74	549	149.4	26.0	4.58
	173		1138	5.75	550	134.5	23.4	3.67
	174		1138	5.76	550	120.7	21.0	2.94
ł	175	ŧ	1138	5.76	550	104.3	18.1	2.20
5 7	176	70	1141	4.81	549	133.4	27.7	4.41
	177		1146	4.80	548	120.5	25.1	3.58
	178		1148	4.76	548	110.7	23.3	3.06
. ↓	179	. ↓	1154	4.77	549	96.4	20.2	2.33

Table 17

Jet Pump Test Results Extended Constant Area Mixing Tube LD#1 Nozzle Cluster Case 3 Nozzles

Test No.	Run	P po psia	T po °F	W _p lbm/min	T _{so} °F	W _s lbm/min	m
61	228	100	1151	6.80	88	167.6	24.65
	229		1130	6.80	87	148.0	21.76
	230		1124	6.79	87	124.6	18.34
ł	231	ŧ	1110	6.77	87	99.1	14.63
62	232	85	1161	5.82	86	156.3	26.86
	233		1159	5.81	81	143.7	24.74
	234		1154	5.80	83	121.6	20.96
	235		1153	5.80	85	98.3	16.95
	240	1	1159	5.73	92	150.4	26.25
Y	241	v	1149	5.76	93	152.1	26.40
63	236	70	1158	4.81	86	141.8	29.47
	237		1155	4.81	86	126.2	26.23
	238		1152	4.81	86	112.5	23.38
ŧ	239	ł	1152	4,81	86	92.8	19.29

Table 18

Jet Pump Test Results Short Constant Area Mixing Tube LD#1 Nozzle Cluster

Case 3 Nozzles

L

-



BLC Jet Pump Duct Arrangement



Jet Pump Definitions



Figure 3 Mixing Tube Wall Friction Approximation





Block Diagram of the Computer Program

.

132

. . .





Constant C_p C_p as a function of temperature

Solution Conditions

$$T_{po} = 1200^{\circ} F \qquad T_{so} = 70^{\circ} F$$

$$P_{po} = 350 \text{ psia} \qquad P_{so} = 14.6 \text{ psia}$$

$$A_{th} = 0.000175 \text{ ft}^2 \qquad C_w = 0.935$$

$$A_m = 0.08726 \text{ ft}^2$$



PUMP CHARACTERISTICS. PARAMETER IS PRESSURE RATIO



I

Figure 7

Jet Pump Performance Characteristics Momentum Augmentation (τ) vs. Entrainment Ratio

$$\begin{array}{rcl} \underline{A} &=& 0\,.\,001\\ \\ \underline{T} &=& 1.5\\ \hline{P} &=& 25,\ 30,\ 35,\ 40,\ 45 \end{array}$$

135

- -



PUMP CHARACTERISTICS, PARAMETER IS PRESSURE RATIO

Q.

Figure 8

Jet Pump Performance Characteristics Dimensionless Pressure Rise (ΔP_t^*) vs. Entrainment Ratio

$$\overline{\frac{A}{T}} = 0.001$$

$$\overline{\frac{T}{T}} = 3.5$$

$$\overline{P} = 40, 45, 50, 55, 60, 65$$



MOMENTUM RATIO, PARAMETER IS PRESSURE RATIO

Figure 9

Jet Pump Performance Characteristics Momentum Augmentation (τ) vs. Entrainment Ratio

 $\overline{\underline{A}} = 0.001$ $\overline{\underline{T}} = 3.5$ $\overline{\underline{P}} = 40, 45, 50, 55, 60, 65$



Entrainment Ratio



Jet Pump Performance Characteristics

Dimensionless Pressure Rise ($\triangle P_t^*$) vs. Entrainment Ratio

$$\overline{A} = 0.001$$

 $\overline{T} = 8.0$
 $\overline{P} = 60, 70, 80, 90$

Momentum Ratio - Exit/Primary



Entrainment Ratio

Figure 11

Jet Pump Performance Characteristics

Momentum Augmentation (τ) vs. Entrainment Ratio

$$\overline{A} = 0.001$$

 $\overline{T} = 8.0$
 $\overline{P} = 60, 70, 80, 90$



PUMP CHARACTERISTICS. PARAMETER IS PRESSURE RATIO


l

 $\overline{\underline{A}} = 0.003$ $\overline{\underline{T}} = 1.5$ $\overline{P} = 9, 10, 11, 12, 13, 14$



PUMP CHARACTERISTICS. PARAMETER IS PRESSURE RATIO

Jet Pump Performance Characteristics Dimensionless Pressure Rise (ΔP_t^*) vs. Entrainment Ratio

 $\overline{\overline{\underline{A}}} = 0.003$ $\overline{\underline{T}} = 3.5$ $\overline{\underline{P}} = 13, 15, 17, 19, 21$



MOMENTUM RATIO. PARAMETER IS PRESSURE RATIO



Entrainment Ratio

Figure 16

Jet Pump Performance Characteristics

Dimensionless Pressure Rise (ΔP_t^*) vs. Entrainment Ratio

$$\overline{A} = 0.003$$

 $\overline{T} = 8.0$
 $\overline{P} = 20, 22, 24, 26, 28, 30$

Momentum Ratio - Exit/Primary



Entrainment Ratio

Figure 17

Jet Pump Performance Characteristics

Momentum Augmentation (τ) vs. Entrainment Ratio

$$\overline{A} = 0.003$$

 $\overline{T} = 8.0$
 $\overline{P} = 20, 22, 24, 26, 28, 30$



PUMP CHARACTERISTICS. PARAMETER IS PRESSURE RATIO

Jet Pump Performance Characteristics Dimensionless Pressure Rise (ΔP_t^*) vs. Entrainment Ratio

 $\overline{\overline{\mathbf{A}}} = 0.007$ $\overline{\overline{\mathbf{T}}} = 1.5$ $\overline{\mathbf{P}} = 4, 4.5, 5, 5.5, 6$



MOMENTUM RATIO, PARAMETER IS PRESSURE RATIO

Jet Pump Performance Characteristics Momentum Augmentation (τ) vs. Entrainment Ratio

 $\overline{\underline{A}} = 0.007$ $\underline{T} = 1.5$ $\overline{P} = 4, 4.5, 5, 5.5, 6$

147

-- -



PUMP CHARACTERISTICS. PARAMETER IS PRESSURE RATIO

Figure 20

Jet Pump Performance Characteristics Dimensionless Pressure Rise (ΔP_t^*) vs. Entrainment Ratio

$$\overline{\overline{\underline{A}}} = 0.007 \overline{\underline{T}} = 3.5 \overline{\underline{P}} = 5, 6, 7, 8, 9$$



MOMENTUM RATIO. PARAMETER IS PRESSURE RATIO



PUMP CHARACTERISTICS. PARAMETER IS PRESSURE RATIO

Jet Pump Performance Characteristics Dimensionless Pressure Rise (ΔP_t^*) vs. Entrainment Ratio

$$\overline{\overline{A}} = 0.007$$

 $\overline{\overline{T}} = 8.0$
 $\overline{\overline{P}} = 9, 10, 11, 12, 13, 14$



MOMENTUM RATIO. PARAMETER IS PRESSURE RATIO

Jet Pump Performance Characteristics Momentum Augmentation (τ) vs. Entrainment Ratio

l

 $\overline{\underline{A}} = 0.007$ $\overline{\underline{T}} = 8.0$ $\overline{P} = 9, 10, 11, 12, 13, 14$



Figure 24

Jet Pump Pressure Rise in Relation to Geometry (\overline{A}_1) and Mixing Tube Exit Mach Number



MOMENTUM RATIO, PARAMETER IS AREA RATIO

- A.

Figure 25 Jet Pump Momentum Ratio in Relation to Geometry (\bar{A}_1) and Mixing Tube Exit Mach Number



Figure 26 Jet Pump System Operating Point



Influence of Pressure Ratio on Jet Pump-System Operating Points



Figure 28 Loss Coefficients for Straight Conical Diffusers

-



Figure 29

Chart for Determining M_{b} (Equation 66)

 $\Delta P_t(psf)$ q_m(psfa) ^Mm ΔP_{i} 1.00-. 90 . 80. $P_m^{(psfa)}$.70 .60 . 50 Μ .40 . 30 .20 .10

Entrainment Ratio, m



Jet Pump Operating Characteristics



Figure 31

Determination of System Operating Points





Influence of Blowing Duct Diffusion Upon Entrainment Ratio



Figure 33 Jet Pump Thrust Augmentation Characteristic



П

I II



Influence of Blowing Duct Diffusion Upon System Thrust Augmentation



Figure 35

Relationship of Thrust Augmentation to Entrainment Ratio









Figure 37 Original Mixing Tube Geometry



Low Drag Nozzle Cluster #1



â 🖺

ł



Low-Drag Nozzle Cluster #2



:

.

Figure 40 Dynatech Jet Pump Test Instrumentation



(Measurements Made at End of Short Mixing Tube, L/d_m = 1.35)



Figure 42 NAS 2-2518 BLC System Suction Duct



Stagnation Pressure Loss

 Static Pressure Change Corresponding to the Stagnation Pressure Loss

Figure 43

Pressure Loss Characteristics of the Extended Constant Diameter Mixing Tube (Pressure Measurements Taken 2.72 Duct Diameters Apart; $d_m = 4.0$ Inches)



Figure 44

Velocity Profiles at Diffuser Discharge

$$d = 5.084'' \qquad \frac{A_d}{A_m} = 1.71$$

NAS 2-2518 Mixing Tube Original Nozzle Cluster Case 4 Nozzles (350 psia, 1090°F)

$$W_p = 6.69$$
 lbm/min



Velocity Profiles at Diffuser Discharge

$$d = 5.084'' \qquad \frac{A_d}{A_m} = 1.71$$

NAS 2-2518 Mixing Tube Original Nozzle Cluster Case 4 Nozzles





Figure 47 Variation of Static Pressure in Mixing Tube and Diffuser For Three Configurations Case 4 Nozzles $P_{po} = 350 \text{ psia}$ $T_{po} = 1100^{\circ} \text{F to } 1200^{\circ} \text{F}$ $W_{m} = 124.9 \text{ lbm/min}$

Stagnation Pressure Rise Parameter +0.08 KEY Р ро Analytical Results 350 (psia +0.06 Experimental Results 300 260 +0.04 Constant Area Short Mixing Tube 225 Data (psia) P_{po} I +0.02 $P_{mo} - P_{so}$ 350 \odot P_{mo} Calculated SO 0 · 300 from one Velocity Ppo <u>∕</u> ⊽ 260 Traverse 225 50 (psia Static Pressure Parameter, $\mathbf{\overline{0}}$ \mathbf{P}_{mo} Calculated 300 +0.02 from two Perpendicular Vel-0 ocity Traverses -0.02 NAS 2-2518 Mixing Tube -0.04 ⊗ P_{po} = 350 psia ı - P_{so} -0.06 SO -0.08 Ъ -0.1 0 10 20 30

. . . .

m - Entrainment Ratio

Figure 48 Jet Pump Pressure Rise Parameters Calculated at Mixing Tube Discharge Original Nozzle Cluster Case 4 Nozzles
V - Axial Velocity fps



Velocity Profiles at Diffuser Discharge

d = 5.084"
$$\frac{A_d}{A_m}$$
 = 1.61

Short Constant Area Mixing Tube Original Nozzle Cluster - Case 4 Nozzles







120

100 3.0

Original Cluster A with Case 4 Nozzles and 2" Extensions (LD#1) Low Drag Nozzle ×

Original Cluster

with Case 4 Nozzles

D

Cluster with Case LD#2-4 Nozzles (LD#2)

W_p - Primary Flow Rate lbm min

6.0

7.0

Figure 51

5.0

4.0

Secondary Flow Vs. Primary Flow Wide Open Throttle Condition Short Constant Area Mixing Tube



m - Entrainment Ratio



Jet Pump Static Pressure Parameter Calculated at Mixing Tube Discharge

> Short Constant Area Mixing Tube Low-Drag Nozzle Cluster #1 Case 4 Nozzles



se 4 nozzies





Variation of Static Pressure in Mixing Tube and Diffuser for Three Nozzle Clusters

> $P_{po} = 350 \text{ psia}$ $T_{po} = 1100^{\circ} \text{F to } 1200^{\circ} \text{F}$ $W_{m} = 124.9 \text{ lbm/min}$



m - Entrainment Ratio

Jet Pump Static Pressure Parameters Calculated at Mixing Tube Discharge Short Constant Area Mixing Tube

> Low-Drag Nozzle Cluster #2 LD #2-4 Nozzles (7 Nozzles)



LD#2-4 Nozzles



LD #2 Nozzle Cluster LD #2-4 Nozzles



Jet Pump Pressure Parameters Extended Constant Area Mixing Tube LD#1 Nozzle Cluster

Case 4 Nozzles



Figure 60







I



Velocity Profiles Extended Constant Area Mixing Tube LD #1 Nozzle Cluster - Case 4 Nozzles







Variation of Static Pressure in Mixing Tube and Diffuser Extended Constant Area Mixing Tube LD #1 Nozzle Cluster - Case 4 Nozzles



m - Entrainment Ratio

Jet Pump Static Pressure Parameter Calculated At the Mixing Tube Discharge Short Constant Area Mixing Tube LD #1 Nozzle Cluster - Case 4A Nozzles



Velocity Profiles at Diffuser Discharge

$$d = 5.087"$$

 $A_d / A_m = 1.61$

Short Constant Area Mixing Tube LD #1 Nozzle Cluster - Case 4A Nozzles



·





Jet Pump Static Pressure Parameter

Extended Constant Area Mixing Tube LD #1 Nozzle Cluster

Case 4A Nozzles





Static Pressures in Extended Constant Area Mixing Tube LD #1 Nozzle Cluster - Case 4A Nozzles





Case 4A Nozzles





Manaurud Ctatia Dressure - Indres of Water Core





P_{po}

m - Entrainment Ratio



Static Pressures in Extended Constant Area Mixing Tube LD #1 Nozzle Cluster Case 2 Nozzles

Measured Static Pressure - Inches of Water Gage



Variation of Static Pressure in Mixing Tube and Diffuser Extended Constant Area Mixing Tube LD #1 Nozzle Cluster - Case 2 Nozzles





m - Entrainment Ratio

Jet Pump Static Pressure Parameter Calculated at the Mixing Tube Discharge Short Constant Area Mixing Tube

> LD #1 Nozzle Cluster Case 2 Nozzles

204

1

L

į

Static Pressure 0.81 Inches Downstream P - of Primary Nozzle Discharge-Inches of Water Gage (tap #18)



Д

ł





m - Entrainment Ratio

Figure 78

Jet Pump Static Pressure Parameter Extended Constant Area Mixing Tube LD #1 Nozzle Cluster Case 3 Nozzles





m - Entrainment Ratio

Figure 80

Static Pressures in Extended Constant Area Mixing Tube LD #1 Nozzle Cluster

Case 3 Nozzles





Variation of Static Pressure in Mixing Tube and Diffuser Extended Constant Area Mixing Tube

LD #1 Nozzle Cluster Case 3 Nozzles



m - Entrainment Ratio



Short Constant Area Mixing Tube LD #1 Nozzle Cluster Case 3 Nozzles



Velocity Profiles at Diffuser Discharge Short Constant Area Mixing Tube d = 5.084" A_d/A_m = 1.61 LD #1 Nozzle Cluster Case 3 Nozzles



Static Pressures in Mixing Tube Short Constant Area Mixing Tube LD #1 Nozzle Cluster Case 3 Nozzles


Test	P po psia	T po °F	W _p lbm/min	T so °F	W _s lbm/min	m	Probe	Nozzles
0	313	140	5.95	71	146.9	24.7	Kiel	9 Case 2
	313	218	5.75	70	146.0	25.4	Wedge	9 Case 2
	313	223	6.04	79	154.9	25.6	Kiel	8 Case 2 1 Case 4 in center

Figure 85 Velocity Profiles with Unheated Primary Flow Short Constant-Area Mixing Tube LD#1 Nozzle Cluster

213





Case 2 Nozzle, Extended Mixing Tube $\overline{P} = 17.7, P_{po} = 260 \text{ psia}$ \bar{T} = 1.678, T_{po} = 455°F

> NASA-Langley, 1970 -- 1